

Understanding the Health Effects of Ambient Ultrafine Particles

HEI Review Panel on Ultrafine Particles

HEALTH EFFECTS INSTITUTE

Understanding the Health Effects of Ambient Ultrafine Particles

HEI Review Panel on Ultrafine Particles

HEI Perspectives 3 Health Effects Institute Boston, Massachusetts

Trusted Science • Cleaner Air • Better Health

Publishing history: This document was posted at www.healtheffects.org in January 2013.

Citation for document:

HEI Review Panel on Ultrafine Particles. 2013. Understanding the Health Effects of Ambient Ultrafine Particles. HEI Perspectives 3. Health Effects Institute, Boston, MA.

© 2013 Health Effects Institute, Boston, Mass., U.S.A. Cameographics, Blue Hill, Me., Compositor. Printed by Recycled Paper Printing, Boston, Mass. Library of Congress Catalog Number for the HEI Report Series: WA 754 R432.

Cover paper: made with at least 55% recycled content, of which at least 30% is post-consumer waste; free of acid and elemental chlorine. Text paper: made with 100% post-consumer waste recycled content; acid free; no chlorine used in processing. The book is printed with soy-based inks and is of permanent archival quality.

CONTENTS

About HEI		vii
Contributors		ix
Executive Summary		I
PROJECT REPORT HEI Review Panel on Ultrafine Particles		7
CHAPTER I. Introduction		7
What Are UFPs and Why Is There Concern Ab	out Them?	7
Time for a Broad Perspective		8
CHAPTER 2. Ambient UFPs: Sources, Emissions, Exposures. To What Extent do Motor Vehicles C	, and Contribute?	10
How Are UFPs Measured?		10
Measurement of Ultrafine Particle Mass Reconstructed Particle Mass (PM _{0.1}) Time-resolved Measurements of UFP		
Chemical Composition	13	
Number Concentration	13	
Summary of UFP Measurements	14	
Sources of Ambient UFPs		14
Emissions Inventories	14	
Source Apportionment	15	
Particle Number Counts by Location	16	
Summary of Ambient UFP Sources	17	
Emissions of UFPs from Motor Vehicles		17
Generation and Characterization of	10	
UFPs in Engine Emissions	18	
Diesel Engines	18	
Spark-ignition Engines	21	
Potential Effects of Other New Fuels	22	
Buses Powered by CNG	22	
Biodiesel	22	
Ethanol	22	
Electric Drive Technologies	23	
Summary of Motor Vehicle UFP Emissions	23	
Characterizing Human Exposure to Ambient UFPs		23
Factors Affecting Concentrations and Composition of Ambient UFPs	23	
Spatial Variation of Ambient UFP Concentrations	25	

Sidebar I. Regional Nucleation Events	25	
Temporal Variations	27	
Copollutant Concentrations	28	
Microenvironmental Exposures		
to UFPs	28	
Ambient Contributions to Indoor Concentrations of UFPs	29	
In-Vehicle Exposures	29	
Implications for Other Countries	30	
Modeling UFP Concentrations	30	
Summary of Evidence Characterizing Human Exposure to Ambient UFPs	31	
CHAPTER 3 Do LIEPs Affect Health? What is th	e Fvidence	
from Experimental Studies in Animals and Huma	ans?	31
Deposition Clearance and Translocation of L	FPs	30
Particle Clearance and Potentian	113	JZ
in the Respiratory Tract	33	
Particle Translocation	33	
Laboratory Animal Studies	33	
Human Studies	34	
Summary of Particle Deposition,		
Clearance, and Translocation	35	
Experimental Studies of Adverse Effects of		
Exposure to UFPs in Animals and Humans		35
Laboratory Animal Studies	36	
Sidebar 2. Diesel Engine Exhaust —		
Components and Health Effects in	27	
Allergic Responses	20	
Cardiovascular Besponses	30	
	41	
Summary of Animal Studies	41	
Experimental Human Exposures		
to UFPs	41	
Respiratory Responses	42	
Cardiovascular Responses	44	
Other Responses	47	
Summary and Conclusions for		
Experimental Studies	47	
CHAPTER 4. Do UFPs Affect Human Health at		
Environmental Concentrations? VVhat is the		17
		1/
Evidence Base		48
Previous Reviews		48

Long-Term Effects		49
Short-Term Effects		49
Mortality	51	
Cardiorespiratory Acute Morbidity	52	
Respiratory Effects	53	
Respiratory Symptoms	53	
Pulmonary Function	53	
Allergy and Atopy	54	
Cardiovascular Effects	54	
Heart-Rate Variability	54	
Arrhythmia and Related		
Endpoints	54	
Ischemia	55	
Vascular Reactivity	55	
Blood Pressure	55	
Soluble Markers	55	
Exposure Assessment Considerations		57
Copollutant Confounding	57	
Exposure Measurement Error	57	
Population-Based (Time-Series)		
Studies	57	
Panel Studies	57	
Addressing Potential Exposure Error	58	
Epidemiologic Studies Using Measures of UFP Mass	58	
Summary and Conclusions from Epidemiology		63
Inconsistency of Results by Endpoint	63	
Exposure Assessment	63	
Assessment of the Independence		
of UFP Effects	63	
HAPTER 5. Summary and Conclusions		64
Summary		64
Conclusions		65
Where Do We Go From Here?	66	
Experimental Studies	66	
Experimental Studies	66	
Bottor Characterization of Ambient	00	
UFP Exposures	67	
EFERENCES		67
PPENDIX A. Overview of HEI Research Program on UFPs		86
PPENDIX B. Primary Research Articles		92
BBREVIATIONS AND OTHER TERMS		106
El Board, Committees, and Staff		107

ABOUT HEI

The Health Effects Institute is a nonprofit corporation chartered in 1980 as an independent research organization to provide high-quality, impartial, and relevant science on the effects of air pollution on health. To accomplish its mission, the institute

- Identifies the highest-priority areas for health effects research;
- Competitively funds and oversees research projects;
- Provides intensive independent review of HEI-supported studies and related research;
- Integrates HEI's research results with those of other institutions into broader evaluations; and
- Communicates the results of HEI's research and analyses to public and private decision makers.

HEI typically receives half of its core funds from the U.S. Environmental Protection Agency and half from the worldwide motor vehicle industry. Frequently, other public and private organizations in the United States and around the world also support major projects or research programs. For this project, the preparation and publication of this document was partially supported by the Federal Highway Administration.

HEI has funded more than 280 research projects in North America, Europe, Asia, and Latin America, the results of which have informed decisions regarding carbon monoxide, air toxics, nitrogen oxides, diesel exhaust, ozone, particulate matter, and other pollutants. These results have appeared in the peer-reviewed literature and in more than 200 comprehensive reports published by HEI.

HEI's independent Board of Directors consists of leaders in science and policy who are committed to fostering the public-private partnership that is central to the organization. The Health Research Committee solicits input from HEI sponsors and other stakeholders and works with scientific staff to develop a Five-Year Strategic Plan, select research projects for funding, and oversee their conduct. The Health Review Committee, which has no role in selecting or overseeing studies, works with staff to evaluate and interpret the results of funded studies and related research.

All project results and accompanying comments by the Health Review Committee are widely disseminated through HEI's Web site (*www.healtheffects.org*), printed reports, newsletters and other publications, annual conferences, and presentations to legislative bodies and public agencies.

CONTRIBUTORS

In Spring 2011, the Health Effects Institute appointed an expert panel to review and critique the scientific literature on the ultrafine particles — their sources, the role of automobile emissions, and their potential health effects at ambient levels of exposure. The panel consisted of scientists from a variety of disciplines and was chaired by Mark Frampton, a professor of medicine and environmental medicine at the University of Rochester

HEI Review Panel on Ultrafine Particles

- Mark W. Frampton, Chair, Professor of Medicine and Environmental Medicine, University of Rochester Medical Center; member, HEI Review Committee
- Michael Brauer, Professor, School of Population and Public Health, University of British Columbia; member, HEI Review Committee
- Michael Kleeman, Professor, Department of Civil & Environmental Engineering, University of California–Davis

Peer Reviewers

- **Nigel N. Clark,** George Berry Chair of Engineering, West Virginia University
- Jack R. Harkema, University Distinguished Professor, Department of Pathobiology and Diagnostic Investigation, Michigan State University College of Veterinary Medicine
- Philip K. Hopke, Bayard D. Clarkson Distinguished Professor; Director, Institute for a Sustainable Environment; and Director, Center for Air Resources Engineering and Science, Clarkson University
- Nino Künzli, Head, Department of Epidemiology and Public Health, and Deputy Director, Swiss Tropical and Public Health Institute

HEI Science Staff

Katherine Walker, Project Leader, Senior Scientist Kate Adams, Senior Scientist Rashid Shaikh, Director of Science

HEI Publications Staff

Suzanne Gabriel, Editorial Assistant Hope Green, Editorial Assistant Fred Howe, Consulting Proofreader Medical School. HEI is indebted to the panel for its expertise, cooperation, and enthusiasm. The panel also received support from a team of HEI staff, under the leadership of Katherine Walker, Senior Scientist. A draft of the resulting report was submitted for outside peer review; the help of the peer reviewers in improving the quality of this document is gratefully acknowledged.

- Wolfgang G. Kreyling, Biophysicist Emeritus, Institute of Lung Biology and Disease, Helmholtz Zentrum München, German Research Center for Environmental Health
- Leonidas Ntziachristos, Assistant Professor, Laboratory of Applied Thermodynamics, Mechanical Engineering Department, Aristotle University Thessaloniki
- **Stefanie Ebelt Sarnat,** Assistant Professor, Department of Environmental Health, Rollins School of Public Health, Emory University
- Jonathan Levy, Professor of Environmental Health, Boston University School of Public Health
- Morton Lippmann, Professor, Nelson Institute of Environmental Medicine, New York University Medical Center
- Juha Pekkanen, Professor, Department of Environmental Health, National Institute for Health and Welfare, Unit of Public Health and Clinical Nutrition, University of Eastern Finland
- Kent Pinkerton, Professor and Director, Center for Health and the Environment, University of California–Davis
- Ira Tager, Emeritus Professor, School of Public Health, Division of Epidemiology, University of California–Berkeley
- Clifford P. Weisel, Professor, Environmental and Occupational Health Sciences Institute, Rutgers University

Geoffrey H. Sunshine, Senior Scientist Jacqueline Presedo, Research Assistant Morgan Younkin, Research Assistant

Carol Moyer, Consulting Science Editor Ruth Shaw, Consulting Compositor

Understanding the Health Effects of Ambient Ultrafine Particles

INTRODUCTION

Over the past 30 years, a large body of scientific literature has emerged that provides evidence of associations between short-term and long-term exposures to ambient particulate matter (PM) and increased mortality and hospitalization from cardiovascular and respiratory diseases. Most of the evidence is based on epidemiologic studies of human exposure to PM with aerodynamic diameters ≤ 10 micrometers (PM₁₀) or ≤ 2.5 micrometers (PM_{2.5}). However, scientists and regulators have long known that PM in the ambient air is a complex mixture including particles of different sizes and chemical composition. What has been less clear is whether certain characteristics of the ambient mixture are more harmful to public health than others and are therefore the most important to control. In its 1998 blueprint for a research program on airborne PM, the United States National Research Council identified improved understanding of ultrafine particles (UFPs) as a priority.

UFPs make up the smallest size fraction in what is a continuum of airborne particles with diameters ranging from a few nanometers to several micrometers. By convention, UFPs have been defined as particles that are 100 nanometers or less in diameter (\leq 100 nm). Given their small size, UFPs contribute little to the mass of PM in ambient air, but they are the dominant contributors to particle number. Motor vehicles, especially those powered by diesel engines, have often been cited as a leading source of ambient UFP emissions and of human exposure. Concern about UFPs developed from early evidence, primarily from animal and in vitro studies, that suggested that they could be inhaled more deeply into the lung and might be more toxic than larger particles. The first epidemiologic studies that included particle number measurements also suggested that UFPs might be associated with the same adverse effects in humans that have been attributed to larger particle size fractions. Scientists hypothesized that UFPs would have greater toxicity than larger particles in part because their vast numbers and small diameters mean that they have a high surface area, a potentially important interface through which to transmit any toxic chemicals that might be adsorbed.

In the decades since concerns were first raised about UFPs, the role they might play in the adverse health effects associated with exposures to air pollution has remained an important research target at institutions around the world, including HEI. National and local air quality authorities in the United States and in other regions of the world continue to assess the need for specific action on UFPs in reviews of ambient air quality standards and other regulatory programs. At the same time, under existing regulatory and technological changes, UFP emissions from motor vehicles are already changing. The resulting impacts on ambient concentrations, and ultimately on human exposures, are difficult to predict.

TIME FOR A BROAD PERSPECTIVE

Given this context, HEI formed a special panel (see Contributors list) to review the scientific evidence available on UFPs and to present its evaluation in this third issue of the HEI Perspectives series: Understanding the Health Effects of Ambient Ultrafine Particles.

The work of the HEI Review Panel on Ultrafine Particles was supported with funding from the United States Environmental Protection Agency (Assistance Award CR-83234701) and motor vehicle manufacturers. Support for the preparation and publication of this document was provided by the Federal Highway Administration (Grant DTFH61-09-G-00010). This report has not been subjected to peer or administrative review by any of the sponsors and may not necessarily reflect their views, and no official endorsement should be inferred.

The Panel structured its assessment of the scientific evidence regarding ambient UFPs as responses to three questions:

- Ambient UFPs sources, emissions, and exposures: To what extent do motor vehicles contribute? (Chapter 2);
- Do UFPs affect health? What is the evidence from experimental studies in animals and humans? (Chapter 3);
- Do UFPs affect human health at environmental concentrations? What is the evidence from epidemiologic studies? (Chapter 4).

Chapter 2 explores the contribution of motor vehicles within the broader context of the multiple sources of ambient UFPs. It discusses in detail the changing profiles of mobile-source emissions, the spatial and temporal patterns of ambient UFP concentrations, and the implications of all these factors for the design and interpretation of studies of UFP exposure and health.

The next two chapters explore the health evidence on UFP exposures from a broad array of study designs using animal and human subjects. Chapter 3 focuses on the evidence from experimental studies in animals and in humans because they can directly test hypotheses about the causal role of specific exposures.

Chapter 4 focuses on observational epidemiologic studies of people exposed to UFPs in the environment, in mostly urban settings. Because they involve studies of people exposed to concentrations of air pollutants found in the real world, epidemiologic studies of UFPs have the potential to provide more direct evidence with which to determine whether UFPs affect human health at concentrations found in the environment.

Chapters 3 and 4 both focus on various measures of intermediate markers and health endpoints that represent the multiple hypothesized pathways for UFP effects. Most of these pathways are shared by PM generally, but some pathways may be especially relevant for UFPs.

In identifying experimental and epidemiologic studies for its assessment, the Panel made a number of choices to make sure that responses to the questions were most informed by studies relevant to the understanding of the potential risks of inhaling ambient UFPs, particularly those related to motor vehicle exhaust. For the experimental studies, it considered only studies involving exposures to UFPs via the inhalation route, which is physiologically relevant and directly comparable with the results of epidemiologic studies. The Panel therefore excluded in vitro studies or studies in which particles were directly instilled into the lungs or airways. The Panel focused on exposures to combustion-related UFPs and therefore largely excluded the vast literature on engineered nanoparticles. The Panel also placed particular emphasis on both experimental and epidemiologic studies of UFPs that included analyses of exposures to copollutant gases and larger particle size fractions, because of the potential of such studies to provide insight into the role of UFPs themselves in any health effects observed.

Finally, Chapter 5 summarizes each chapter's main conclusions and attempts to identify some of the broader lessons, about both the specific health effects associated with exposures to UFPs and possible directions for future studies that could enhance our understanding of emissions, exposures, and effects of UFPs.

SUMMARY AND CONCLUSIONS

A substantial body of literature has now been published on the sources of UFPs, their spatial and temporal distribution in ambient air, their inhalation and fate in the body, their mechanisms of toxicity, and their adverse effects in animals and in humans. The purpose of this issue of HEI Perspectives is to provide a broad assessment of what has been learned about UFPs and what remains poorly understood. The Panel's findings in response to the three questions posed at the outset of this Executive Summary are summarized briefly below.

AMBIENT UFPS — SOURCES, EMISSIONS, AND EXPOSURES: TO WHAT EXTENT DO MOTOR VEHICLES CONTRIBUTE?

As products of combustion and secondary atmospheric transformations, ambient UFPs have multiple sources whose relative contributions to ambient concentrations vary with location, season, and time-of-day. However, in urban areas, particularly in proximity to major roads, motor vehicle exhaust can be identified as the major contributor to UFP concentrations. Diesel vehicles have been found to contribute substantially, sometimes in disproportion to their numbers in the vehicle fleet.

However, the absolute and relative contributions of different vehicle types to motor vehicle emissions are changing rapidly. On the one hand, under the force of regulations to reduce particle mass and number emissions from diesel and other vehicles, the emissions, and therefore ambient levels, of UFPs will decrease. On the other hand, this decrease may be partially offset by UFP emissions from the growing use of certain types of gasoline direct injection technology to boost fuel efficiency. The role that will be played by new fuels, such as ethanol and biodiesel blends and natural gas, remains largely ill-defined. The collective effect of all these changes has not been thoroughly explored and will likely vary regionally, depending on the rate and extent to which they are deployed in different parts of the world.

It has been more challenging to characterize human exposure to ambient UFPs than to the more regionally dispersed and routinely monitored pollutants, such as PM_{2.5}. UFP concentrations are highly variable spatially, declining rapidly with distances from roadways, for example, such that UFPs often differ substantially from one location to another within the same city. Given their small contribution to mass, UFPs are not well reflected in PM mass measurements and they are not routinely monitored in most locations. Studies of UFPs have relied on a variety of detection methods, most commonly measures of number concentration. In addition, UFPs are highly correlated with other combustion-related pollutants, such as carbon monoxide and nitrogen oxides. These correlations must be taken into account when evaluating exposure to sources such as traffic, or when designing epidemiologic studies and interpreting their results. Reliance on measurements at central-site monitors to represent broad population exposure — a central feature in epidemiologic studies of longterm exposures to PM_{2.5} and other pollutants — is likely to lead to errors in estimates of exposure to UFPs.

Despite the high spatial variability of UFPs, the UFP number concentrations measured at multiple locations within cities do tend to be reasonably correlated in time, rising and falling in similar patterns over the course of a day. Moderately high temporal correlations between UFP number concentrations at central monitors, outdoors at residences, and even indoors at residences have been observed in some, but not all, cities. The correlations are not always as strong as those observed for PM_{2.5}, but in some locations they can be sufficient to support epidemiologic studies on the effects of short-term variations of number concentrations on human health, using study designs that have been employed for larger particle size fractions. However, the temporal variability in UFP number concentration can be similar to that of other PM size fractions and gaseous pollutants, making it difficult to differentiate the effects of UFP number concentration in such study designs.

DO UFPS AFFECT HEALTH? WHAT IS THE EVIDENCE FROM EXPERIMENTAL STUDIES IN ANIMALS AND HUMANS?

Experimental studies have provided a rationale for the hypothesis that the adverse health effects of exposure to UFPs differ from those of larger particles. As a result of their physical characteristics, inhaled UFPs differ from larger particles in their deposition patterns in the lung, their clearance mechanisms, and in their potential for translocation from the lung to other tissues in the body. Some animal studies have also demonstrated translocation of UFPs via the olfactory nerve to the brain.

Both animal and human studies provide evidence for respiratory and cardiovascular effects associated with exposure to UFPs. Observed effects in selected studies include lung function changes, airway inflammation, enhanced allergic responses, vascular thrombogenic effects, altered endothelial function, altered heart rate and heart rate variability, accelerated atherosclerosis, and increased markers of brain inflammation. Largely, with the exception of brain effects, the findings are similar to those observed for exposures to fine particles.

While selected studies show evidence for UFP effects, the current evidence, when considered together, is not sufficiently strong to conclude that short-term exposures to UFPs have effects that are dramatically different from those of larger particles. There are limitations and inconsistencies in the findings from short-term studies on UFP health effects, and there are no long-term animal exposure studies of UFP health effects. Relatively few studies have directly compared UFPs with other particle size fractions. These factors constrain our ability to draw definitive conclusions about the specific consequences of exposure to UFPs.

DO UFPS AFFECT HUMAN HEALTH AT ENVIRONMENTAL CONCENTRATIONS? WHAT IS THE EVIDENCE FROM EPIDEMIOLOGIC STUDIES?

A growing number of epidemiologic studies conducted over roughly the past 10 years have evaluated impacts of UFPs. These studies have provided suggestive, but often inconsistent, evidence of adverse effects of short-term exposures to ambient UFPs on acute mortality and morbidity from respiratory and cardiovascular diseases. One explanation that must be considered for the results to date is weakness in the true underlying relationship between UFP exposures and adverse effects — that the null hypothesis being tested by these studies is true. However, limitations of the current studies are likely to play a role; UFPs have not been assessed routinely in large epidemiologic studies of air pollution health effects, in part because ambient monitoring of UFPs has not been conducted in most locations or has not been done with the same measurement techniques. As a result, studies tend to be smaller and the likelihood of exposure measurement error tends to be greater for UFPs relative to $PM_{2.5}$ and other pollutants; both of these factors reduce statistical power to test confidently for what may be small but important health outcomes.

The available observational study designs have also not been able to clearly determine whether UFPs have effects independent of those for related pollutants. Where studies have measured UFPs, few have assessed whether the effects associated with UFPs are independent of other pollutants. When they have, the effects of UFPs have not been consistently discernible from those of other pollutants with which they often occur or share similar sources (e.g., traffic). Of 42 articles published since 1997 that cited any significant health associations with UFPs measured as number concentration, 37 articles also noted significant effects for other particle size fractions or traffic-related pollutants, and 10 articles did not consider any traffic-related gases in the analysis.

No epidemiologic studies of long-term exposures to ambient UFPs have been conducted. This is because the most common epidemiologic study designs for long-term exposures are dependent on spatial contrasts in concentrations that have been more difficult to characterize for UFPs than for PM_{2.5}.

OVERALL CONCLUSIONS

Airborne PM has been the focus of extensive research and debate in the United States and around the world for several decades. Considerable evidence from a broad array of experimental and epidemiologic studies has led to strong scientific consensus on the independent associations of airborne PM, in particular $PM_{2.5}$ and PM_{10} , with adverse respiratory and cardiovascular effects on human health. This evidence has provided the foundation for many regulatory decisions to limit both PM emissions, including those from motor vehicles, and ambient PM concentrations to which people might be exposed.

What role have ambient concentrations of UFPs played in the adverse effects that have been observed in human populations exposed to ambient air pollution?

Several factors — the unique physical properties of UFPs, their interactions with tissues and cells, their potential for translocation beyond the lung — have led scientists to expect that UFPs may have specific or enhanced toxicity relative to other particle size fractions and may contribute to effects beyond the respiratory system. However, the considerable body of research that has been conducted has not provided a definitive answer to this question. Toxicologic studies in animals, controlled human exposure studies, and epidemiologic studies to date have not provided consistent findings on the effects of exposures to ambient levels of UFPs, particularly in human populations. The current evidence does not support a conclusion that exposures to UFPs alone can account in substantial ways for the adverse effects that have been associated with other ambient pollutants such as $PM_{2.5}$.

The fact that the current database of experimental and epidemiologic studies does not support strong and consistent conclusions about the independent effects of UFPs on human health does not mean that such effects, as one part of the broader effects attributable to $PM_{2.5}$, can be entirely ruled out. There are limitations in the evidence base attributable to underlying deficiencies in exposure data, to numerous challenges in comparing and synthesizing results of existing studies, and to the inherent complexity of the task that scientists have set out to accomplish.

WHERE DO WE GO FROM HERE?

There are many considerations beyond the scientific opinions expressed in this issue of HEI Perspectives that inform the level of confidence in the evidence necessary for policy makers to "ensure that resources spent in the future on control technology and regulatory compliance will have a reasonable probability of success" (U.S. National Research Council 1998). Among them is the need to weigh carefully the value to scientific understanding and to regulatory decisions of continuing to treat UFPs as an individual pollutant versus alternative approaches that focus on the health effects of exposure to traffic or to the broader air pollution mixture.

As part of that discussion, this report lays out possible research steps toward addressing some of the limitations of the current evidence on the specific role of UFPs. Experimental study designs could include controlled exposures to UFPs and related copollutants in studies that replicate key animal research results on effects beyond the lung (e.g., in the cardiovascular and central nervous systems), that extend analyses to other animal species and disease models, and that involve long-term exposures. Epidemiologic studies could include more carefully targeted designs that exploit contrasts in ambient UFP exposures but that improve the ability to characterize the independent effects of exposure to UFPs, more consistent and comparable study designs that would support meta-analyses, and designs that permit assessment of the impacts of long-term exposures. Ultimately, many of the underlying challenges posed by the existing evidence on ambient UFPs relate to limitations in characterization and analysis of exposure, so recommendations for exploration of alternative exposure metrics, spatial modeling techniques, and statistical methods are also included.

Regardless of the evidence for a specific role for UFPs, many of the recent PM regulatory decisions affecting fuels, engine designs, and exhaust aftertreatment in countries around the world are likely to result in significant reductions in emissions of both fine and ultrafine particles. The time course of these and other changes in the emissions of UFPs or their precursors and their impact on ambient concentrations will depend on a number of factors, including shifts in the size, age, and composition of the vehicle fleet in particular regions. Monitoring and evaluation of such changes will be essential in the years to come; without them, questions will remain about whether or not these changes have addressed the most important characteristics of the air pollution mixture.

REFERENCE

U.S. National Research Council. 1998. Research Priorities for Airborne Particulate Matter: I. Immediate Priorities and a Long-Range Research Portfolio. National Academy Press, Washington, D.C.

Understanding the Health Effects of Ambient Ultrafine Particles

HEI Review Panel on Ultrafine Particles

CHAPTER 1. Introduction

The history of air pollution research and management has been characterized by efforts to identify the key pollutants responsible first for the major air pollution episodes that plagued the industrializing countries in the early part of the 20th century, and later, for the more subtle geographic variations in air pollution levels and health effects. Terms like smoke and haze have given way to more specific chemical entities - specific gaseous pollutants and solid particulate matter. In turn, the study of particulate matter (PM*) exposures, which began with crude measures of total suspended particles, evolved to focus on increasingly smaller particle size fractions that are more likely to be inhaled, beginning with $PM \leq 10$ micrometers in aerodynamic diameter (PM₁₀), then ≤ 2.5 micrometers in aerodynamic diameter (PM_{2.5}), and recently on the complex mixture of constituents of which they are comprised. The fundamental motivation underlying these research efforts has been to identify those characteristics of air pollution that are most hazardous to human health and whose control would most likely lead to reductions in risks to public health. Interest in ultrafine particles (UFPs) — particles \leq 100 nanometers in diameter — is very much a part of this history.

WHAT ARE UFPS AND WHY IS THERE CONCERN ABOUT THEM?

UFPs in ambient air make up the smallest size fraction in what is a continuum of particles with diameters ranging from a few nanometers to several micrometers (illustrated in Figure 1 for a typical roadway aerosol). By convention, UFPs have been defined as particles less than or equal to 100 nanometers in diameter (\leq 100 nm or \leq 0.1 µm). UFP size fractions may also be characterized more generally in terms of the processes by which they are formed; *nucleation mode* particles (< 50 nm) and the larger *accumulation mode* particles (> 50 nm) (HEI 2010). UFPs technically are part of the larger size ranges that have been the primary subjects of air pollution studies (i.e., PM₁₀ and PM_{2.5}). They contribute little to the mass of particles measured in these ranges, but are the dominant contributors to particle number.

Over the last 30 years, a large body of scientific literature has emerged that provides evidence of associations between short-term and long-term exposures to ambient PM_{10} and $PM_{2.5}$ and increased rates of mortality and hospitalization, primarily from cardiovascular and respiratory diseases. The most influential evidence first came from observational epidemiologic studies in the United States and around the world (e.g., Pope et al. 1992; Dockery et al. 1993; Anderson et al. 1997; Samet et al. 2000a,b; HEI 2003; Pope and Dockery 2006). The biological explanations for these findings were then, and continue to be, the subject of substantial research and debate.

Evidence from studies in laboratory animals had begun to accumulate by the early 1990s, which suggested that UFPs might penetrate more deeply into the lung and might be more toxic than larger particles (Oberdörster et al. 1990; Ferin et al. 1992; International Commission on Radiological Protection [ICRP] 1994). Of concern to air pollution scientists is that particles in the UFP size range account for the vast percentage of particle numbers in ambient air, even though they make up a small fraction of the $PM_{2.5}$ or PM_{10} mass (see Figure 1). A measured $PM_{2.5}$ mass concentration of 10 µg/m³ for example, might contain as many as 2.4 million 20-nm particles/cm³, but could also be represented by a single 2.5 µm particle (Oberdörster et al. 1995). Seaton and colleagues (1995) hypothesized that this *urban particulate cloud* of UFPs, could cause "alveolar inflammation,

The work of the HEI Review Panel on Ultrafine Particles was supported with funding from the United States Environmental Protection Agency (Assistance Award CR-83234701) and motor vehicle manufacturers. Support for the preparation and publication of this document was provided by the Federal Highway Administration (Grant DTFH61-09-G-00010). This report has not been subjected to peer or administrative review by any of the sponsors and may not necessarily reflect their views, and no official endorsement should be inferred.

 $^{^{\}ast}$ A list of abbreviations and other terms appears at the end of this document.



Figure 1. Normalized particle size distributions of typical roadway aerosol. Dp represents the particle diameter; $(1/C_{total})dC/dlog$ Dp represents the logarithmic particle-concentration-distribution function weighted by number, volume (surface), and mass. Here, C is the concentration (number, surface, or mass) in a particular size range and C_{total} is total concentration summed over all sizes. (Source: David Kittelson and Win Watts, reprinted from HEI 2010.)

with release of mediators capable, in susceptible individuals, of causing exacerbations of lung disease and of increasing blood coagulability, thus also explaining the observed increases in cardiovascular deaths associated with urban pollution episodes."

The high surface area per unit of mass of UFPs, a function of their vast numbers and small diameters, has also been hypothesized to be an important characteristic that might predict greater toxicity of particles in that size range. An early toxicologic study by Oberdörster and colleagues (1992), in which they instilled 20 nm titanium dioxide particles in the tracheas of rats, found that increased pulmonary toxicity was associated with the surface area of the particles. Other investigators have noted that surface area is a potentially important interface by which particles interact with biological systems and help to transport toxic metals or chemicals that may be adsorbed to the particles (Donaldson et al. 2005; Kreyling et al. 2006a; Maier et al. 2008).

Such early studies, coupled with later epidemiologic evidence that UFPs might be associated with adverse effects in humans similar to those observed for other particle size fractions (Pekkanen et al. 1997; Peters et al. 1997; Wichmann et al. 2000), motivated the U.S. National Research Council to identify UFPs as a research priority in their series of reports laying out a blueprint for a multifaceted research agenda on airborne PM (U.S. National Research Council 1998, 1999, 2001, 2004).

TIME FOR A BROAD PERSPECTIVE

In the decades since concerns were first raised about UFPs, these particles have remained an important research target at institutions around the world. At HEI, research on UFPs has been part of an ongoing effort to advance understanding of the associations between exposures to ambient PM and adverse effects on human health (see Appendix Table A.1 for a complete overview of HEI's research program involving UFPs, including published HEI reports with their related journal articles and ongoing HEI research). HEI's work has included some of the early research by Oberdörster and colleagues (2000) on the pulmonary effects of model UFP exposures in susceptible rats and mice, as well as the first epidemiologic study to investigate and to observe an association of short-term daily exposure to UFPs with mortality (Wichmann et al. 2000). Several additional studies are still underway and include novel methods to improve real-time measurement and characterization of UFPs, modeling of personal exposures to UFPs from primary exhaust and secondary aerosols, and toxicologic and epidemiologic studies in settings with distinct variations in UFP concentrations.

Motor vehicles have often been cited as a key source of exposure to UFPs, and two recent HEI reviews have laid the initial groundwork on this issue. The HEI Special Report on traffic (HEI 2010) was an extensive review of the literature on traffic-related emissions, exposures, and associated health effects. As part of that report, UFPs were explored as one of several possible surrogate markers for exposure to traffic, but the evidence for supporting such a role for UFPs was found to be limited. HEI's Communication 16, a report of the HEI Special Committee on Emerging Technologies (HEI 2011), identified potential changes in UFP number emissions and composition that might be associated with future fuels and technologies.

Reflecting these ongoing concerns, UFPs continue to be considered by national and local air quality authorities in the United States and elsewhere in reviews of ambient air quality standards and other regulatory programs. The U.S. Environmental Protection Agency (U.S. EPA) assessed the evidence on UFPs as part of its most recent scientific review of the National Ambient Air Quality Standards (U.S. EPA 2009). While the agency judged the present health effects and air quality data to be insufficient to support an individual standard for UFPs, its scientific advisory panel, which reviews the U.S. EPA assessments, suggested that the role of UFPs continue to be evaluated and that future PM monitoring efforts be extended to the ultrafine range (Clean Air Scientific Advisory Committee [CASAC] 2010). In Europe, UFPs are one of several air pollutants being evaluated under the World Health Organization-led project known as REVIHAAP, which is designed to inform revisions of European Union policies on air quality in 2013.

Though not directly based on health considerations, the European Union has introduced particle number emissions standards for all diesel passenger and commercial vehicles, which are being introduced gradually over the period 2011–2013, and has required recently that they be extended to gasoline vehicles. These limits regulate the number of nonvolatile particles and effectively ensure that particle filters will be installed on both diesel and gasoline vehicles.

Future trends in the United States and other industrialized countries in the ambient levels of UFPs are somewhat hard to predict. On the one hand, the emissions, and therefore ambient levels, of UFPs will decrease under the force of regulations to reduce particle emissions from diesel vehicles; on the other hand, the growing use of gasoline direct injection technology — which raises fuel efficiency — is likely to increase UFP levels. The role that will be played by new fuels, such as ethanol and biodiesel blends and natural gas, remains largely ill-defined at this point.

Where are we now? With nearly two decades of research behind us, regulatory actions are underway that will influence emissions of UFPs. However, resolving questions about the specific role that ambient levels of UFPs may play in potential adverse effects on human populations remains a challenge. Given this context, HEI decided to form a special panel (listed on page ix) to review the scientific evidence available on UFPs. The expert panel held an initial meeting in July 2011 to lay the groundwork for the report and worked collaboratively thereafter with HEI staff to draft it. The draft was sent to 10 external peer reviewers and was revised extensively in response to their comments. The final result of this process is the current issue in the series of HEI Perspectives, in which we have sought to provide a broad overview of the scientific evidence regarding ambient UFPs, structured as responses to three auestions:

1. Ambient UFPs — sources, emissions, and exposures: To what extent do motor vehicles contribute?

Mobile sources are often cited as the leading source of human exposure to ambient UFPs. Chapter 2 explores the basis for this common statement by examining the many factors that affect the magnitude and potential for human exposures to ambient UFPs. We begin with a survey of the multiple sources of ambient UFPs and how they are measured. We then discuss in more detail the changing profiles of mobile-source emissions, the spatial and temporal patterns of ambient UFP concentrations, and the implications of all these factors for the design and interpretation of health studies of UFP exposures.

2. Do UFPs affect health? What is the evidence from experimental studies in animals and humans?

Experimental studies play a critical role in the study of relationships between exposure and disease. They can be designed to characterize the patterns and mechanisms of particle deposition, clearance, and uptake that may be important to understanding the potential differential toxicity of particles. With carefully designed exposures and selected health endpoints in a controlled setting, these studies provide direct tests of hypotheses about the causal role of particular exposures. Chapter 3 focuses on the evidence from experimental studies involving exposures to UFPs via the inhalation route, which is both more physiologically relevant and more directly comparable with the results of epidemiologic studies reviewed for the next question. We therefore excluded in vitro studies or studies in which particles have been directly instilled or deposited into the lungs or airways.

3. Do UFPs affect human health at environmental concentrations? What is the evidence from epidemiologic studies?

Because they involve studies of people exposed to concentrations of air pollutants found in the real world, epidemiologic studies of UFPs have the potential to provide more direct evidence with which to answer whether UFPs affect human health at concentrations found in the environment. In Chapter 4, we evaluate the epidemiologic evidence: 1) for specific health endpoints, with an assessment of the consistency and coherence of observed associations, and 2) with respect to key study design and data issues, including how UFPs are measured, how exposures are assigned to subjects, and the extent to which potential confounding of the UFP effects by copollutants has been assessed.

Our responses to each of these questions have attempted to focus on literature most germane to the key issues. For example, our review of sources, emissions, and exposure has focused on studies that exemplify the major phenomena related to assessing ambient UFPs. In identifying experimental and epidemiologic studies for our assessment, we have focused on those involving combustionrelated UFPs in order to make our assessment most relevant to conclusions about ambient UFPs, particularly those related to motor vehicle exhaust. We have therefore largely excluded the vast literature on engineered nanoparticles that has developed over the last decade, although we recognize that contributions from that literature may also provide insights. We also have placed particular emphasis on both experimental and epidemiologic studies of UFPs that include analysis of exposures to copollutant gases and larger particulate fractions because of their potential to provide insight to the role of UFPs themselves in any health effects observed.

Finally, Chapter 5 summarizes the main conclusions from Chapters 2–4 and attempts to draw some of their broader lessons. In particular, we discuss what the evidence to date allows us to conclude about the health effects associated with exposures to UFPs themselves, and how they differ from those of other particle size fractions and combustion-related copollutants. Possible directions for future studies that could enhance our understanding of emissions of, exposures to, and effects of UFPs are also provided.

CHAPTER 2. Ambient UFPs: Sources, Emissions, and Exposures. To What Extent do Motor Vehicles Contribute?

Many concerns about ambient UFPs have focused on their relationship to motor vehicle emissions. However, as a product of many combustion processes, as well as of secondary chemical and physical processes in the atmosphere, UFP concentrations measured in ambient air are affected by many factors over space and time. This chapter therefore seeks to provide greater perspective on the extent to which motor vehicle emissions may contribute to ambient concentrations of UFPs and ultimately to overall human exposure to ambient UFPs.

The chapter begins with a basic summary of the various methods by which UFPs are measured and characterized that provides a common terminology for the remaining chapters in which the methods are applied to health studies. It then presents a general overview of sources of ambient UFPs at regional and urban scales, offering some perspective on the relative contribution of motor vehicles and traffic. The next subsection provides some background on exhaust and non-exhaust emissions from current engine technology and briefly discusses the implications of changes in engine and fuel technologies for future emissions. The remainder of the chapter focuses on the potential for human exposure to ambient UFPs - in particular, on how concentrations and composition of UFPs vary over time and in different locations where human populations live, travel, and work. We conclude with a discussion of how such data might be used in animal and human studies to better assess the potential effects of UFPs on health.

HOW ARE UFPS MEASURED?

Several different sampling and analysis methods have been developed to analyze UFPs at different points in emission and in ambient air. Single-vehicle measurement offers the possibility of discerning engine, fuel, and aftertreatment effects on UFP emissions and assessing the effect of technology on emissions. Typically conducted in laboratory settings, these measurements may be made under controlled conditions, with set vehicle operating conditions, in a repeatable manner. Such measurements are dedicated to only a few vehicles, may involve stepwise in-laboratory dilution that differs from the faster and continuous atmospheric dilution, and may therefore result in sampling artifacts.

To avoid such limitations, investigators look to ambient sampling of vehicular aerosol to provide a more representative picture of the aerosol to which people in the vicinity of roads may actually be exposed. Ambient aerosols reflect interactions of emitted particles with the environment (e.g., mixing with other particles and gases, photochemical reactions) that can occur over time and space. Disadvantages of this approach for characterizing the contributions of specific vehicles are that the ambient air may include aerosols contributed from different vehicles and vehicle types and typically reflects contributions from background pollutant concentrations and other local emission sources.

These challenges are not unique to the characterization of vehicular emissions and their contributions to ambient UFP concentrations. They affect efforts to study other sources as well. In general, all ambient measurements reflect the product of dynamic atmospheric and chemical processes that are likely to differ over time and geography.

However, one of the factors that has sometimes complicated comparison of data on UFP emissions, concentrations, exposures — and ultimately health effects — has been the many technologies and metrics (mass, composition, surface area, and particle number counts) that have been used to measure and describe them. To support discussions in this and later chapters, the following section describes the various methods by which UFPs are currently characterized; a simple summary of the terminology and the particle size ranges typically measured by the methods discussed is provided in Table 1.

Measurement of Ultrafine Particle Mass

While the measurement of the ambient concentrations of larger particle size fractions (i.e., $PM_{2.5}$ and PM_{10}) is typically based on the total mass per unit volume of air, or mass concentration, several factors make such measurements problematic for UFPs defined as those with diameters less than 100 nm ($PM_{0.1}$). First, direct measurements of UFP mass are challenging because the mass concentration of particles in the UFP range is very low. Ambient $PM_{0.1}$ concentrations are typically less than 1 µg/m³, and commercial balances usually have practical detection limits of $\pm 1-5 \mu g$ for collection media (for example, filters) that weigh a few milligrams. These collection media can sustain only low flow rates (< 100 liters/min), so long collection times are required for sufficient mass to be collected for measurement. These sampling requirements in turn can influence another factor that can significantly affect sensitivity of mass measurements methods for these small size ranges - gas-to-particle artifacts. That is, chemical compounds in the gaseous phase may adsorb on particles to produce a positive artifact or, vice versa, desorb from the particle to produce a negative artifact. Although such processes generally occur in atmospheric conditions as well, the prolonged exposure of the PM to the air flow through the instrument may exacerbate these effects during sampling.

In practice, particle mass concentrations have been more typically estimated for UFPs in larger size fractions known as *quasi-ultrafines*; that is, UFPs < 0.180 µm (PM_{0.18}) or < 0.250 µm (PM_{0.25}) in diameter. Measures of quasi-ultrafine particle mass have been most commonly obtained with a class of instruments known as cascade impactors (see Table 1). In these instruments, particles are sorted and collected on a series of impactor surfaces, each corresponding to a successively smaller aerodynamic diameter cut-point. After collection on the cascade impactor stages, the mass for the fractions of interest is measured by gravimetric analysis. Or, in the case of the electrical low pressure impactor (ELPI) the particles are precharged and then the resultant current is measured on the impaction stages corresponding to various particle size ranges.

Multiple stages of cascade impaction are necessary for accurate UFP mass measurements, in part to remove larger particles from collection on UFP impaction stages. However, larger particles do sometimes bounce or otherwise pass through, and the results can be significantly altered by the accidental inclusion of a few larger size fraction particles with masses that are orders of magnitude greater than those of the UFPs. All of these factors have tended to favor forms of detection other than direct gravimetric analysis.

Reconstructed Particle Mass (PM_{0.1})

As an alternative to direct UFP mass measurements, the concentration of individual chemical components can be measured in the $PM_{0.1}$ size range and combined to effectively reconstruct the total $PM_{0.1}$ mass. Detection limits for the dominant chemical components of $PM_{0.1}$ are usually much better than gravimetric detection limits for UFPs, making reconstructed $PM_{0.1}$ mass potentially more accurate. These measurements have the added advantage that they provide composition data that can also be used in $PM_{0.1}$ source-apportionment analyses.

The bulk of $PM_{0.1}$ mass is typically composed of carbonaceous material with smaller contributions from inorganic ions, reflecting the dominant combustion sources for these particles. The carbonaceous material can be broadly defined as compounds containing elemental carbon (EC) and organic carbon (OC). The EC and OC categories can be further separated into individual compounds, which can be especially useful for $PM_{0.1}$ source-apportionment studies. Due to the large fraction of carbonaceous compounds in

Metric (units)	Abbreviation	Particle Size Ranges	Time Resolution	Method	Selected Instruments
Size-distributed particle mass concentration (µg/m ³)	PM _{x-y}	10 nm-18 μm Various cut-points: < 56 nm < 100 nm < 180 nm < 250 nm < 2.5 μm < 10 μm	Integrated (hr)	Cascade impaction	Cascade Impactors: MOUDI Nano-MOUDI Sioutas
Size-distributed number concentration (particles/cm ³)	PM _{x-y}	7 nm–10 μm Various stages	Integrated (hr)	Electrical low pressure impaction	ELPI
Number concentration (particles/cm ³)	Total NC	2.5 nm– 1000 nm (range can vary)	1 sec	CPC	Many models
Size distributed number concentration (particles/cm ³)	NC _{x-y}	Various ranges: 3–30 nm 30–300 nm 300–800 nm 2 nm–1 µm	1 min	DMA	SMPS
	NC _{x-y}	5.6–560 nm	1 sec	DMA	FMPS, ELPI
Size distribution (dN/dlogDp)		5.6–560 nm	10 samples/sec	DMA	EEPS
Particle size, number and mass		5 nm–2.5 μm	10 Hz data, 200 ms T10-90% response	Electrical mobility measurement	DMS500
Surface area (µm²/cm³)	SA _{x-y}	10 nm– 1000 nm	1 sec	Diffusion charging and electrometer	Nanoparticle Surface Area Monitor, DiSCmini, AeroTrak
		20 nm– 100 nm		Ionization and attachment of lead (Pb)	Epiphaniometer
		2.5 nm– 1000 nm (from SMPS)	1 min (from SMPS)	Derived from size counts assuming spherical particles, uniform density	SMPS
Composition		Various cut-points (see above)	Integrated (hr)	Cascade impaction, extraction, mass spectrometry	Cascade Impactors (see above)
		40 nm–1 μm	1–10 sec	Mass spectrometry	Aerodyne AMS
		30–300 nm	< 1 sec	Aerosol time of flight mass spectrometry	TSI 3800-030
		10–30 nm	< 1 sec	Aerosol time of flight mass spectrometry	NAMS

հե f I I]+ f: .1. th daa

^a AMS = aerosol mass spectrometer; EEPS = engine exhaust particle sizer; ELPI = electrical low pressure impactor; FMPS = fast mobility particle sizer; CPC = condensation particle counter; MOUDI = micro-orifice uniform deposit impactor; NAMS = nanoaerosol mass spectrometer; SMPS = scanning mobility particle sizer.

UFPs, efforts to reconstruct particle mass have focused more on using trace carbonaceous compounds to reconstruct total OC or EC in the particles (Kleeman et al. 2009). While larger dust particles may be mistakenly collected on UFP impaction stages, these tend to be made up primarily of crustal material, such as aluminum and silica.

The measurement of chemical composition is more costly and cumbersome than that of mass and is not currently suited to routine monitoring. However, information about chemical composition may be valuable when testing health effects hypotheses related to specific UFP components or sources.

Time-Resolved Measurements of UFP Chemical Composition

With the previous methods, the chemical composition of UFPs or quasi-UFPs is measured by collecting particles on substrates over some period. More recently, aerosol mass spectrometers and aerosol time-of-flight mass spectrometers have been developed that can measure the chemical composition signature of individual ambient UFPs (or groups of particles) over very short periods, seconds to minutes, for example (Bein et al. 2005; Toner et al. 2008; Klems et al. 2011). With these methods, the sampled particles are first broken down into their component major ions (laser ablation/ionization), which are then analyzed by mass spectrometer. The qualitative spectra produced by each particle can be grouped with similar spectra and compared to source libraries to identify probable emissions sources (Toner et al. 2008). These methods enable highly time-resolved source-apportionment studies of UFP number concentrations (UFP NC). As part of an HEIfunded project, Klems and colleagues (2011) have been developing a nanoaerosol mass spectrometer to measure the composition of individual particles in the 18-24 nm size range and have deployed it to assess the contribution of particular motor vehicles to ambient UFP number and mass concentrations at a major intersection in Delaware, Maryland. However, such methods are still undergoing development and have not been widely applied.

Surface Area Concentration

Given hypotheses about the biological relevance of the high surface area to mass ratio for UFPs, scientists have been particularly interested in measures of surface area to characterize UFP concentrations for use in health studies. As with determining particle size, however, defining a surface area that best describes biological interactions has not been straightforward. Most often, surface area is estimated from particle number and size distribution data and then making assumptions about particle shape, density, and other factors. Instruments have been developed to measure UFP surface area directly. The epiphaniometer estimates the Fuchs surface area as a function of radioactive decay from ²¹¹Pb atoms attached to the measured particles (Gäggeler et al. 1989). Other surface area measurements entail exposing the UFPs to an electrical charge and measuring the resultant current (DiSCmini; Nanoparticle Surface Area Monitor; AeroTrak). Whether calculated or direct measurements characterize the surface area concentration of UFPs in ways that are biologically relevant has not yet been thoroughly studied.

Number Concentration

Perhaps the most straightforward measurement of UFP concentration is to count the total number of particles per unit volume of air, typically referred to as particle number concentration (NC) or for this document total NC. Given the relative ease and reliability with which they can be measured, total NC data are far more common than measures of particle mass, composition, or surface area. NC is also often assumed to be a reasonable proxy for surface area, as NC is assumed to be dominated by smaller particles and basic geometry dictates that, for a given mass of particles, surface area increases rapidly with decreasing particle diameter.

Total NC is generally measured continuously by condensation particle counters (CPC) in which particles previously enlarged by condensation of vapor on the particle surface are counted when they pass through a laser beam. Depending on the instrument, this method can count particles as small as 3 nm in diameter and typically includes particles up to 1000 nm or more in diameter. Although total NC measured this way is not strictly delimited by the 100 nm size definition for UFPs, it is often assumed to be dominated by particles in the UFP range (refer to Figure 1).

These instruments are very versatile and can provide number concentrations for discrete size ranges within the full distribution when used in combination with particle sizers such as differential mobility analyzers (DMA) which separate particles by size before they undergo condensation. The scanning mobility particle sizer (SMPS), for example, is an instrument consisting of both a DMA and a CPC. Due to cost and ease of use considerations, measurement networks have generally favored deployment of particle counters alone, although the use of size-differentiated particle counters is increasing. For this document, we have attempted to distinguish number concentrations for specific size ranges, when available, from total NC, rather than referring to all as UFPs.

Despite the appeal of these instruments, it is worth noting that number counts obtained by the different methods have potential limitations as indicators of UFP concentrations that should be taken into account when interpreting and comparing results of different studies (Morawska et al. 2008). Although total NC has frequently been assumed to be synonymous with UFPs < 100 nm, particle size distributions in other environments are likely to differ from the idealized example for roadside aerosols shown in Figure 1. As indicated in Table 1, particle counters have different lower size limits, and unless specifically limited to 100 nm or another upper limit, number concentrations can have different meanings. Indeed, in a review of 52 studies, Morawska and colleagues (2008) compared total NC measurements obtained using CPCs with those obtained using methods that provide number counts for discrete particle size ranges (i.e., either differential mobility particle sizers [DMPS] or SMPS) for similar environments and estimated that mean and median number concentrations measured by CPC were significantly higher than those measured using DMPS or SMPS. Estimates of, or assumptions about relative surface area concentration, could also be affected. The implications of such differences for health studies have not been evaluated.

Summary of UFP Measurements

Numerous methods have been developed to measure the ambient concentrations and composition of UFPs. The various sampling and analysis techniques offer possibilities to analyze ambient particles in different ways, each with its own advantages and disadvantages. Given their simplicity and potential biological relevance, measurements of total NC have been the most common method used to measure UFPs, although as interest grows in the health relevance of particular size fractions, the use of methods that provide size-specific count data is increasing.

Since studies often report measurements or exposures for "ultrafines" or UFPs regardless of the specific measurement and sampling techniques that have been used, care should be taken when comparing, synthesizing, and interpreting results across studies. Emission factors and characteristics for UFPs should always be given with reference to the sampling conditions and procedure utilized. Otherwise, inconsistent findings in concentrations, size distributions and chemical composition are likely to exist and can complicate comparison of results.

SOURCES OF AMBIENT UFPs

Ambient UFPs have numerous sources. Most are related to combustion processes that include the burning of wood and other forms of biomass, and the combustion of fossil fuels for transportation, home heating, and cooking. UFPs may be emitted directly or may be formed secondarily through chemical reactions or particle–gas interactions in the atmosphere. This section focuses on primary emissions and briefly mentions the secondary formation of aerosol by a number of dynamic processes that occur in the seconds, minutes, hours, and days after an emission occurs.

Emissions Inventories

Emissions inventories, estimated from pollutant emissions factors for individual source categories, have long been used in air pollution studies to provide a regional perspective on the relative contribution of different sources to overall emissions of pollutants. However, few emissions inventories have been created for UFPs. Three are summarized in this section: one in California, one in the United Kingdom, and one in continental Europe. Though constructed using different approaches, they each point to a similar set of source categories, but the relative importance of particular sources has varied by location and time.

The one emissions inventory identified in the United States was conducted for California's South Coast Air Basin using $PM_{0.1}$ mass emissions data from 1996 (Figure 2); it estimated that on-road vehicles and other mobile sources (mostly off-road diesel) accounted for about 53% of UFP emissions (Cass et al. 2000). The emissions inventory from the United Kingdom is based on UFP mass emissions data for 1970 to 2007 from that country's National Atmospheric



Figure 2. Source contributions to UFP emissions in California's south coast air basin (1996) that surrounds Los Angeles. Total $PM_{0.1}$ emissions were 13.25 metric tons per day. (Adapted from Cass et al. 2000, Figure 3a, with permission from the Royal Society.)

Emissions Inventory and is illustrated in Figure 3 (Kuhlbusch and Asbach 2011). It shows an overall decline in the total anthropogenic emissions of UFP mass over time, but shows traffic accounting for about 40% of total emissions in 2007, followed by industrial sources at 30%. The third emissions inventory, illustrated in Figure 4, is based on particle number emissions factors (for particles < 300 nm) for Europe in 2005 (Denier van der Gon et al. 2010; Kulmala et al. 2011). It suggests that road and nonroad transport together accounted for 51% of particle number emissions, followed by residential and commercial heating (21%), and various industrial processes (16%) accounting for most of the remainder (Denier van der Gon et al. 2010).

Although these inventories suggest similar source contributions, limitations of these inventories are that they are typically region specific, not verified with field measurements, and need to be updated over time as changes in emissions occur.

Source Apportionment

A second category of methods, source apportionment, has been used to estimate more directly the contribution of different sources to ambient UFP concentrations. These methods rely primarily on different statistical models (for example, chemical mass balance, principal components, and factor analysis) to infer the contribution of different sources from the chemical composition of particles measured at a given location.

Several source-apportionment studies of $PM_{0.1}$ mass alongside larger size fractions ($PM_{0.18}$) have been conducted at an urban location downwind of Los Angeles and in urban and rural locations of central California (Kleeman et al. 2009; Ham and Kleeman 2011), in the vicinity of major ports (Minguillon et al. 2008), and adjacent to roadsides (Riddle et al. 2008).

Such studies have traced PM_{0.1} to a variety of sources including diesel and gasoline engines, residential wood burning, and cooking from fast food restaurants, among others (Martin et al. 2009: Ham and Kleeman 2011). Data from the same and related studies have also shown that the chemical signature of traffic sources can be estimated for the PM_{0.1} size fraction several meters from major freeways (Kleeman et al. 2008a, 2009; Riddle et al. 2008) and that the relative contribution of different engine types and sources may vary with distance from roadways, by season, by time of day and by location (Kleeman et al. 2008a; Ham and Kleeman 2011). Figure 5 illustrates such variations by juxtaposing results of source-apportionment analyses based on the OC content of PM_{0.1} from those studies. The chemical signature of the organic material in the PM_{0.1} size fraction is altered by atmospheric chemical reactions as these particles age for several days in the atmosphere, making it difficult to link these particles to their emissions



Figure 3. Emission inventory of PM_{0.1} emissions in the United Kingdom from 1970 to 2007. Basic data from the National Atmospheric Emission Inventory (NAEI 2007). (Source: Kuhlbusch and Asbach 2011, reprinted with permission from John Wiley and Sons).



Figure 4. Source sector contributions in 2005 to European particle number emissions < 300 nm. (Total number of particles = 1.9×10^{27}). The estimated contribution from fossil fuel production was negligible. (Adapted from Denier van der Gon et al. 2010, using data provided by the author.)



Figure 5. Predicted source contributions to $PM_{0.1}$ OC at varying distances from roadways. A: Roadside sampling site was 37 m downwind of Interstate 5 in San Diego, CA in summer; $PM_{0.1}$ OC concentration was 1.1 µg/m³. B: Community sampling site was an urban site 400 m from a busy regional highway in Fresno, CA during the winter; $PM_{0.1}$ OC concentration was 0.07 µg/m³. C: Rural sampling site was a rural site in Westside, CA in winter; $PM_{0.1}$ OC concentration was 0.09 µg/m³. (Sources: Ham and Kleeman 2011, reprinted with permission from Elsevier; Riddle et al. 2008, reprinted with permission from the American Chemical Society.)

source using methods such as chemical mass balance that rely on the conserved chemical fingerprint of emissions.

Studies that analyze numerous measurements at a receptor site without the benefit of known emission chemistry profiles have generally yielded results that are consistent with the chemical mass analysis for UFPs. Kim and colleagues (2004) identified four dominant sources of UFPs in Seattle over a one-year study involving over 1000 measurements of particle size distributions. The lack of chemical fingerprint information made definite source identification impossible, but circumstantial evidence suggested contributions from traffic, wood burning, and secondary aerosol production from atmospheric chemical reactions.

Ogulei and colleagues (2006, 2007a,b) used a similar technique to identify primary $PM_{0.1}$ source contributions from traffic and industrial point sources in Baltimore, Maryland (2006); Buffalo, New York (2007b); and Rochester, New York (2007a), confirming that point sources can be important in regions downwind of industrial activities. In a further analysis of the Rochester data, Wang and colleagues (2011) measured a ~50% reduction in UFP concentrations when a local coal-fired power plant was converted to natural gas.

Two European studies have identified road traffic as a dominant source in Europe, using different methods to apportion sources. Pey and colleagues (2009) in Barcelona, Spain, used factor analysis and multilinear regression analysis to analyze $PM_{2.5}$ and particle number measurements (0.013 µm–0.800 µm). They attributed between 54% and 86% of UFP numbers in the 0.030–0.2 µm size range to road traffic. Lonati and colleagues (2011) used factor analysis and other methods to analyze the daily patterns of particle number size distributions in Milan, Italy; they concluded from their results that fresh traffic exhaust emissions were the primary source of UFPs in the city and were a strong contributor to their concentrations at urban background sites as well.

These studies have been important in extending sourceapportionment methods to the UFP size fraction and comparing the source contributions to different particle size fractions. However, they also illustrate that the same sources contribute to multiple size fractions at the same locations, which can complicate exposure and health studies focused on UFPs.

Particle Number Counts by Location

A third, more general approach to illustrating the effect of traffic has been to report and compare the particle concentrations observed in different locations presumed to be differentially affected by traffic. Morawska and colleagues (2008) conducted a meta-analysis of 71 studies of NCs in multiple geographic locations and showed that particle counts were progressively higher as the potential for traffic effects became greater. These findings are summarized in Figure 6 and show that mean NCs ranged from 2,600 particles/cm³ in clean background areas to 10,760 particles/cm³ in urban areas, to 48,180 particles/cm³ at roadsides, and to 167,700 particles/cm³ inside traffic tunnels where ventilation is relatively low. The relative contribution of other sources to the number concentration in urban areas cannot be assessed in these studies.

Summary of Ambient UFP Sources

Ambient UFPs have many sources, most related to combustion processes. They may be emitted directly or may also be formed from multiple precursors as part of secondary atmospheric processes. In urban areas, motor vehicles are often the leading source, particularly in proximity to roads. Source-apportionment studies indicate that other point sources may be important contributors to UFP concentrations at increasing distances from roads, and that the relative contribution may vary by geographic location, season, and time of day.

EMISSIONS OF UFPs FROM MOTOR VEHICLES

This section of the report seeks to provide an understanding of emissions of UFPs from the major classes of engine technologies, the characteristics of these particles, and how they may change over time as new technology is introduced.

In recent decades, attention has focused on the emissions associated with diesel engines. Figure 7 helps explain why. For the transport sector of the 2005 European particle number emissions inventory shown earlier in Figure 4, it shows a breakdown by engine type (Denier van der Gon et al. 2010). In almost every particle size category, diesel engines including both light-duty (passenger and light commercial vehicles) and heavy-duty (trucks, buses) accounted for a large portion of the total particle number emissions from the transport sector. Contributions to total emissions from other engine types and to a much lesser extent, from non-exhaust sources (tire and brake wear) are also projected.

However, the relative contribution of diesel engines to total and transport-related particle number emissions is likely to vary locally or regionally depending on the composition of the vehicle fleet. The work in southern California that was discussed in the previous section (Kleeman et al. 2008b; Ham and Kleeman 2011) illustrated how the source contributions to UFP concentrations can differ depending on the type of vehicles dominant on particular roads. On a regional scale, Keogh and colleagues (2009) modeled emissions contributions to PN emissions for urban southeast Queensland in Australia where they estimated that 93% of the vehicle kilometers traveled were accounted to light-duty, gasoline-powered vehicles and about 6% to



Figure 6. Mean and median particle number concentration (log scale) for different environments. The number of sites for each environment are in parentheses (e.g., 3 tunnel studies). (Source: Morawska et al. 2008, reprinted with permission from Elsevier.)



Figure 7. Estimated particle number emissions from road transport and non-road transport (railway, inland navigation, and mobile machinery) in Europe for 2005, excluding international shipping. The diesel exhaust emissions were based on the diesel fleet composition in 2005. (Adapted from Denier van der Gon et al. 2010.)

heavy-duty diesel vehicles. In that example, heavy-duty diesel engines still accounted for more than 50% of daily particle number emissions (3 to 1000 nm) but light-duty vehicles also contributed 45%. Particle number emissions from diesel and compressed natural gas (CNG) buses were comparatively much lower. Equivalent inventories of number emissions for the United States were not identified.

Generation and Characterization of UFPs in Engine Emissions

UFPs emitted from motor vehicles are primarily a product of the combustion process. They are formed in the engine during the combustion process itself as well as during the journey of the exhaust as it moves through the exhaust line and then through aftertreatment devices to the tailpipe from which it is released to the atmosphere. As the exhaust gradually cools during this journey, volatile and semivolatile materials such as organic components and ions may nucleate, thus forming new particles. This process produces UFPs in the few nanometer size range that may coagulate to form larger particles, may gradually grow in size as material condenses on their surface, or if composed solely of volatile species, may evaporate completely. However, particles formed from nucleation — sometimes called *nucleation mode* particles — are almost fully confined within the UFP size range (Kittelson 1998).

The formation and subsequent physical and chemical changes in UFPs prior to their release from the tailpipe is a function of engine characteristics and aftertreatment technology, fuel type, engine operating conditions (including state of maintenance), and ambient conditions. This section provides an overview of the typical number, size, and mass ranges of UFPs from compression-ignition (diesel-powered) and spark-ignition (gasoline-powered) vehicles, giving special emphasis to the formation characteristics of nucleation mode particles and the effects of aftertreatment technology such as diesel particle filters (DPFs) on particle numbers and composition. It also reflects on differences in emissions from light-duty and heavy-duty engines, on the effects of DPF regeneration, and finally on the implications of new fuel and technology specifications for future emissions.

Diesel Engines Particle emissions from diesel engines have been studied and characterized extensively; indeed, a large part of our understanding about the physicochemical and toxicologic properties of PM is based on studies that have used diesel emissions. Although several general conclusions can be reached about the nature of PM emissions from earlier generations of diesel engines, diesel engine technology has undergone radical changes during the last decade. Improvements in engine design and operating conditions, a combination of low-sulfur diesel fuel, and the use of highly efficient aftertreatment systems have led to significantly reduced diesel PM emissions. New diesel engines currently being sold in the industrialized countries are contributing to an important and noteworthy shift in the composition of the diesel fleet in these regions. However, during this transition, many older technology engines will remain on the road both in industrialized countries and particularly in developing countries where the introduction of the newer technology has not yet occurred. Consequently we discuss emissions from oldand new-technology engines and also describe differences between their exhaust emissions.

Emissions from Old-Technology Diesel Engines Diesel ultrafine PM is one of the most well analyzed components of vehicular PM. David Kittelson (1998) first introduced the concept of a trimodal exhaust size distribution, illustrated in Figure 1. Results from that study are representative of emissions from the older heavy-duty diesel vehicles. In particular, it showed that a large number of particles in the size range below 50 nm may be formed by nucleation, especially from engines that produce a high concentration of volatile and semivolatile components. The concentration of nucleation mode particles measured in the on-road exhaust plume can reach 10⁹/cm³ with a mean size around 10 nm (Giechaskiel et al. 2005; Kittelson et al. 2006). Such nucleation mode particles have since been observed by measurements on a number of different vehicles and sampling conditions; the many mechanisms proposed for the formation of such volatile particles have been summarized in a review by Seigneur (2009) of key studies reported in the period between 1998, when the Kittelson paper was published, and 2007.

The majority of accumulation mode particles (> 50 nm) also fall well within the UFP size range on the basis of number but not mass (see Figure 1); this distinction between size and mass distribution is particularly important to keep in mind. Particles in this mode consist of a nonvolatile agglomerate core on which volatile and semivolatile material adsorbs or condenses. The size distribution curve for typical accumulation mode particles from diesel engines without DPFs has a lognormal shape with a mean particle size of approximately 60 nm and a tail extending down to the 20–30 nm range. Peak concentrations on the order of 10^8 particles/cm³ have been observed at the tailpipe or nearby on the road (Kittelson et al. 2006; Giechaskiel et al. 2009).

While nonvolatile particles have generally been associated with the accumulation mode, a growing body of evidence suggests that they may also appear in the nucleation mode. This phenomenon has primarily been observed during engine idling (Kittelson et al. 2006) and has been attributed to metallic ash formation, primarily from lubrication oil. More recently, the presence of a combustiongenerated nonvolatile solid core in the size range below 10 nm has been recorded in emissions from heavy-duty diesel vehicles (Lahde et al. 2010) and light-duty diesel vehicles (De Filippo and Maricq 2008). There are indications from one study that such particles can be formed by sparkignition combustion as well (Sgro et al. 2008). The exact origin of these particles, the conditions favoring their production and the frequency of appearance in vehicle exhaust are not yet identified. However, the concern is that these may act as condensation sites for the formation of UFPs before or while the exhaust is diluted in the ambient air.

Emissions from New-Technology Diesel Engines In view of health and other concerns about emissions of PM from diesel engines, the United States and other industrialized countries have mandated stringent regulations to control emissions. One of the first regulatory steps taken was to reduce the levels of sulfur in diesel fuel from about 2000 ppm to 500 ppm in 1995, with further reduction to 15 parts per million (ppm) in the United States by 2006; equivalent reductions were made in diesel fuels in Europe by 2010. These changes reduced the levels of particulate sulfate emitted. In combination with the low-sulfur diesel fuel. DPFs have now been introduced. DPFs are made from ceramic or other porous materials and are generally coated with metallic catalysts; in some cases a diesel oxidation catalyst device is also positioned upstream of the DPF to enhance effectiveness (HEI 2011). DPFs have been a pivotal factor in emission reductions; they have been shown to be effective in practically eliminating particle emissions across the size spectrum, including the UFPs (Coordinating Research Council 2009; Tzamkiozis et al. 2010; Khalek et al. 2011). However, because the DPF technology is relatively new, and also because the numbers and mass of particles emitted are extremely low, relatively few studies have been done to characterize the PM emissions in detail.

Although DPFs have been highly effective at reducing particulate emissions, two issues related to the use of DPFs deserve further attention. First, several studies now suggest that under certain conditions, DPFs contribute to the formation of nucleation mode UFPs with a high fraction of sulfate (Tzamkiozis et al. 2010; Herner et al. 2011). In a detailed study, Herner and colleagues (2011) report that the emission of nucleation mode particles depends on the condition and configuration of the aftertreatment system, engine operating conditions (particularly exhaust temperature in the aftertreatment system), and the sulfur content of fuel and lubrication oil (see also Hesterberg et al. 2011).

Secondly, high-efficiency particle filters can become loaded with soot particles, which must be removed to prevent plugging. Removal is done by oxidizing the collected soot particles in place in a process called regeneration. During regeneration, transient high UFP emissions have been observed (Bergmann et al. 2009; Khalek et al. 2009, 2011). For example, in the testing of new-technology diesel engines described above, most of the UFP emissions were confined to the regeneration phase which generally lasted 30 to 45 minutes (Khalek et al. 2011). Still, the use of modern aftertreatment technologies represents a very important advance in reducing diesel emissions and are expected to improve air quality.

Comparison of Emissions from Old-Technology vs. New-Technology Diesel Engines As mentioned above, the PM emissions from old- and new-technology diesel engines are different in several respects; these differences are discussed below and summarized in Figure 8.

Mass: The PM mass emitted from the new-technology engines is far lower than that from the old engines. For example, comparing the results of two series of tests, with 2004 engines (Coordinating Research Council 2007) and 2007 new-technology engines (Khalek et al. 2011), the mass of emitted total PM is reduced by 89% (Coordinating Research Council 2009). Other authors have reported similar reductions.

Number emissions: Based on a comparison of two series of tests reported in Coordinating Research Council 2007 and 2009, the number of particles emitted by the new-technology (2007 model year) vs. the old-technology (2004) engines is lower by more than 100-fold; during regeneration events, when most of the PM is emitted, the particle numbers are still 10-fold lower as compared to the 2004 engine tests.

Chemical Composition: New-technology diesel engines also have a significant effect on the chemical composition of diesel emissions (Maricq 2007; Biswas et al. 2009; Coordinating Research Council 2009, Hesterberg et al. 2011). PM from old-technology engines contains significant amounts of both OC and EC (with the latter being in excess), along with sulfate, metals and other ions. DPFs reduce the mass of EC present in solid particles, reduce volatile components that contribute to OC, and render the proportion of OC greater than EC. Also, the relative proportion of sulfate in the particles is now higher than it is in old-technology engine emissions. DPFs also effectively (> 95%) reduce the metal content of PM exhaust over typical driving cycles (Hu et al. 2009; Cheung et al. 2010).



Figure 8. Comparison of the mass, numbers, and composition of emissions from old-technology and new-technology diesel engines. (Sources: 1998 data from Khalek, Personal Communication 2012; 2007 and 2009 data from Coordinating Research Council 2007, 2009.)

To meet the stringent 2010 nitrogen oxides (NO_x) standards established by the EPA and similar rules in Europe, manufacturers of diesel vehicles have developed methods for selective catalytic reduction (SCR) of NO_x compounds. Some SCRs use vanadium (as V_2O_5) in the catalytic formulation. However Hu and colleagues (2009) have shown that vanadium-containing SCR systems, may release vanadium in the UFP size range if the temperature of the exhaust is too high. In the United States, vanadium-based SCR are used for agricultural and other nonroad applications, but on-road vehicles use copper-zeolite catalyst, which is stable at higher temperatures. In Europe for heavy-duty applications (and increasingly in developing countries), vanadium-based catalysts are more common.

Spark-Ignition Engines Gasoline spark-ignition engines, provided that they are well maintained, produce only small amounts of PM under normal operating conditions. Recently, in response to the call for increased fuel efficiency, gasoline direct injection or direct injection spark ignition (DISI) technology is being widely adopted. While helping to boost fuel economy, however, DISI engines produce increased numbers of UFPs. Concern about the increased UFP emissions from DISI technology has led to the development of alternative fuel injector designs; the spray-guided, center mounted injector appears to be particularly promising in reducing particulate mass and number emissions.

Conventional Gasoline Engines The dominant method used to introduce fuel in the combustion chamber of gasoline engines has been port injection. Under normal operating conditions in modern gasoline engines, negligible numbers of UFPs are formed. Conditions do exist, some transient and some longer term, under which these engines may be significant contributors to UFP emissions. During start up under cold temperature conditions (cold starts), UFP emissions can increase, as can emissions of other gaseous pollutants. For example, Mathis and colleagues (2005) showed that gasoline particle number may reach diesel-like levels in tests at -7°C and -23°C. Such emissions are clearly of greater concern in colder climates where they may be important contributors to UFP concentrations under certain conditions.

Poor engine maintenance can also lead to increased emissions of UFPs. Light-duty gasoline vehicles with visible smoke emitted from their tail pipes due to engine malfunction (so-called *smokers*) may be significant contributors due to emissions from partial combustion of lubrication oil. Robert and colleagues (2007) estimated the UFPs emitted by smokers to be 10 mg/km compared with 0.05 mg/km for modern gasoline cars operating on mild driving cycles, and with 2 mg/km for cars operating on more aggressive driving cycles. In gasoline PM, OC dominates the mass of UFPs, followed by EC with traces of ions, including calcium, ammonium, sulfate, and various metals (Geller et al. 2006; Robert et al. 2007). Typically, gasoline UFPs contain a higher fraction of heavy polycyclic aromatic hydrocarbons (PAHs) than diesel exhaust (DE) which may have implications for the differential toxicity of these particles (Geller et al. 2006; Cheung et al. 2010).

Gasoline Direct Injection Engines The direct injection of fuel into the cylinders of gasoline engines is increasingly being used because it improves fuel efficiency and performance. The gasoline DISI provides better control of the airto-fuel ratio, especially while starting an engine and during warm up. Another important feature of the DISI is that it allows the use of a higher engine compression ratio, made possible because of cooling of the contents of the combustion cylinder as the direct-injected fuel spray evaporates. Because of the less complete mixing of fuel vapor and air, however, the particulate emissions of the engine increase, including the number of UFPs (HEI 2011).

Studies have shown the size distribution of particles emitted by DISI engines is similar to that emitted from non-DPF diesel engines (Harris and Maricq 2001), and the particle numbers emitted can sometimes be only 4–5 times lower than typical non-DPF diesel cars (Ntziachristos et al. 2004). The high particle number of DISI vehicles has raised questions about the need to install particle filters in DISI vehicles; it has also led to an intense interest in fine tuning the injection–combustion control system that can reduce the UFP emissions problem. The development of spray-guided fuel injectors shows a great deal of promise in this regard.

Other Types of Combustion Engines Small gasoline engines, in particular two-stroke engines installed in mopeds and scooters, are a significant source of particle emissions. In such engines, rich combustion, early scavenging of combustion products, in-cylinder injection of lubricant oil, and poor maintenance lead to elevated hydrocarbon emissions which may condense to form UFPs. UFP emissions from such engines have been shown to exceed typical diesel engines by more than one order of magnitude (Ntziachristos et al. 2005). The air quality implications of such high UFP emissions are of particular importance in developing countries where a significant population of such vehicles is still in operation (Begum et al. 2006).

Off-road machinery, such as construction equipment, diesel power generators (especially those used in developing countries), lawn mowers, and marine engines, has been a significant source of pollution in certain locations. Among the industrialized countries, regulations are gradually being introduced to control such sources. However, the smaller of such sources — such as lawn mowers and other gardening equipment used by large numbers of consumers — are still not well controlled and may be a source of PM exposure. The situation in developing countries also deserves much attention.

Non-Exhaust Sources

Vehicles also generate PM through mechanical processes, namely wear on tires and brakes, abrasion of road surfaces, and resuspension of road dust. The particle number emissions inventory, displayed in Figure 7, projects a contribution to total 2005 UFP emissions from tire and brake wear (but not including road-surface wear or resuspension) that is substantially smaller than that from exhaust, on the order of a few percent at most depending on particle size (Denier van der Gon et al. 2010).

A laboratory study of debris released during braking events found that mechanical processes generally resulted in a distribution of particles with mean diameters in the range of a few micrometers but with tails extending to the UFP range; the number and size distribution varied with brake materials and other factors (Sanders et al. 2003). In addition to mechanically-generated particles, there has been some evidence of thermal particle formation due to the heat produced in cornering and braking (Dahl et al. 2006; Kukutschova et al. 2011). These particles have not been well studied to date.

Vehicles contribute to an increase in ambient PM through abrasion of road surfaces and resuspension of road dust in the vehicle wake. However, the particles that accumulate on the road are in the micrometer size-range or larger. Smaller particles are not likely to settle on the road surface, but they are scavenged out of the atmosphere by precipitation or by photochemical reactions. Hence, the contribution of particle resuspension to UFP concentrations is likely to be negligible, although measurements to confirm this assumption are necessary.

Finally, in old engines, crankcase emissions have also been a source of UFPs (Rim et al. 2008; Tatli and Clark 2009). These should have been effectively addressed with the post-2007 emission standards that have implicitly called for closed-type crankcase breathing systems.

Potential Effects of Other New Fuels and Technologies

The development of new fuels and technologies is progressing rapidly in response to various pressures for alternative fuels, the need for greater fuel economy, and efforts to reduce air pollution. Such changes in fuels and technologies are likely to affect both overall emissions and the relative contributions from different vehicle classes. However, these effects are not yet well characterized. Buses Powered by CNG UFP emissions from CNG buses have been studied in some detail in an effort to understand whether CNG buses can be a successful alternative to diesel buses with advanced aftertreatment systems. Particle number emissions of CNG buses are typically one order of magnitude lower than those of diesel buses without DPFs at low loads, but the CNG bus emissions can reach diesel-like concentrations during acceleration and at high load (e.g., Nylund et al. 2004; Javaratne et al. 2009, 2010). In all cases, particle mass is a fraction of total PM emissions, implying that these particles are in the UFP size range. It appears that emissions largely result from lubricating oil consumption and are affected by catalyst location and the engine type (lean-burn versus stoichiometric). It has also been shown that, similar to spark-ignition engines, OC dominates the mass of particles from CNG buses; EC and inorganic species make up some 30% of the total mass (Okamoto et al. 2006).

Biodiesel Use of biodiesel is seen as one part of a multipronged approach to reducing greenhouse gas emissions from transport, but the implications for UFP emissions deserve some attention. Biodiesel is a synthetic fuel derived from plant or animal products and blended with fossil diesel fuel at low percentages (10% to 20% by volume, sometimes lower in Europe). These blends generally produce lower emissions of total PM mass. Under certain conditions, however, and even in low blending ratios, biodiesel may enhance the formation of nucleation mode particles (Heikkila et al. 2009; Fontaras et al. 2010; Chuepeng et al. 2011). The UFPs produced contain reduced amounts of EC as a result of decreased soot formation and enhanced in-cylinder oxidation of any particles formed (Jung et al. 2006; Hoekman et al. 2009).

Ethanol Gasoline containing 10% ethanol is widely sold in the United States, and the use of ethanol is poised to increase in the coming years with higher blend levels. The blending of ethanol in fuel can lead to higher volatility of the blended fuel. To compensate for this, lower volatility base gasoline is usually used, which could contribute to less homogeneous mixing of fuel and air in the combustion chamber and poor evaporation, and in turn to higher UFP emissions. Data concerning the effect of ethanol on UFP emissions is limited; however, no substantial variations of the already low UFP emissions of gasoline-powered vehicles appear likely (Lee et al. 2009). After extensive modeling, the U.S. EPA (2010) concluded that the use of ethanol blends will lead to only minor changes in annual or daily PM₁₀ or PM_{2.5} concentrations; the effects on UFPs were not specifically evaluated. Emissions of PM may also arise from poor maintenance of vehicles (e.g., erroneous

recognition of the petroleum/ethanol blend ratio could lead to nonstoichiometric combustion with significant effects on releases of UFPs).

Electric Drive Technologies Hybrid electric vehicles, plug-in hybrid electric vehicles, battery electric vehicles, and fuel cell vehicles have started to be marketed in many countries during the last few years. With tailpipe emissions at least comparable to the cleanest available gasoline engines (in the case of hybrid electric vehicles and plug-in hybrid electric vehicles) to zero emissions (in the case of battery electric vehicles and fuel cell vehicles), the increasing employment of such vehicles in the market should eventually have a net beneficial effect on all traffic-related emissions, including those of UFPs.

Summary of Motor Vehicle UFP Emissions

UFPs from motor vehicles are emitted primarily in exhaust from internal combustion engines. Non-exhaust sources such as mechanical wear on tire and brakes, abrasion of road surfaces, and resuspension of road dust have begun to receive some attention, but their contributions to emissions of UFPs have not been extensively studied.

Diesel engine technology, in particular, has historically favored the formation of particles in the ultrafine range, so their emissions and the factors giving rise to them have been extensively characterized. Changes in the sulfur content of diesel fuels, optimization of engine design and operating conditions, and the use of modern aftertreatment technologies have led to substantial reductions in diesel engine UFP emissions as well as to significant changes in their chemical composition. However, emissions of UFPs during DPF regeneration events and under other engine operating conditions deserve further attention.

Well-maintained gasoline spark-ignition engines using conventional port injection technology produce little UFP or other PM emissions under normal operating conditions. Newer fuel-efficient gasoline engines using DISI technology have been found to release UFPs in similar size ranges as diesel engines, but at a lower rate of emissions. Optimization of engine design and operating conditions to reduce emissions are being pursued, and the need for DPFs is being considered.

These two engine technologies, while they currently dominate the automotive fleet and consequently receive the most focus, are not the only types of combustion engines that contribute to UFP emissions. Rapid changes are occurring in fuels and in technologies that are likely to affect overall emissions, the relative contributions from different vehicle classes, and the relative importance of non-exhaust sources. The collective impact of all these changes on either overall emissions or ambient concentrations has not been thoroughly explored and is likely to vary regionally depending on the rate and extent to which they are deployed in different parts of the world.

CHARACTERIZING HUMAN EXPOSURE TO AMBIENT UFPs

The characterization of the sources and emissions of UFPs is an important first step. However, understanding the potential implications of these emissions for human health requires characterization of potential human exposures — how the concentrations and composition of UFPs vary over time and in the different locations where people live, travel, and work. Such information helps inform the design of relevant exposures for use in experimental settings with animal and human subjects (Chapter 3). Epidemiologic study designs take advantage of these variations in ambient concentrations to explore their implications for human health (Chapter 4). In the case of PM_{10} and $PM_{2.5}$, such studies have played key roles in determining the numerical levels at which ambient standards are set.

Providing a comprehensive characterization of UFP concentrations is challenging because no networks of UFP monitors currently exist. Instead, studies often provide a snapshot of specific locales at particular points in time, often relying on different monitoring methods. They may, or may not, include measurements of other pollutants, including other particulate size fractions and gaseous copollutants, that may ultimately be needed to understand more specifically the role of UFPs and their sources on health. To augment measurement data, various efforts to model UFP concentrations are also under development.

This section of the document provides an overview of what these studies tell us about how ambient UFP concentrations vary over time and space, in particular in relation to traffic in urban areas. Like the majority of studies, this summary focuses variation in measures of NC with some data on differences in particle mass and composition.

Factors Affecting Concentrations and Composition of Ambient UFPs

Numerous processes influence the concentrations and composition of ambient UFPs over different spatial and temporal scales. At a very local scale, Figure 9 schematically illustrates the typical evolution of an exhaust aerosol packet immediately before and after it leaves the tailpipe of a diesel vehicle not equipped with DPFs. It describes the processes leading to changes in the size distribution and dilution ratio with increasing distance from the tailpipe until particles merge into the urban background.

Once emitted directly to the atmosphere or nucleated in the cooled exhaust from combustion sources such as motor vehicles, UFPs undergo coagulation and gas-particle exchange with the surrounding atmosphere (Zhang et al. 2004). Coagulation (particle collision and adherence) favors the transfer of the smallest UFPs to the larger size fractions, usually with diameters > 100 nm, over timescales of a few hours. This process can be an effective atmospheric removal mechanism for primary UFPs, which have very low settling velocities (Herner et al. 2006). Gasparticle exchange (condensation or evaporation) favors the growth or shrinkage of particles in the cooled combustion exhaust depending on the concentration of the surrounding gas-phase material over timescales of seconds to minutes (Zhang et al. 2004). As dilution with ambient air cools the exhaust, the gas-phase material initially becomes supersaturated, leading to nucleation and growth of semivolatile organic compounds. Continued dilution reduces the gas-phase concentration below the saturation level, causing the nucleated particles to evaporate completely or leaving the solid primary cores of UFPs that previously acted as condensation sites.

UFP concentrations beside busy roadways also depend strongly on emissions patterns, but the diurnal or seasonal cycle of temperature can strongly modify UFP NCs (Charron and Harrison 2003; Kuhn et al. 2005). Lower ambient temperatures favor the formation of greater numbers of the smallest particles (< 50 nm) in the roadside environment (although these particles may evaporate completely within 300 meters downwind of roadways (Zhu et al. 2004; also discussed below). Relatively low temperature and high humidity are associated with higher rates of new particle formation and slower atmospheric dispersion, indicating that UFP concentrations will generally be higher in the winter than in the summer (Sioutas et al. 2005).

Lower temperatures near the ground at night also contribute to the formation of stable atmospheric layers that trap primary pollutants near their emissions source (Herner et al. 2006); this effect can dominate UFP concentrations in regions that are not heavily influenced by photochemistry. As an example, the highest concentrations of UFP number and mass during a winter pollution event in the San Joaquin Valley were measured during the evening hours, with lower concentrations measured during the day (Herner et al. 2005).

When photochemistry is important at a location, the opposite diurnal pattern is often observed. Numerous studies have observed that total NCs are positively correlated with ozone (O_3) concentrations during the summer period, suggesting that the highest number concentrations occur on the warmest days (Sioutas et al. 2005).



Figure 9. Typical evolution of an exhaust aerosol packet immediately before and after it leaves the tailpipe of a diesel vehicle not equipped with DPF. The initial size distribution is engine and operation condition dependent. Also, the exact time evolution of the size distribution (illustrated in columns I through V) and the dilution ratio (red line) will depend on the exhaust, traveling, and ambient conditions. In general, five phases are observed: I) A lognormal distribution of nonvolatile particles is produced in the engine and leaves the tailpipe. II) Rapid dilution with ambient air takes place that decreases the concentration of non-volatile particles. Depending on traveling speed, dilution ratio can reach 102:1 up to 104:1 during the first second after emission. In parallel, a nucleation mode of volatile particles forms in the sub-50 nm size range. III) Further dilution downwind of emission production takes place that decreases concentration of volatiles and leads to mild evaporation of volatile nanoparticles. IV) Particle concentration almost reaches background levels and nanoparticles have almost completely disappeared. V) A new (secondary) nanoparticle mode may be formed as a result of photochemical reactions.

Physical geography, such as topography and altitude also influence dispersion; low-lying valleys collect PM, and high elevations have greater atmospheric dispersion (Sardar et al. 2004; Zhou and Levy 2007). Urban street canyons are subject to low wind speeds and poor mixing during most times of the day, so UFP concentrations at these locations are dominated by a diurnal cycle of traffic emissions. The higher concentrations in street canyons relative to those near roads or in urban background sites were evident in the meta-analysis by Morawska and colleagues (2008) discussed earlier (see Figure 6).

Scientists have expressed increasing interest in regionalscale *nucleation events* where large numbers of particles can be formed across distances of hundreds of kilometers through nucleation processes similar to those discussed in the context of vehicle exhaust as it cools near roadways.
Nucleation events and their potential contribution to ambient UFP number concentration levels and human exposures have not been a focus of this issue of HEI Perspectives. They are discussed briefly in Sidebar 1.

Spatial Variation of Ambient UFP Concentrations

As the schematic illustration of the fate of a diesel aerosol packet in Figure 9 would suggest, observed spatial gradients of UFPs in the atmosphere are sharp, with the highest concentrations generally observed in the immediate proximity of combustion sources followed by a rapid decay. Zhu and colleagues (2002) were among the earliest investigators to monitor the change in UFP numbers and size distributions with distance from major freeways. UFP measurements were taken near a major interstate highway (freeway 710) in Los Angeles, California, where approximately 25% of the traffic came from heavy-duty diesel trucks. Figure 10 illustrates the highest NCs of the smallest sized UFPs nearest the road, followed by a rapid drop-off in concentration, transition to larger particles with increasing distance, and blending into background levels at approximately 300 meters. Subsequent studies by these investigators and others have shown that these gradients can show diurnal and seasonal patterns, for example, with the distance required to reach background extending to 500 meters or more under nighttime conditions (Zhu et al. 2006).

Karner and colleagues (2010) have now conducted a meta-analysis of 41 studies that evaluated gradients in UFPs and other traffic-related pollutants as a function of distance from roadways. Their analysis is notable because of their efforts to normalize concentrations to account for differences among studies in background concentrations (background normalization) and the distances from the edge of road at which measurements are made (edge-of-road normalization). Furthermore, the authors analyze and compare the concentration gradients for several particulate size fractions (UFPs > 3nm [UF1], > 15 nm [UF2], PM_{2.5}, and PM₁₀) as well as for several other traffic-related pollutants measured in the same studies (carbon monoxide [CO], EC, benzene, nitric oxide [NO], nitrogen dioxide [NO₂], NO_x, and VOCs).



Figure 10. Ultrafine particle size distribution at different sampling locations near the 710 freeway in Los Angeles, CA. (Source: Zhu et al. 2002, reprinted with permission from Elsevier.)

Sidebar 1. Regional Nucleation Events

Regional nucleation events differ from near-roadway nucleation events because the super saturation of semivolatile compounds across the regional events is driven by the buildup of chemical reaction products rather than by the cooling of hot exhaust gases near the roadway. On a global basis, particle nucleation events are best known as an important source of cloud condensation nuclei that influence cloud properties and climate. On a regional and urban scale, nucleation can significantly increase the NC over large population centers (Stanier et al. 2004; Cheung et al. 2012). In the Stanier study for example, conducted in Pittsburg, Pennsylvania, nucleation events were observed on ~30% of the days during an extended study period.

Despite the common occurrence of regional nucleation events, the mechanisms that control nucleation rates and the chemical composition of nucleated particles are poorly understood. Venkatachari and colleagues (2007) have studied nucleation events in Flushing, New York, and suggest that concentrations of reactive oxidative species were higher in the submicron fractions relative to larger particles. Recent results from the European Integrated project on Aerosol Cloud Climate and Air Quality suggest that sulfuric acid plays a central role in most nucleation events, although some other stabilizing compound such as ammonia or amines must also play a role.

Since most epidemiologic studies have assessed associations between total ambient UFP NC and health, the specific implications of regional nucleation events, independent from other ambient UFP sources, for public health are not yet known. No epidemiologic studies have been done to isolate the effect of nucleation events on short- or long-term exposures and on health.

Figure 11 displays their results, which essentially show the percentage decrease in the concentrations of several pollutants from the roadway edge to various distances up to 500 meters. It confirms the rapid decline in the concentrations of the smallest UFP size fraction (UF1) within the first 100 meters with a more gradual decline in the UF2 size fraction to near background levels at distances of over 500 meters.

The comparisons with other pollutants are useful for indicating those pollutants whose decay patterns are similar to those of UFPs and therefore are more likely to be correlated with one another. Like the different UFP size fractions, several of the other pollutants (CO, EC, NO, NO₂, NO_x, VOCs) showed steep declines in the first 100 meters from the road. $PM_{2.5}$ and PM_{10} appear somewhat elevated nearer to roads, but generally appear to be much less spatially variable and more representative of background levels. Correlations between pollutants that show similar patterns of decay are thus higher than between those that are distributed differently (see, for example, Kaur and colleagues' study [2005] of personal exposure to UFPs, CO, and $PM_{2.5}$ at an urban intersection in London).

Given the steep gradients in UFP concentrations near sources like traffic, substantial spatial variation in UFPs can exist across a single city. In a study designed specifically to compare variation in different particle metrics, Puustinen and colleagues (2007) measured PM_{10} , $PM_{2.5}$, and total NC (using CPC) outdoors and indoors at a total of about 150 home sites spread across four European cities (Athens, Amsterdam, Birmingham, and Helsinki). Figure 12 compares variation in 24-hour average total NC and $PM_{2.5}$. Although variation was observed in both total NC and $PM_{2.5}$ across sites in the individual cities, the degree of variation tended to be greater for total NC than for $PM_{2.5}$.

The high degree of spatial variation in UFP concentrations poses both an opportunity and a challenge for scientists trying to represent population exposure to UFPs for health studies, particularly for longer-term average exposures. On the one hand, variation in concentrations of a pollutant is essential to investigate whether pollutant exposures may be related to variations in health outcomes. On the other hand, the high spatial variability makes it more difficult to rely on measurement strategies that have been adequate for more spatially homogenous particulate fractions like PM_{2.5}. PM_{2.5} measurements taken at different locations around a city are usually better correlated with one another than are measurements of spatially heterogenous particulate fractions; exposure can therefore more reliably be represented by a city-wide average or by a central site monitor over the longer term.

For example, when the correlations between the central site and residential outdoor 24-hour concentrations of $PM_{2.5}$ and NC were compared for all sites within each of the four cities shown in Figure 12, the correlations for $PM_{2.5}$ were generally higher (city medians: 0.79–0.98) and less variable across the sites than those for total NC (city medians: 0.67–0.76) (Figure 13). For total NC, median correlations varied considerably by residential site. The greater variability in correlations for total NC measurements suggests the potential for greater error in the degree to which central site measurements represent individual exposures for UPFs relative to $PM_{2.5}$. This greater measurement error



Figure 11. Local regression of road-edge normalized concentrations on distance from the edge of road. The horizontal black lines indicate reductions of 50% (0.5) and 90% (0.1) from the concentrations measured at the edge of the road. The regression sample size, *n*, is given in parentheses after each pollutant. (Source: Karner et al. 2010, reprinted with permission from the American Chemical Society.)

can limit the statistical strength of epidemiologic studies to observe any true associations that might exist. Either many more monitoring sites, or a reliable modeling strategy, would be necessary to characterize the UFP concentrations experienced in a population across a city, particularly over the longer term. **Temporal Variations** Even in the case of strong differences in absolute concentrations among sites (i.e., spatial variability), particular geographic locations are influenced by common diurnal patterns and meteorological influences. Consequently, measurements at those locations may be temporally correlated. High temporal correlations among



Figure 12. Distribution of 24-hour average central site (left box plot) and residential outdoor concentrations (right box plot) of total NC and PM_{2.5}. The center line of the box is the median, the dotted line is the mean. The outer lines of the box represent the 25th and 75th percentiles, and the whiskers represent the 10th and 90th percentiles. (Source: Puustinen et al. 2007, reprinted with permission from Elsevier.)



Figure 13. Distribution of individual Pearson correlation coefficients of 24-hour central site and residential outdoor concentrations for total NC and PM_{2.5}. The center line of the box is the median, the outer lines of the box represent the 25th and 75th percentiles, and the whiskers represent the 10th and 90th percentiles. (Source: Puustinen et al. 2007, reprinted with permission from Elsevier.)

monitoring sites indicate that ambient fixed-site monitoring may be adequate for estimation of population exposure in study designs that examine the effect of short-term changes in air quality.

The study by Puustinen and colleagues again provides a good example of high temporal correlations between sites using data from one of their study sites, a residence at an urban background site in Helsinki. Figure 14 compares the hourly variation in total number concentrations measured outdoors and indoors at their study site with those measured at the central monitoring site over the course of one week in January 2004. Other studies also suggest that correlations between monitoring sites can vary more from location to location. Cyrys and colleagues (2008) reported high correlation coefficients (r > 0.8) for site-to-site hourly average measurements among four traffic-affected sites in Augsburg, Germany. However, Moore and colleagues (2009), who measured total NC over a period of about one year at 14 sites in the Los Angeles area of the United States, found that the median hourly correlation coefficient across all sites varied from 0.3-0.56. They reported a 10-fold variability in hourly UFP count measurements (10,000-90,000) calculated by month. Tuch and colleagues (2006) reported a correlation of 0.31 between two locations (one roadside, one mixed industrial area) that are 1.5 km apart, across all days, in Leipzig, Germany.

Given the site-specific nature of these correlations, and their implications for how well human exposure may be



Figure 14. Illustrative example of hourly total NC measured indoors, outdoors, and at a central site for one week in January 2004 for an urban background study site in Helsinki, Finland. The correlation between hourly residential outdoor and central site concentrations was 0.89, although the concentrations were lower at the home than at the central site (ratio 0.37). (Source: Puustinen et al. 2007, reprinted with permission from Elsevier.)

measured, such correlations need to be carefully evaluated when designing studies and when reporting and interpreting study results.

Copollutant Concentrations Given their sources, UFPs are typically found in the presence of a number of other pollutants of interest to human health (e.g., CO, NO, NO_x , NO_2 , EC, $PM_{2.5}$, PM_{10}) as illustrated in Figure 11. Understanding the spatial and temporal relationships between UFPs and the other pollutants with which they may covary is critical for efforts to assess their independent effects. However, copollutant exposures have not been consistently measured or reported in studies.

Some authors have suggested that NO_2 may be acting as a surrogate for other harmful pollutants in the traffic pollution mixture, including UFPs, based on associations of within-city NO_2 concentrations and adverse health effects in some epidemiologic studies (Seaton and Dennekamp 2003; WHO 2006). There is some evidence to suggest relatively high correlations between UFPs and NO_x (Sardar et al. 2004; Vinzents et al. 2005; Andersen et al. 2008b). While the recent HEI Special Report on Traffic (2010) cautioned that none of the traffic-related pollutants evaluated (including NO_2 , UFP, CO, EC, or black carbon [BC]) met all the criteria for an ideal surrogate for traffic), it is possible that observed associations between health effects and spatial patterns in NO_2 related to traffic also reflect spatial patterns in traffic-related UFPs.

Consistent with the evidence provided by Karner and colleagues (2010), results from several studies seem to suggest that UFPs and PM_{2.5} can be governed by different processes, so their concentrations are less likely to be well correlated. Investigators interested in the impact of local traffic restrictions on air quality in New York City found that, in contrast to PM_{2.5}, near-road NC (for particles 5–560 nm in diameter) varied linearly with measures of traffic flow, suggesting that they were highly influenced by traffic sources (Whitlow et al. 2011). Atmospheric processing may lead to inverse correlations between UFP concentrations and PM_{2.5} concentrations, as coagulation and condensational growth moves material from the UFP size range to the accumulation mode size range over time (Chung et al. 2001; Herner et al. 2006). PM_{2.5} concentrations increased by a factor of three during a winter stagnation event in central California, but UFP number and PM_{0.1} mass concentrations remained relatively constant (Herner et al. 2005; Kelly et al. 2011).

Microenvironmental Exposures to UFPs

The preceding sections have focused on how well spatial and temporal patterns in ambient concentrations of

UFPs, often measured at some central location, represent those that individuals might experience at their homes. Such central site ambient measurements are the most common indicators of exposure used in epidemiologic studies. However, scientists know that an individual's total personal exposure to any air pollutant is actually a function of microenvironments, the places where people spend time during the day in which the air pollutant concentrations may differ (e.g., at home, at work, during a commute). It is also well known that concentrations in individual microenvironments can have origins both in ambient air and within the microenvironment. Though sources or precursors of UFPs within microenvironments can be substantial (Abt et al. 2000; Diapouli et al. 2008; Guo et al. 2010; Wang et al. 2010; Hovorka and Braniš 2011), and may themselves merit evaluation in health studies, they are not the focus of this document. We have focused this discussion on whether concentrations measured outdoors are in fact a good representation of human exposure to particles of ambient origin and the circumstances under which they may fall short. Such insights are important for the interpretation of health studies that may rely solely on outdoor measurements.

Ambient Contributions to Indoor Concentrations of

UFPs Indoor microenvironments (e.g., home, work, schools, stores, etc.) are some of the most important determinants of personal exposure simply by virtue of the time we spend in them. Most people in the United States and in Europe spend a large fraction (90% or more) of their time indoors.

One broad approach to understanding how well ambient measurements may represent indoor exposures to particles of ambient origin is to evaluate whether the variations in ambient air concentrations are temporally or spatially correlated with those of indoor concentrations. A study by Hoek and colleagues (2008) is one of the few studies that has systematically examined such relationships for UFPs, along with other pollutants, inside a large number of homes in multiple cities. Using the same dataset as Puustinen and colleagues (2007) for 152 homes in four European cities (Amsterdam, Athens, Birmingham, and Helsinki), Hoek and colleagues analyzed the 24-hour correlations between indoor and central site concentrations for particle number, $\mathrm{PM}_{2.5},$ soot, and sulfate over a one-week period. They reported that correlations were lower on average for particle number (0.18-0.45) than they were for the other pollutants (PM_{2.5} [0.40-0.80], soot [0.64-0.92], and sulfate [0.91-0.99]), a finding that the authors suggested might be related to the higher spatial variability in ambient UFPs discussed earlier, as well as to the lower infiltration of UFPs and to the presence of indoor sources.

The results of Hoek and colleagues (2008) can be interpreted with the help of a number of other studies that have evaluated indoor and outdoor particle NC relationships, including estimating infiltration and the effect of indoor sources. These studies have reported moderately high infiltration fractions for UFPs and have noted that infiltration varies with particle size. Zhu and colleagues (2005) measured indoor UFPs in four apartments near a major freeway in Los Angeles and reported low (0.1-0.4) infiltration fractions for the smallest (10–20 nm) particles but moderate to high infiltration fractions (0.6–0.9) for 70–100 nm particles. Sarnat and colleagues (2006a) described a similar sizerelated pattern of infiltration fractions in 17 homes of nonsmokers in Los Angeles. These results are consistent with a study in 4 homes of nonsmokers in Boston, where Abt and colleagues (2000) reported Spearman correlations between home indoor and outdoor concentrations of 0.67 for 20-100 nm particles, 0.90 for 100-500 nm particles, and 0.83 for 700-2500 nm particles. The indoor:outdoor UFP ratios estimated from measurements in seven primary schools in Athens, Greece, ranged from 0.33 to 0.74 and were lower in general that those for PM₁₀ and PM_{2.5} (Diapouli et al. 2008). Studies have also suggested that the composition of particles that infiltrate to the indoors may also differ from that of outdoor particles (Sarnat et al. 2006a; Polidori et al. 2007).

Other factors can strongly influence estimated UFP infiltration rates including: air exchange or ventilation rates within buildings, presence of local outdoor sources, wind speed, season, numbers of occupants, and time of day (which may be related to indoor activities that generate particles, like cooking) (Koponen et al. 2001; Sarnat et al. 2006a; Polidori et al. 2007; Guo et al. 2008; Hoek et al. 2008; Parker et al. 2008; Weichenthal et al. 2008; Wang et al. 2010).

Collectively, these factors help explain why the relationships between outdoor and indoor concentrations of UFPs are more variable and the correlations generally lower than they are for $PM_{2.5}$ and other pollutants. They need to be considered carefully when interpreting the results of epidemiologic studies based on ambient measurements.

In-Vehicle Exposures Given the high concentrations of UFPs reported on or near roads, a large number of studies have evaluated concentrations in vehicles and as a function of mode of transport. For example, Westerdahl and colleagues (2005) reported in-vehicle total NC measurements in Los Angeles of 55,000–200,000 on freeways, 40,000 on arterial roads, and 14,000–27,000 in residential areas (background) averaged over several hours. Concentrations measured while following diesel vehicles resulted in peak concentrations of up to 800,000. In-vehicle concentrations

were also strongly dependent on the number of vehicles in front of the measurement vehicle at intersections.

Based on these and other measurements, Fruin and colleagues (2008) conducted a microenvironmental analysis and estimated that 33%–45% of total UFP exposure for Los Angeles residents was due to time spent traveling in vehicles (Fruin et al. 2008). In another study of Los Angeles freeway exposures, Zhu and colleagues (2007) estimated that a 1-hour commute accounted for 10%–50% of daily exposure to traffic-generated UFPs. Further evaluation is required to determine whether these study results of the importance of commuting exposures in the hightraffic areas around Los Angeles are representative of commuting exposures in areas with less traffic or shorter commuting times.

In-vehicle UFP concentrations can be affected by a number of factors, including temperature, wind speed, traffic counts, and numbers of passengers (Gong et al. 2009; Knibbs et al. 2011) as well as vehicle ventilation (and filtration). Standard automobile filters result in reductions of between approximately 30% and 60% in invehicle UFP concentrations (Zhu et al. 2007; Pui et al. 2008; Qi et al. 2008), while these percentages can be increased with advanced filters (Burtscher et al. 2008). Zhu and colleagues (2007) found that the lowest in-vehicle concentrations were observed (~85% reduction) when fans were operated on recirculation mode, and that standard filters provided reductions of ~50% for the smallest particles (7–40 nm), but that this decreased to ~20%–30% for particles in the 40–200 nm size range.

In a study in the Netherlands designed to examine the effects of transport method (car, bus, bicycle), route (high and low traffic), and fuel type (gasoline, diesel, and electric) on commuter exposures to total particle numbers and other pollutants, Zuurbier and colleagues (2010) reported no significant differences between concentrations of particle numbers inside diesel and gasoline automobiles. They suggested that this result may be a reflection of the ambient environment surrounding the automobiles and infiltration of UFPs rather than self-pollution. Commuter exposures to total particle NC in this study were lowest among those riding electric buses. Exposures via all modes of transport were elevated when following high-traffic routes.

Knibbs and colleagues (2011) conducted a meta-analysis of 47 in-transit studies to assess the differences in microenvironmental exposures experienced using different modes of transit (e.g., travel by bicycle, automobile, walking, ferry, rail, automobile tunnel). They reported overall tripweighted mean UFP concentrations to be lowest for bicyclists (34,000 particles/cm³) and highest when riding in an automobile in a traffic tunnel (300,000 particles/cm³).

Implications for Other Countries

Most of the studies of UFP concentrations reported above have been performed in locations in the United States or Europe; they may not be representative of concentrations in other countries where the mixture of vehicle type, emission controls, and fuel composition are different. For example, Lung and colleagues (unpublished data, 2005) measured exposures of pedestrians standing at intersections in Taiwan and found variable, but much higher concentrations (123,639 particles/cm³) than reported elsewhere. Recently, Apte and colleagues (2011) studied PM levels on the roads of Delhi, India, which have a large number of auto-rickshaws. Trip-averaged concentrations were about 280,000 particles/cm³, which corresponded to about eight times the ambient levels. Peak concentrations of 800,000 particles/cm³ were measured over a 10-second interval.

Modeling UFP Concentrations

In the absence of extensive monitoring networks for UFPs, investigators have begun to augment the limited monitoring data with mathematical modeling approaches for predicting spatial and temporal concentrations of UFPs over broader areas. Two methods have been explored: regional transport models and land-use regression models. In theory, numerous regional advection and dispersion models could be used to study the fate of primary UFPs (as discussed, for example, in the review by Holmes and Morawska [2006]), but special care must be exercised to specify the correct emissions rate, coagulation rate, and nucleation rate in order to represent ambient UFP NCs. Regional transport models that have incorporated these additional parameters, like the Community Multiscale Air Quality Model used the by U.S. EPA for regulatory analyses, have not been able to accurately predict UFP NCs (Elleman and Covert 2009, 2010).

As of this writing, two groups have explored land-use regression methods to model UFP concentrations in urban areas. Land-use regression models predict pollutant concentrations using relationships with land-use features such traffic intensity, building density, industrial development, and the amount of green space. Hoek and colleagues (2011) developed a land-use regression model with which they were able to explain 67% of the variability in measured total particle NC in Amsterdam. Terms in the model included the product of traffic intensity and the inverse distance to the nearest road squared (as measured in field observations), address density, and location near the port. When the variables obtained from field observations were removed, substantially less variability was explained $(R^2 = 44\%)$. The median temporal correlation between concentrations at the central site and the outdoor locations

was fairly high (r = 0.72). At the urban background location, there was a very low temporal correlation between $PM_{2.5}$ and particle NC (r = 0.19) and between particle NC and soot (r = 0.38). Abernethy (2012) developed a land-use regression model for UFPs using one-hour particle NC measurements (using CPC) at 80 locations and 135 geographic predictors in Vancouver, Canada. The strongest model predicted NC on the basis of length of truck routes within 50 meters, density of fast food locations within 200 meters, and natural log of the distance to the nearest port, but accounted for only half the variability in measured particle NC ($R^2 = 0.48$). Hourly median particle NCs were highly variable across the city; two-week average number concentrations were well-correlated with NO₂, NO, and NO_x concentrations at the same sites (r = 0.64, 0.65, and 0.70 respectively). Broader application of these models has been limited by sufficient UFP measurements with which to develop models.

Summary of Evidence Characterizing Human Exposure to Ambient UFPs

Several factors acting on the emission, transformation, and dispersion of UFPs contribute to the substantial spatial variability that exists in NCs within urban areas. This high degree of spatial variability presents an opportunity to study the potential related health effects but also suggests that epidemiologic studies of long-term exposures would require detailed spatial characterization of UFP concentrations.

Despite this spatial variability, NCs within an urban area have been shown to be reasonably correlated over time. In other words, a central monitoring site, while not accurately characterizing the concentration of UFPs elsewhere in the city, can be a reasonable measure of within- and between-day changes in UFP concentrations throughout the urban area. However, variation in the degree of correlation between sites over time may vary by city and should be confirmed before reliance on a central site monitor can be uniformly advised.

Analysis of UFP exposures by microenvironments indicates that, while indoor sources of UFPs can contribute to high indoor concentrations, contributions from outdoor sources can be substantial. Infiltration of UFPs from outdoor air varies with particle size but is relatively efficient for particles in the 70–100 nm size range. In the absence of major indoor sources, ambient UFP concentrations are moderately correlated with indoor concentrations. Where reasonably high correlations between indoor and outdoor concentrations exist, central monitoring sites may be adequate to characterize changes in personal exposure to UFPs in studies that rely on temporal variability. Microenvironmental exposure analyses suggest that time spent in proximity to motor vehicles is a major contributor to personal exposure to ambient UFPs (NC) in urban areas. Depending upon commuting mode, route type, and duration, exposures during commuting may account for as much as 50% of an individual's daily UFP exposure.

Land-use regression and other approaches to modeling ambient UFP concentrations may eventually provide an important alternative to or complement of intensive measurement campaigns. However, further work is necessary to assess the accuracy of the models' predictions in more locations and under different conditions. Adequate measurement data will still be necessary to build models and to evaluate their performance.

CHAPTER 3. Do UFPs Affect Health? What Is the Evidence from Experimental Studies in Animals and Humans?

Concern about the possible role of UFPs in air pollution health effects was originally driven by a greater understanding of the unique physical and chemical properties of UFPs. This concern was supported by laboratory animal exposure studies suggesting that UFPs are more toxic than larger particles at an equivalent mass dose. More recently, with the development of systems for concentrating and delivering ambient particles in the UFP size range, studies to investigate the effects of human exposure to UFPs in clinical settings have become possible.

This chapter will first address the unique physical properties of UFPs, summarizing what is known about their deposition, clearance, and translocation. The physical characteristics of particles in the UFP size range profoundly affect their behavior after inhalation into the respiratory system. These characteristics are hypothesized to account in part for potential differences in toxicity in comparison with larger particles in the accumulation mode or fine particle (PM_{2.5}) size range.

We will then review the key findings of experimental studies in laboratory animals and humans. Our primary focus will be on studies considered most relevant to effects of ambient UFPs, defined as \leq 100 nm in diameter. We will discuss some studies of deposition, clearance, and translocation that have used model UFPs, and some toxicologic studies involving exposures to concentrated ambient particles in the quasi-UFP range (particles < 150–180 nm).

Our focus on inhalation studies reflects our emphasis on evaluating the effects of exposure to ambient UFPs via the normal, physiological route. For this reason, we have not taken into account studies that have exposed laboratory animals via different routes of exposure, in particular, by intratracheal instillation. Although the dose administered by this method can be controlled, particle distribution in the lung is less uniform and far from physiological (Oberdörster 2010). In addition, intratracheal instillation administers particles in a bolus, so the dose-rate is much higher than via inhalation. Furthermore, we have chosen not to review the growing body of in vitro UFP toxicology research, while nevertheless recognizing the importance of such studies in understanding specific mechanisms. We do cite select studies that help in understanding UFP disposition after inhalation.

Along with the experimental studies with humans in laboratory settings, we also discuss real-world panel studies of human exposures to ambient environments rich in UFPs (for example, in areas with heavy traffic), as long as the study design included a control exposure with reduced concentrations of UFPs.

DEPOSITION, CLEARANCE, AND TRANSLOCATION **OF UFPs**

To fully understand the deposition of particles entering the human body, it is useful to review the relevant anatomy of the human respiratory tract, illustrated in Figure 15. Upon inhalation, air moves through the upper respiratory tract: first through the nasal or oral cavities, then into the pharynx, or throat, and then into the larynx and upper trachea. These constitute the extra-thoracic airways. The trachea enters the thorax and splits into the two tubular bronchi, which lead to the left and right lungs. Inside the lungs, the bronchi divide repeatedly into progressively smaller tubes that end in the bronchioli, the smallest subdivision of the bronchi. Attached to the end of the bronchioli are the alveoli, tiny air sacks covered with capillaries where gas exchange takes place.

UFP deposition in the respiratory tract differs importantly from that of larger particles and can be affected by such factors as exercise, oral versus nasal breathing, disease status, and age (Daigle et al. 2003; Chalupa et al. 2004;



Particle density: 1 g cm⁻³ Respiratory flow rate: 300 cm3 s⁻¹

Figure 15. Total and regional deposition of inhaled particles in the adult human respiratory tract during mouth breathing at rest according to "Human Respiratory Tract Model" of the ICRP. Note particle diameter > 0.5 µm relates to the aerodynamic diameter and a particle diameter < 0.5 µm relates to the thermodynamic diameter. (Source: Kreyling et al. 2006a, Figure 2, reprinted with permission from Springer Science+Business Media.)

The ICRP and others have developed models to predict the respiratory *deposition fractions* of inhaled particles based on particle characteristics and lung anatomy and physiology (ICRP 1994; Kreyling et al. 2006a,b). As shown in Figure 15, UFPs are predicted to deposit with highest efficiency in the bronchioles and alveoli, whereas larger particles (1 to 10 µm) preferentially deposit in the extrathoracic region and bronchi. A large fraction of very small particles (in the 1 to 15 nm range) also deposits in the extrathoracic airways, including the nose. The diffusional deposition probability of inhaled UFPs for the alveolar region peaks at 20-30 nm. For UFPs < 20 nm, alveolar diffusional deposition decreases, in part because increasing numbers of these small particles have already deposited in the upper airways. Furthermore, due to their ability to move via diffusion, UFPs deposit more homogeneously onto the epithelia of the various regions than fine particles or coarse particles $(PM_{2.5-10})$, and they can diffuse into nonventilated air volumes within the alveolar region.

Particle Clearance and Retention in the Respiratory Tract

Inhaled particles of all size ranges are generally cleared from different regions of the airways via both physical and chemical clearance processes (reviewed in detail in Oberdörster et al. 2005). In the bronchioles and alveoli, the major clearance mechanism results from particle phagocytosis by alveolar macrophages. If particles are not cleared from the lung, they may be retained over prolonged periods, which results in their accumulation in airway tissue.

UFPs appear to be cleared less quickly and completely from the lung than larger particles. Möller and colleagues (2008) studied the deposition of radiolabeled 100 nm particles in humans, using a shallow bolus inhalation technique that targeted deposition in the distal airways and alveoli. There was negligible particle clearance from the peripheral regions of the lung 24 hours after exposure. Findings in airways of dogs are similar (Kreyling et al. 1999).

The reasons for the slower clearance and hence prolonged retention of UFPs remain unclear. Mucociliary clearance may be less effective for UFPs, either because the particles penetrate through the mucus deep into the periciliary phase, a continuous aqueous layer of relatively low viscosity surrounding the cilia (Schürch and Gehr 1990), or they deposit in areas with a lung-lining layer in which mucous is reduced or absent. Whatever the mechanism, the lack of UFP clearance may lead to accumulation, furthering interaction of particles with lung cells, and particle translocation beyond the epithelial barrier.

Particle Translocation

UFPs have been hypothesized to have unique effects because of their potential for translocation into the blood via the lung, with subsequent transport to other organs, including the heart and brain. The mechanisms for translocation of UFPs into tissues are not well understood. Evidence suggests that UFPs may either be transported by endocytotic and exocytotic mechanisms or they may diffuse across membranes into airway cells (Geiser and Kreyling 2010); similar evidence has not been reported for fine particles. UFPs are not only taken up by macrophages but are endocytosed by epithelial lining cells to a greater extent than are larger particles (Geiser et al. 2008; Takenaka et al. 2012). In contrast to fine particles that deposit on the surface of epithelial cells, UFPs enter these cells rapidly; in this case they are no longer accessible for phagocytosis by alveolar macrophages (Rothen-Rutishauser et al. 2007). UFPs, by virtue of their small size, may form complexes with proteins in the epithelial lining fluid of the lung that enhance their movement into cells (Cedervall et al. 2007; Kreyling et al. 2007; Lynch et al. 2007). This is likely a complex process, involving protein adsorption and desorption to UFPs, and the kinetics may differ with different body and cellular fluids, organs, and tissues in ways that are not yet understood.

We will briefly review the evidence from laboratory animal and human studies for translocation of UFPs (also see Geiser and Kreyling 2010).

Laboratory Animal Studies Several experimental studies in animals have provided evidence for the translocation across the air-blood barrier of model UFPs such as gold, silver, TiO₂, polystyrene, and carbon, in the 5–100 nm size range. UFPs were found in the pulmonary vasculature and blood (Figure 16) (Berry et al. 1977; Kapp et al. 2004; Geiser et al. 2005) and in extrapulmonary organs, including the liver, spleen, kidneys, heart, brain, and reproductive organs (Takenaka et al. 2001, 2006; Kreyling et al. 2002, 2009; Oberdörster et al. 2002; Semmler et al. 2004; Semmler-Behnke et al. 2007). Estimates of the total translocated fraction of UFP range from 1%-2% of 50 nm polystyrene particles (Chen et al. 2006) to as much as 10% of 20-nm diameter radiolabeled iridium UFPs when translocation to connective tissue and bone were included (Kreyling et al. 2009). Furthermore, 20-nm iridium UFPs were poorly cleared from secondary target organs. Six months after a single one-hour UFP inhalation exposure, the total UFP fraction in all secondary target organs was



Figure 16. Images of particles (arrows) in the lung parenchyma (using energy-filtering transmission electron microscopy). Image A shows an 81 nm particle in the cytoplasm of a capillary endothelial cell (EN). Image B shows a 41 nm particle within an erythrocyte (EC) in the capillary lumen. (Source: Geiser et al. 2005, reproduced with permission from Environmental Health Perspectives.)

still close to 0.1% of the initial UFPs deposited in the lungs, and all organs studied still contained UFPs (Semmler et al. 2004; Semmler-Behnke et al. 2007).

Inhaled UFPs may translocate to the brain (Elder and Oberdörster 2006). UFPs that deposit in the olfactory turbinates of the nose may enter the olfactory nerve and be transported to the olfactory bulb of the brain. While UFPs may access the brain via this pathway, their effects on the central nervous system have not been evaluated in great detail (see *Neurological Responses* in the Experimental Studies section).

Human Studies We know from histopathologic evidence in studies of long-term, heavy particle exposure in smokers, coal miners, and asbestos workers (showing particle or fiber accumulation in the liver and other organs of the reticuloendothelial system) (Auerbach et al. 1980; LeFevre et al. 1982) that particles can be found in organs beyond the lung. However, these studies offer little insight as to the importance of particle size or the relative importance of inhalation and ingestion pathways and have questionable relevance to the inhalation of ambient UFPs. Comprehensive biokinetic analysis of particle translocation is not feasible in humans for ethical and technical reasons. Thus to date, very little direct evidence of UFP accumulation and retention in organs and tissues is available from experimental human studies.

A paper by Nemmar and colleagues (2002) has been widely cited as direct evidence for UFP translocation from the airways into the circulation. However, subsequent work has failed to confirm their findings. In this study, Nemmar and colleagues exposed young volunteers to an aerosol of 5-10 nm carbon UFPs labeled with ⁹⁹technetium. They detected the tracer in the blood within minutes after the exposure and in the liver and stomach within an hour. They interpreted their findings to indicate that insoluble UFPs passed rapidly into the blood, and were circulated to organs throughout the body. These investigators subsequently developed a pharmacokinetic model of inhaled UFP distribution to blood and other organs based on these findings (Péry et al. 2009). However, other investigators (Mills et al. 2006; Wiebert et al. 2006a) have repeated these studies with similarly sized carbon UFPs and were unable to find evidence for particle translocation into the blood. Mills and colleagues (2006) found

that radioactive technetium leached off 4-20 nm carbon UFPs when they entered the airways, and that the radioactive moiety, rather than the particle itself, was detected rapidly in circulating blood. Möller et al. (2006) also found that the radiolabel is rapidly leached off unless it has been stabilized on the carbon UFPs. Wiebert and colleagues (2006a) did not find evidence of significant translocation of inhaled 35 nm radiolabeled carbon UFPs into the systemic blood circulation over a 24-hour period. These human studies suggest that under these specific experimental conditions, less than 1% of the inhaled dose of UFPs enters the blood and is available for translocation beyond the lung. It remains unknown what extrapulmonary burden of UFP is required to elicit health effects, and indeed whether translocation of UFPs beyond the lung is responsible for any of the health effects associated with PM exposure (Brown et al. 2002; Mills et al. 2006; Wiebert et al. 2006a,b; Möller et al. 2008).

Summary of Particle Deposition, Clearance, and Translocation

Compared with larger particles, UFPs deposit with higher efficiency, are cleared more slowly, and are retained longer. This raises the concern that chronic or repeated exposure to UFPs may lead to more accumulation within the lung of UFPs than of larger particles. Laboratory animal studies demonstrate that inhaled UFPs, but not fine or coarse particles, can translocate across the lung epithelium into the circulatory system and then be transported throughout the body where they have the potential to affect directly the cardiovascular system and other organs. UFPs depositing in the nose may also translocate via the olfactory nerve to the brain. These studies in animals also suggest that UFPs, to a greater degree than fine or coarse particles, may accumulate in organs and tissues under normal physiological conditions with the potential for long-term adverse effects. However, human studies to date have been limited and have failed to find substantial translocation of inhaled UFPs beyond the lung. It is unknown whether translocated UFPs cause or contribute to the adverse effects of PM exposure that have been observed in humans.

EXPERIMENTAL STUDIES OF ADVERSE EFFECTS OF EXPOSURE TO UFPS IN ANIMALS AND HUMANS

Experimental studies provide insight to the effects of exposure to any toxicant. The nature and level of exposure can be carefully controlled and the health endpoints chosen and evaluated against specific scientific hypotheses. For this issue of HEI Perspectives, we have focused primarily on experimental studies that involve exposures to laboratory animals and humans via inhalation, the most physiologically relevant route for exposure to ambient air pollution. We have focused our discussion on studies of exposures to UFPs that are relevant to ambient, and especially combustionrelated, UFPs in the < 100 nm size fraction. In particular:

- Carbon UFPs. The rationale for using these particles is that most combustion-generated particles have a carbon core, and so are representative of a major combustion-derived component of particles in ambient air, particularly those derived from diesel engines. Carbon UFPs are generated by a spark discharge of graphite electrodes; 95% of those produced consist of EC and have a median diameter of approximately 25 nm. Experiments with these particles have provided useful information about the effects of pure ultrafine carbon particles, but these particles lack the other components that are typically adsorbed to UFPs found in ambient air.
- Ambient UFPs. Some studies have compared the effects of exposures in contrasting environments with high versus low concentrations of ambient UFPs. A very few studies, conducted in New York State, have examined the effects of animal exposure to on-road emissions.
- ٠ Concentrated ambient UFPs. In order to study exposures to UFPs separately from other size fractions and at concentrations that are higher than those in ambient air, investigators have relied on UFP concentrators. These devices concentrate UFPs first by growing the particles in a supersaturated chamber, concentrating them using virtual impaction, and then drying them to their original size distribution. This technology becomes less efficient for particle sizes below 35-40 nm. They also usually includes a portion of particles > 100 nm, and consequently are considered quasi-UFPs. It should also be noted that an intrinsic limitation of UFP concentrator technology is the potential for chemical reactions with the condensed water, which may change the physicochemical properties of the original UFPs.

Finally, we acknowledge an entire class of experimental studies of inhalation exposures to DE in laboratory animals and in humans. Because DE from older engines has been an important source of UFP emissions, these studies are often taken to represent the effects of UFPs themselves. However, whole DE is a complex mixture of both gases and particulates, and few studies have attempted to account for the role of the various components in the health effects observed. In addition, few of those studies measured particle number. Like others who have reviewed this literature (U.S. EPA 2009; Hesterberg et al. 2011), we view the findings from this body of work to provide supportive but not direct evidence on the role of UFPs and have therefore summarized them in Sidebar 2.

Our evaluation of the studies in this chapter has focused on the set of health effects with which PM has been associated more generally in the scientific literature and for which UFPs have also been hypothesized to play a particular role given their physical and chemical characteristics:

- effects on the respiratory system, including increases in lung inflammation and allergic responses, and decreased lung function;
- effects on the cardiovascular system, including progression or exacerbations of cardiovascular disease (CVD); and
- effects on the neurological system, including increases in inflammatory responses and adverse effects on cellular function in the brain.

There are multiple pathways by which PM in general is hypothesized to exert adverse effects on various organ systems (Brook et al. 2010), but some pathways may be especially relevant for UFPs, for example, effects on the brain via translocation from the nose. Figure 17 provides a schematic of the multiple pathways via which inhaled UFPs are hypothesized to cause effects, directly or indirectly, in different organ systems. As we described in earlier sections, inhaled UFPs and their chemical constituents may act directly on the airways, the first point of contact for the particles. The indirect pathways involve mechanisms by which inhaled particles or their chemical constituents may trigger a series of responses in airway cells, in particular the synthesis of reactive oxygen species and induction of oxidative stress that may in turn result in inflammatory responses in the lungs, blood, or remote tissues. These inflammatory responses may themselves have adverse effects, including changes in the balance of the autonomic nervous system, which governs critical body functions like heart rate (HR) and respiration rate. Finally, particles may translocate from the airways to other organs and directly induce effects at sites such as the heart, liver, or brain. The animal and human studies discussed in this chapter evaluated the evidence that UFPs may be exerting influence via these various pathways using a variety of measures.

Laboratory Animal Studies

Experimental studies in animals offer certain advantages over studies in humans. Exposures can be conducted, and the toxicologic endpoints followed, over longer periods than in human studies. Endpoints can include sampling of tissues and organs. Animal studies can also include exposures during sensitive life stages such as fetal development and the extremes of age. Furthermore, special animal

Sidebar 2. Diesel Engine Exhaust — Components and Health Effects in Clinical and Animal Studies

Several controlled-exposure studies in human volunteers and rodents have examined the effects of inhaling whole diesel engine exhaust, in particular on the cardiopulmonary system. These studies are of some relevance to this issue of HEI Perspectives, because the particulate emissions from diesel engines include UFPs. Diesel particulate emissions vary in composition and in how they were formed. Nuclei mode particles are composed largely of volatile organic and sulfur compounds with smaller amounts of solid material (carbon and metallic compounds), and are generally less than 30 nm in diameter. They are the major contributor to particle number. The accumulation mode particles range in size between 30 and 500 nm and consist largely of soot (solid carbonaceous material and ash) and of adsorbed organic and sulfur compounds. These particles straddle the ultrafine and fine ranges and contribute to some extent to the UFP number (Kittelson et al. 2002). DE also contains many gases, including NO_x (primarily NO and NO₂) and carbon dioxide. The relative proportion and composition of the two particle modes and of the gaseous compounds depends on a number of factors, including testing conditions, engine type and operating condition, and fuel type.

Human clinical studies of DE have been conducted primarily in two research centers: at the University of Umea, Sweden (using either a 1990 or 1991 4-cylinder 4.5 liter Volvo diesel engine), and at the University of Washington, in Seattle (using a 2002 5.9 liter Cummins diesel engine). Details about these studies can be found in Hesterberg and colleagues (2010, 2011).

The studies investigated the effects of short-term exposure (1-2 hours) to DE on pulmonary function and immunologic and inflammatory endpoints in healthy individuals, as well as those with cardiopulmonary conditions, including asthma, metabolic syndrome, and post-myocardial infarction. Exposures were conducted at PM mass concentrations between 100 and 300 µg/m³, but particle number or UFP concentrations were generally not reported.

These studies used differing exposure protocols and outcome measures. Nonetheless, they provide evidence that DE can cause airway inflammation and changes in systemic vascular endothelial function and thrombus formation — physiologic endpoints with relevance to both acute respiratory and cardiovascular effects. One study (*Continued next page*)

models of diseases such as asthma or atherosclerosis can be used to test hypotheses about disease-related susceptibility. The major weaknesses of these studies are the failure of many animal models to replicate all aspects of disease states and the difficulty in extrapolating findings in animals to humans. We have limited our review to studies of inhalation exposure, to studies that provide exposures specifically to UFPs, with appropriate characterization of the UFPs and, where possible, to studies that compare the effects of UFPs to other particle size ranges.

Respiratory Responses Experimental inhalation of specific kinds of model UFPs can cause airway inflammation in rats, which can be more intense than with larger (fine) particles at equal mass concentrations (Elder et al. 2000a,c; Oberdörster et al. 2000). However, inhalation in mice of carbon UFPs at concentrations considerably higher than ambient (380 µg/m³, < 100 nm) did not cause an increase of inflammatory cells in bronchoalveolar lavage fluid (Andre et al. 2006; Maier et al. 2008). Similarly, Elder and colleagues (2004b) found no airway inflammation in response to 6-hour inhalation of carbon UFPs at 150 µg/m³. A higher concentration (1.7 mg/m³ UFP, median diameter 114 nm) did produce clear signs of lung inflammation (Gilmour et al. 2004). In rodent models of respiratory compromise, such as aging rats and rats with respiratory infection, shortterm inhalation of carbon UFPs enhanced pulmonary inflammation and oxidative stress (Elder et al. 2000a,c).

The lab-generated carbon UFPs used in these studies are likely to be less toxic than ambient UFPs at an equivalent concentration because ambient UFPs contain reactive organic and other chemical species. However, Elder and colleagues (2000a,b) exposed aged rats, with or without pretreatment with lipopolysaccharide or influenza virus, to freshly generated on-road aerosols for 6 to 18 hours. In general, on-road particles (which were predominantly UFPs) did not cause airway inflammation, or significantly enhance the background airway inflammation caused by the priming agents.

Several studies from investigators at University of California–Davis have explored the effects of exposure to UFPs in neonatal rats — a critical period of lung development (Pinkerton et al. 2004, 2008; Zhong et al. 2010). Specifically, the investigators evaluated the effects on neonatal lungs of exposures to combinations of laboratory-generated ultrafine iron and soot particles (20 nm in diameter, comprising both EC and OC), which were intended to model components of combustion-source–derived emissions.

Sidebar 2 (Continued)

found alterations of electroencephalogram signals in the frontal cortex of the brain during and up to one hour after exposure, suggesting possible effects on the central nervous system, although this has not yet been confirmed.

Both long- and short-term DE inhalation studies have also been conducted in laboratory animals, particularly rodents. Long-term exposure (up to 24 months) to high concentrations of DE (1 mg/m³ or higher) resulted in an increase in lung tumor incidence in rats, but generally not in mice. However, long-term exposure to high levels of particles other than DE (such as carbon black and titanium dioxide) can also increase lung tumor incidence in rats, but not in other species. For this reason, the increase in carcinogenicity in the DE exposure studies has been attributed to a rat-specific *particle overload* response in the lung, rather than to a response specific to DE. Shorter exposures to DE, particularly in animal models of cardiovascular conditions such as the ApoE knockout mouse, suggest effects on cardiovascular function, such as cardiac ischemia.

Several cautions should be noted about the extrapolation of the results of these controlled human and laboratory animal studies to real-world exposures to UFPs. First, in all these studies, the particle concentrations were above those encountered in typical ambient exposures, even in heavy-traffic situations; in several rodent studies the particle concentrations were orders of magnitude higher. Second, only a few of the studies reported particle numbers, so the contribution of particles in the ultrafine range to the effects detected is uncertain. Indeed, because DE contains multiple components that may be toxic, it is not possible to conclude from these studies that effects were caused by ultrafine or larger particles, by gaseous components, or by some combination of particles and gases. Finally, regulations introduced in the United States to take effect in 2007 and beyond have mandated reductions in diesel engine emissions; the technologies developed by engine manufacturers (as well as the use of low-sulfur fuel) have substantially reduced the number and mass of particles produced by these newer engines. The health effects of these emissions, currently being investigated in HEI's Advanced Collaborative Emissions Study (ACES) program (Coordinating Research Council 2009; ACES 2012; Mauderly and McDonald 2012), are likely to differ from those of older engine emissions.



Figure 17. Hypothesized pathways via which inhalation of UFPs may lead to effects on cardiovascular and respiratory systems and on the brain.

With exposures of $243 \pm 34 \ \mu g/m^3$ for six hours per day for three days, cell proliferation in the alveolar region of the lung decreased, and there was evidence of oxidative injury and increases in some markers of inflammation. Lee and colleagues (2010) found changes in the architecture of the airways of adult rats that had been exposed as neonates to combustion-generated 73 nm UFP with a high OC/EC ratio; changes in lung architecture were not found after exposures to fine particles (212 nm diameter) generated with a similar OC/EC ratio.

In summary, animal studies suggest that UFPs at high concentrations have the potential to induce airway inflammation, but the concentration of *ambient* UFPs necessary to induce an inflammatory response is not known and may exceed the relatively high concentrations found on a busy roadway. Responses in different species may vary and may also differ by age of the animal. For example, some evidence suggests that neonatal exposures to UFPs may alter lung development, with the potential for lifelong consequences. *Allergic Responses* The potential for UFP exposure to enhance respiratory allergic responses has been of particular interest because several early studies had shown that instillation or injection of diesel particles into animals could enhance characteristics of the allergic response (Muranaka et al. 1986; Takafuji et al. 1987; Fujimaki et al. 1997; Takano et al. 1997).

A series of studies has been conducted in California using concentrated quasi-ultrafine and fine particles collected using the versatile aerosol concentrator enrichment system (VACES). In considering the results, it is important to note that when the VACES system is used to concentrate fine particles, it includes all particles < $2.5 \mu m$ and so also includes particles in the quasi-UFP range (0.01 to < $0.18 \mu m$).

Kleinman and colleagues used VACES in several studies of the effects of inhaling concentrated UFPs collected near roadways in a mouse model of allergy — the sensitization and challenge of BALB/c mice with the allergen, ovalbumin (OVA). In the first such study (Kleinman et al. 2005), mice were exposed (4 hours/day, 5 days/week for 2 weeks) to either concentrated quasi-UFPs (< 150 nm) or concentrated fine particles (< 2.5 µm) collected at different distances (50 m or 150 m) from a roadway used by many diesel trucks. Particle mass in both quasi-UFP and fine particle groups averaged approximately 400 μ g/m³, and particle counts averaged 200,000/cm³. No differences in effects were found between exposure to concentrated fine particles (including UFPs) or to UFPs alone. However, mice exposed to either size fraction of concentrated particles at 50 meters from the roadway showed greater increases in markers of inflammation and of the allergic response (immunoglobulin [Ig]E, IgG1[the mouse equivalent of human IgG4, which is elevated in allergic responses], interleukin [IL]-5, and eosinophils) compared to mice exposed at 150 meters from the roadway.

Similar results were found in a follow-up study by the same investigators (Kleinman et al. 2007); IL-5 and IgG1 levels were increased in mice exposed nearest to the road (50 m) but not at the greater distance (150 m) from the road. These increases were associated with EC and OC components of both fine particles and UFPs, suggesting the importance not just of particle size but of particle composition. Li and colleagues (2010) also found that exposure to concentrated ambient quasi-UFPs, (< 180 nm) collected close to a freeway in Los Angeles, enhanced the secondary or memory-type response in this animal model: exposure to UFPs increased features of the allergic response (influx of eosinophils into the airways, increased levels OVAspecific IgE and IgG1, and enhanced expression of the cytokine genes IL-5 and IL-13 in the lung). Exposure to UFPs in this study also enhanced inflammatory-type responses (enhanced expression of IL-17a and influx of neutrophils in the lung).

In addition, in a similar OVA mouse model of allergy, several studies have shown that 24-hour exposures to carbon UFPs (< 100 nm) potentiated the effects of lung allergic inflammation (Alessandrini et al. 2006, 2008; Maier et al. 2008). Alessandrini and colleagues (2006) showed that 24-hour exposure to 526 µg/m³ 35 nm carbon UFPs up to 96 hours before OVA challenge enhanced bronchoalveolar lavage inflammatory cell infiltrate and IL-4, IL-5, and IL-13 levels, as well as mucus production in the airways. Exposure to the same concentration of carbon UFPs 24 or 72 hours after OVA challenge had much milder effects on airway inflammation, suggesting that sensitized animals are more sensitive to the effects of exposure to UFPs if exposed before allergen challenge. Evaluating particle deposition in the BALB/c OVA sensitization and challenge mouse model, Alessandrini and colleagues (2008) found that compared to nonsensitized mice, OVA-sensitized mice exposed for one hour to ultrafine iridium particles radiolabeled with¹⁹²Ir (UF-Ir) before OVA challenge showed a 21% relative increase in the total UF-Ir deposited fraction. When inhalation of UF-Ir was performed after allergen challenge, no differences in total deposited fraction or extrathoracic deposition or regional particle deposition were detected between sensitized and nonsensitized mice.

Alessandrini and colleagues (2009) confirmed that 24-hour exposure to 504 μ g/m³ 35 nm carbon UFPs before OVA challenge enhanced the markers of inflammation they had detected previously, but also found that levels of 8-isoprostane, a marker of lipid peroxidation and oxidative stress, and NF- κ B, a transcription factor that activates genes involved in inflammatory and other responses, were also enhanced. These studies provide support for the hypothesis that UFP exposure may enhance components of the allergic response, perhaps by facilitating the entry or processing of allergen that has deposited in the airway. However, the exposure concentrations in these studies were quite high, and it is uncertain whether these effects occur at concentrations more relevant to ambient levels.

Cardiovascular Responses The relatively few animal studies that have evaluated cardiovascular responses provide some evidence of UFP effects on the cardiovascular system but differ in the types of UFP exposures used and the outcomes examined.

Two studies have found that rodents exposed to carbon UFPs at concentrations relevant to ambient levels (100–200 µg/m³, 1–5 × 10⁶ particles/cm³) for as little as 24 hours showed changes in cardiovascular endpoints. Exposure to carbon UFPs (median diameter 72–74 nm; aggregate: 80% mass < 100 nm) for 24 hours showed thrombogenic effects in the microcirculation of healthy mice without any significant sign of inflammation in the respiratory tract (Khandoga et al. 2010). Furthermore, a mild but consistent increase in HR and a significant decrease in HR variability were found during inhalation of carbon UFPs (median diameter 38 nm) (Harder et al. 2005).

Araujo and colleagues (2008) have conducted one of the few studies designed to compare directly the effects of exposure to UFPs (albeit quasi-UFP < 180 nm) and $PM_{2.5}$ (containing concentrated UFPs). It is also one of the few animal studies of cardiovascular outcomes to compare the composition of the two size fractions. Using particles concentrated by the VACES from ambient air close to a Los Angeles freeway, Araujo and colleagues (2008) compared the effects of exposures to quasi-UFPs and fine PM at

approximately the same NC ($\sim 5 \times 10^5$ particles/cm³). Apo E knockout mice — a strain that develops atherosclerosis more rapidly than normal mice (particularly if fed on a high-fat chow) — developed 25% and 55% larger atherosclerotic lesions when exposed to concentrated UFPs (5 hr/day, 3 days per week for 5 weeks) compared with PM_{2.5} and with filtered air, respectively (see Figure 18).

Exploring the possible mechanisms by which particles might affect the development of atherosclerosis, the investigators found that, compared with exposure to $PM_{2.5}$, exposure to the quasi-UFPs resulted in a decrease in the anti-inflammatory capacity of plasma high-density lipoprotein and in increased measures of systemic oxidative stress.

It remains unclear whether differences in particle composition could at least partially explain these findings. Figure 19 summarizes the percentage contribution by mass of metals, nitrates, sulfates, EC, and OC in each size fraction. The quasi-UFP fraction was enriched in OC, and to a lesser degree in EC, compared with the $PM_{2.5}$ fraction. Further exploration of the implications of composition is needed.

Elder and colleagues (2004a; 2007) have conducted the only studies of rats exposed by inhalation to ambient



Figure 18. Mouse model of atherosclerosis: Near-roadway concentrated fine particulate matter ($PM_{2.5}$) and quasi-ultrafine particles (< $PM_{0.18}$) compared with filtered air. (Source: Araujo et al. 2008, reprinted with permission from Wolters Kluwer Health.)

particles, predominantly UFPs, while being driven along a major highway (I-90 in New York State). They studied both pulmonary and cardiovascular endpoints. Elder and colleagues (2004a) found that a 6-hour on-road exposure $(1-3 \times 10^5 \text{ particles/cm}^3)$ of older rats (21 months), with and without prechallenges using either endotoxin (lipopolysaccharide) or influenza virus, was associated with enhanced plasma endothelin-2, which causes constriction of arteries and increases in blood pressure (BP). Elder and colleagues (2007) later evaluated a similar 6-hour on-road exposure (count median diameter 15-20 nm) on the HR and heart-rate variability (HRV) of spontaneously hypertensive rats. Rats exposed to the highway aerosol had a lower HR compared to rats exposed to clean air, an effect that persisted after exposure. In addition, exposure to the highway aerosol affected several HRV parameters that suggested an effect on the autonomic nervous system, with a shift from parasympathetic to sympathetic (fight or flight) influences.

This collection of animal studies provides evidence of cardiovascular effects associated with UFPs of different size fractions; the study by Araujo and colleagues (2008) suggests that particles in the quasi-UFP fraction alone have a greater effect than an equivalent number of fine particles on the progression of atherosclerosis. However, the animal data are insufficient to provide clear evidence that UFPs have cardiovascular effects that differ from those of fine particles.



Figure 19. Comparison of the chemical composition of concentrated quasi-ultrafine ($PM_{0.18}$) and fine particulate matter ($PM_{2.5}$) in the previous study of mouse atherosclerosis in Figure 18. (Source: Araujo et al. 2008, adapted with permission from Wolters Kluwer Health.)

Neurological Responses A small number of studies have looked at neurological responses in rodents after exposures to ambient UFPs or more specifically, to concentrated quasi-UFPs (< 180 nm). As discussed earlier, a limitation of these studies from the standpoint of this issue of HEI Perspectives is that they involve exposures to particles that are larger than the < 100 nm definition for UFPs. However, a strength is that at least some of these studies also conducted direct comparisons with particles in the $PM_{2.5}$ size range. Collectively, these studies provide some indication that, compared with filtered air, exposure to quasi-UFPs in the vicinity of major freeways induces allergenic responses, inflammatory responses, or both in the brain. However, in studies with parallel exposures to fine particles, similar inflammatory responses were observed.

Campbell and colleagues (2005), studied the brains of the OVA-sensitized BALB/c mice that had been exposed to filtered air, concentrated quasi-UFPs, or the fine particles collected at varying distances from the roadway in the study by Kleinman and colleagues (2005) (discussed earlier in the *Allergic Responses* section). The brains of these mice showed increased levels of mediators associated with the induction of inflammatory responses — interleukin-1 alpha (IL-1 α) and tumor necrosis factor alpha (TNF α), and of the transcription factor NF- κ B. Increases in levels of IL-1 α and TNF α were also detected by Campbell and colleagues (2009) in the brains of ApoE knockout mice after exposure to either quasi-UFPs or fine PM.

Kleinman and colleagues (2008) found that Apo E knockout mice exposed to concentrated UFPs (4- or 15-fold) from ambient air close to a freeway in central Los Angeles for 5 hours/day, 3 days/week for 5 weeks showed changes in brain cell function compared to cells from filtered air controls. Changes included a dose-related increase in nuclear translocation of NF-KB and in another transcription factor, AP-1, that are associated with the induction of immune and inflammatory responses. Exposure to quasi-UFPs also activated NF-κB and AP-1 in the brains of the ApoE knockout mice reported by Campbell and colleagues (2009). Kleinman and colleagues (2008) also found that the lower concentration of concentrated UFPs also increased the activation of a kinase, JNK, which participates in one of the intracellular cascades that leads to the activation of these transcription factors. In addition, UFP-exposed mice showed increased expression of glial fibrillary acidic protein, a molecule expressed on the surface of glial cells. These findings suggest that exposure to concentrated ambient UFPs near a roadway has the potential to induce inflammation in the brain.

Summary of Animal Studies These studies suggest that traffic-related UFPs may enhance allergic responses in allergen-sensitized animals. Furthermore, quasi-UFPs may enhance the progression of atherosclerosis in ApoE knockout mice, and may influence autonomic control of the heart in aged rats. Markers of inflammation in the brain increased after exposure to concentrated traffic-related quasi-UFPs, but similar increases were also observed with exposure to concentrated $PM_{2.5}$. These are intriguing findings, suggesting that there may be extrapulmonary effects of traffic-related UFPs. However, the findings must be considered preliminary; they require confirmation in other models and laboratories.

Experimental Human Exposures to UFPs

Human exposure studies of various designs have constituted an important set of experiments for understanding the health effects of air pollution and have played an important role in establishing rational ambient air pollution standards. Human clinical studies, studies of human exposure to controlled atmospheres usually performed within a specially designed exposure facility or chamber, have several strengths and weaknesses that have been discussed at length in previous reviews (Frampton 2006; Langrish et al. 2011). A major strength, of course, is that humans are the species of most interest. They can be exposed to particles via a physiologically relevant route such as oral or oro-nasal breathing using exposure atmospheres that can be carefully controlled and characterized. Different exposure scenarios can be designed to compare effects, in the same person, of exposure to particles of different size ranges when the individual is engaged in different levels of activity. Exercise for example, increases breathing and can therefore affect particle intake, and deposition (see section, Deposition, Clearance, and Translocation of UFPs). Under carefully controlled conditions, the effects of exposures can be evaluated in potentially susceptible subpopulations, such as those with cardiorespiratory diseases, asthma, and diabetes.

However, such experiments are generally limited to short-term exposures (a few hours maximum) and so do not provide insight into potential chronic effects. Ethical and safety considerations prevent the study of those most susceptible to pollutant health effects, such as people with severe airway constriction or with CVD, and limit the use of invasive outcome measures.

Clinical studies of UFPs have involved unique technical challenges. Prior to the development of UFP concentrators 10–15 years ago, there was no way to study ambient exposures to UFPs that did not include other particles and gases. As indicated earlier, because UFPs have so little mass even the current generation of UFP concentrators does not efficiently concentrate UFPs smaller than about 35–40 nm. UFPs can be generated in the laboratory, but cannot be collected and resuspended for later exposure because the UFPs agglomerate into larger particles. For these reasons, there have been relatively few human controlled-inhalation studies of UFP exposure, and the exposure atmospheres of those studies are not entirely representative of ambient UFPs. Nonetheless, with these caveats in mind, the limited number of such studies has provided valuable information.

To move beyond some of the limitations of clinical studies, we have broadened our discussion of experimental studies of controlled human exposure to include panel studies of exposures to real-world environments, such as walking along a busy street. We have included the latter studies provided they included both an assessment of exposure to UFPs, such as particle number counts, and a cleaner air exposure as a control. These study designs involve other challenges but can also offer additional insights.

Respiratory Responses Respiratory outcome measures include pulmonary function (in particular, forced expiratory volume in 1 second [FEV₁], forced vital capacity [FVC], and peak expiratory flow [PEF]). Pulmonary function testing is used to diagnose and monitor respiratory diseases such as asthma and chronic obstructive pulmonary diseases (COPD), and reductions in FEV₁ correlate with impairments in physical functional status. Even transient reductions in lung function, if accompanied by symptoms, are considered to be adverse health effects (American Thoracic Society 2000). Other respiratory measures include an influx of inflammatory cells into the airways and markers of airway injury or inflammation in exhaled air.

Respiratory responses of UFPs have been studied in clinical chamber studies using both laboratory-generated UFPs and UFP concentrators, as well as in real-world exposure settings enriched in UFPs. While real-world exposures have shown some respiratory effects, most chamber studies have not.

Chamber Studies — *Carbon UFPs* A group in Rochester, New York, led by Drs. Frampton and Utell, has published a series of studies (Frampton et al. 2004; Pietropaoli et al. 2004a,b; Stewart et al. 2010) that focus on the effects of controlled inhalation via mouthpiece of laboratorygenerated carbon UFPs. These experiments provided useful information about the effects of pure carbon UFPs, but in the absence of adsorbed components that would be attached to UFPs found in ambient air. In aggregate, these studies show little evidence for acute effects of carbon UFPs on lung function. One study of exposure to 50 μ g/m³ carbon UFPs for two hours with intermittent exercise found small, marginally significant reductions in the maximal midexpiratory flow rate (about -5% relative to filtered air), suggesting mild obstructive or small airways effects (Pietropaoli et al. 2004a). However, there were no significant effects on FEV₁, and none of the other carbon UFP inhalation studies showed lung function effects (Stewart et al. 2010). Similarly, there was no evidence for increased airway inflammation, assessed by a lack of changes in markers in induced sputum or in the level of exhaled NO.

In a comparative study of ultrafine and fine particles, the investigators exposed resting healthy subjects to clean air, to 500 µg/m³ of ultrafine zinc oxide particles, and to 500 µg/m³ of ultrafine zinc oxide particles that were allowed to agglomerate in an aging chamber to the fine particle size range (Beckett et al. 2005). Zinc oxide particles can be generated by welding processes, and in occupational settings their inhalation can lead to a systemic inflammatory response known as metal fume fever. However, no effects were detected in any of the physiological, airway or systemic inflammatory, or cardiac endpoints examined for exposure to particles of either size range.

Chamber Studies — *Concentrated UFPs* Only a few studies have utilized concentrated ambient UFPs to examine health effects in humans. As with chamber studies of exposure to carbon UFPs, they have provided weak evidence of UFP effects on respiratory outcomes.

Gong and colleagues (2008) exposed healthy and asthmatic volunteers to filtered air or to 7- to 8-fold concentrated quasi-UFPs (mean particle count 145,000/cm³, mean mass concentration 100 μ g/m³, and diameter < 180 nm) for two hours with intermittent exercise in a chamber. The UFPs were collected in a Los Angeles suburb that has heavily-traveled roadways. Exposure to UFPs was associated with a 0.5% mean reduction in arterial oxygen saturation and a 2% mean reduction in FEV₁ the morning after exposure, estimated across all subjects. However, there was no evidence for increased airway inflammation by analysis of induced sputum or exhaled air. Responses in healthy and asthmatic subjects were similar.

Samet and colleagues (2007, 2009) compared the effects of concentrated ambient quasi-UFPs ($PM_{< 0.16 \ \mu m}$), $PM_{2.5}$, and coarse particles ($PM_{2.5-10}$) delivered to young, intermittently exercising human volunteers at the U.S. EPA lab in Chapel Hill, North Carolina. These studies used the Harvard particle concentrator systems which, in contrast to the VACES system, delivers fine and coarse particles that do not include UFPs. Concentrated quasi-UFPs ($1.52 \times 10^5 \pm 1.65 \times 10^5$ particles/cm³) had no effect on pulmonary function or on the influx of inflammatory cells

in bronchoalveolar lavage fluid 18 hours after the exposure, whereas fine and coarse particles caused a modest degree of airway inflammation. Both the Gong and Samet studies also examined multiple cardiovascular endpoints, which are summarized in a later section on cardiovascular responses.

Real-World Ambient Air Studies A series of panel studies compared health responses in individuals exposed to high numbers of UFPs (e.g., high traffic intensity) with those in individuals exposed to cleaner air (e.g., low traffic intensity). An advantage of the study design is the use of realistic exposures to ambient UFPs. However, these exposures necessarily involve mixtures of pollutants, making it difficult to attribute effects to any single pollutant type. Such exposures cannot be blinded, so expectations on the part of subjects or investigators could confound the results. Furthermore, the exposure settings likely differ in ways other than pollutant concentrations. For example, traffic noise, odor, and visual stimuli may differ between the experimental and control exposure settings, with unpredictable results on outcome measures.

Two publications examined the results of a study of lung function and inflammatory responses in 60 subjects with mild or moderate asthma who walked for two hours on separate occasions on Oxford Street and on Hyde Park in London (McCreanor et al. 2007; Zhang et al. 2009). Oxford Street is a busy street on which only diesel vehicles are allowed. Exposures were characterized in real time, including total particle number counts using a CPC. The Oxford Street exposures had higher particle counts compared with the Hyde Park exposures, and they were associated with greater declines in FEV1 and FVC, increased markers of airway inflammation in induced sputum, and a decline in the pH of exhaled breath condensate. These findings indicate that the Oxford Street exposures worsened markers of asthma in these subjects, although self-reported symptoms of asthma did not change (Zhang et al. 2009). Respiratory changes were statistically most strongly associated with UFP exposures, although there were also significant associations with exposure to NO₂. However, in multiple-pollutant models, particle number counts and EC concentrations were most consistently associated with effects.

Rundell and colleagues (2007, 2008) have studied lung function and markers of inflammation in healthy young volunteers exercising in environments with contrasting total NCs ($NC_{0.02-1.0}$ as measured by CPC). In the 2008 study, subjects performed vigorous 30 minute running trials, either on an inner campus loop free of vehicle traffic with low particle counts (mean 7382 particles/cm³), or on a soccer field and trail within 50 meters of a major highway with high

particle counts (mean 252,290 particles/cm³). Concentrations of CO, NO₂, and O₃ were comparable in the two environments. The near-traffic exposures were associated with statistically significant airway effects, including reductions in lung function, alveolar NO concentrations, and nitrate, and with increased levels of malondialdehyde, a marker of oxidative stress in exhaled breath condensate. These findings were interpreted as suggesting pollutantinduced airway effects, but the changes cannot be attributed specifically to UFPs.

Similar studies have been conducted by Strak and colleagues (2010) with bicyclists in Europe and have detected little evidence of an UFP effect on lung function or measures of inflammation. These investigators evaluated a marker of airway inflammation (exhaled NO), and lung function (FEV₁, FVC, and PEF) in 12 healthy adults who cycled on a low- and a high-traffic intensity route in Utrecht, the Netherlands. As expected, particle number counts were higher on the high-traffic intensity route (41,097 particles/cm³) than on the low-traffic intensity route (27,028 particles/cm³); however, the PM₁₀ concentrations were similar on both routes. There were no statistically significant relationships between the exposures and the outcome measures. In a study in Antwerp with a larger number of healthy volunteers (38), Jacobs and colleagues (2010) compared the effects of bicycling in traffic (28,867 particles/cm³) and bicycling in a laboratory with filtered air (496 particles/cm³). There was no effect of exposure on exhaled NO, which is considered a measure of airway inflammation. Bicycling in traffic was associated with an increase in blood neutrophils, but no change was observed in the other blood markers tested.

A recent study of cyclists in Ottawa, Canada did not find strong associations of total NC or BC, NO₂, or O₃ with respiratory outcomes; associations with exhaled NO and pulmonary function measures were inconsistent across endpoints, lags, and pollutants (Weichenthal et al. 2011). In another recent series of studies of commuters' exposures to air pollution in traffic, Zuurbier and colleagues found various associations of total NC with effects on lung function (FEV1, PEF), exhaled NO, and airway resistance in 34 healthy nonsmoking adults directly after and 6-hours after (depending on the endpoint measured) 2-hour commutes by bus, car, or bicycle; select associations with PM₁₀ and soot were also observed (Zuurbier et al. 2010, 2011b). Estimated effects were not consistent for other lung function parameters, respiratory symptoms, and blood markers of inflammation and coagulation (Zuurbier et al. 2011a,b). Potential confounding by stress, noise, or both was not assessed in these studies.

Klepczynska Nyström and colleagues (2010) examined the respiratory effects of exposure for two hours in a subway tunnel in Stockholm, Sweden, using an office environment as a clean-air control. Blood sampling and bronchoscopy were performed 14 hours after exposure. Pollutants measured in the tunnel and in the control environment included not only UFPs (mean NC_{< 100 nm}), but also PM_{2.5}, PM₁₀, NO, and NO₂. The NC_{< 100 nm} in tunnels was 110,000 particles/cm³ versus 8,283 particles/cm³ in the control environment. The investigators found no effects on lung function or airway inflammation associated with exposures in the subway. However, they did observe statistically significant increases in levels of blood fibrinogen and regulatory T-lymphocytes after the subway exposure.

Summary of Human Respiratory Responses Experimental studies with human subjects include a spectrum of exposures, from laboratory-generated model UFPs and concentrated ambient particles delivered in specially designed chambers, to ambient exposures in real-world settings. Relatively few studies have compared the effects of particles of different size fractions or accounted for the presence of other copollutants.

The small number of experimental studies conducted to date show a range of findings on respiratory outcomes, from reductions in lung function and increases in airway inflammation to no effects. In the one study that compared respiratory responses to UFPs, fine particles, and coarse particles, pulmonary inflammation was observed with fine and coarse particles but not with UFPs; no changes were observed in lung function for any exposure. Similarly, the laboratory-generated carbon UFP studies found no evidence for airway inflammatory effects and no convincing effects on lung function. In contrast, some of the realworld studies found changes in lung function and airway inflammatory markers, particularly in subjects with asthma, while others did not. While these diverse findings may reflect some of the inherent limitations of these study designs (e.g., small sample size, short durations of exposure, subjects who are not blinded to the type of exposure they receive), they raise additional questions about the potential importance of the overall mixture in which UFP exposures occur.

Cardiovascular Responses The role of exposures to PM in the development or exacerbation of CVD and in cardiovascular mortality in humans has been of considerable research interest (see, for example, the review by Brook et al. 2010). The following section provides a summary of the contributions from experimental, controlled human exposure studies to the base of evidence on UFPs.

Chamber Studies - Carbon UFPs The studies of laboratory-generated carbon UFP inhalation have suggested effects on both pulmonary and systemic vascular function in both healthy and asthmatic people. Inhalation of 10 or 25 µg/m³ of carbon UFPs ($\sim 2 \times 10^6$ and $\sim 7 \times 10^6$ particles/cm³, respectively) during intermittent exercise showed concentration-related attenuation of the exercise-induced increase in the peripheral blood leukocyte surface expression of adhesion molecules (Frampton et al. 2004, 2006). This finding was considered consistent with transient pulmonary vascular effects of carbon UFP exposure. It was also supported by studies of 50 μ g/m³ carbon UFPs showing reductions in the diffusing capacity for CO approximately 24 hours after a 2-hour exposure (Pietropaoli et al. 2004a). Taken together with the blood leukocyte findings, these results suggest that carbon UFP exposure transiently reduces pulmonary capillary blood volume, and thus affects pulmonary circulatory function. Although the significance of this finding is unknown, such an effect, if repeated and persistent, could contribute to remodeling of the pulmonary circulation and the development of pulmonary hypertension.

With regard to effects on the systemic circulation, a study in healthy exercising people inhaling carbon UFPs at 50 μ g/m³ (Shah et al. 2008) found that carbon UFPs completely inhibited the expected exercise-associated increase in peak hyperemic forearm blood flow (a measure of systemic vascular responsiveness with relevance for cardiac coronary artery disease) 3.5 hours after exposure. One potential mechanism for reduction in vascular responsiveness is a reduction in the circulatory bioavailability of endogenous NO, which causes relaxation of vascular smooth muscle. The investigators observed reduced plasma nitrate concentrations (a product of NO oxidation) in comparison with a control filtered air exposure. These findings support the hypothesis that inhalation of carbon UFPs impairs systemic vascular function and reduces NO bioavailability. A study of people with type 2 diabetes, exposed at rest to 50 μg/m³ carbon UFPs, showed changes in markers associated with enhanced blood coagulation an increase in plasma von Willibrand factor and an increase in markers of platelet activation (Stewart et al. 2010).

A recent study from the same research group focused on cardiac changes measured by electrocardiogram (ECG) in young healthy adults after exposure to 10 and 25 μ g/m³ carbon UFPs (Zareba et al. 2009). Changes were generally small and not significant, but the authors felt that there was a trend in HRV that indicated an increase in parasympathetic tone, the arm of the autonomic nervous system involved in slowing down the HR. The health implications of transient changes in HRV in young healthy people are not clear.

Chamber Studies — *Concentrated UFPs* The studies of Samet and colleagues (2007, 2009) comparing the effects of exposure to concentrated UFPs, fine particles, and coarse particles on respiratory endpoints also examined several indicators of cardiovascular function. As illustrated in Figure 20, exposure to concentrated UFPs increased D-dimer in blood, indicating activation of coagulation, and also transiently increased blood lipids. Continuous ECG monitoring revealed increases in markers of HRV and variance in duration of the QT interval (Samet et al. 2009).

HRV was also measured in the study by Gong and colleagues (2008) in which healthy and asthmatic volunteers were exposed to concentrated UFPs in a Los Angeles suburb (for exposure details, refer to the discussion in the *Respiratory Responses* section). Exposure to UFPs was associated with a transient slight decrease in lowfrequency power, without changes in other measures of HRV. Responses in people with asthma were similar to responses in people without asthma.

A recent study (Mills et al. 2011b) was designed to determine whether the effects of DE exposure on systemic vascular endothelial function were caused by UFPs or by the gaseous component of DE; 16 healthy volunteers inhaled 4 different atmospheres: diluted DE (particle concentration of 300 μ g/m³), laboratory generated carbon UFPs (< 100 nm, 4×10^{6} particles/cm³), filtered DE, or filtered air. After each exposure, forearm blood flow was measured in response to infusion of vasoconstrictors and vasodilators; Figure 21 provides illustrative results for the vasodilator, acetylcholine. The impairment of vascular responsiveness with exposure to whole DE that had been observed in an earlier study (Mills et al. 2005) was confirmed (panel A). However, neither filtered exhaust (panel B) nor pure carbon UFPs (results not shown, but were the same as for filtered exhaust) affected endothelial function measured by forearm blood flow. These experiments indicate that the particulate component of DE was responsible for the vascular effects, and that carbon UFPs alone do not reproduce the effect. Thus, it appears that particle size is not the only factor determining the vascular effects of DE. The findings do not exclude the possibility that some aspect of the gas-particle mixture is responsible.

Real-World Ambient Air Studies Brauner and colleagues (2008) performed an intervention study in 21 nonsmoking couples, ranging in age from 60 to 75 years, in their homes in Copenhagen, Denmark. Investigators installed high-efficiency particle filters in their homes, and the study consisted of consecutive 48-hour periods living with either filtered or unfiltered indoor air. The exposures were double-blinded and randomized, and monitoring included



Figure 20. Effects of concentrated quasi-ultrafine particles (< PM_{0.16 µm}) on D-dimer in blood and other markers of coagulation. (Source: Samet et al. 2009, reprinted with permission of the American Thoracic Society.)

particle counts and fine and coarse mass concentrations. Filtering reduced mean particle counts from 10,016 to 3,206 particles/cm³ and reduced mean $PM_{2.5}$ mass from 12.6 to 4.7 µg/m³. Microvascular function, measured by digital peripheral artery tone after arm ischemia, improved by 8.1% during air filtration. However, this effect was more strongly related to the $PM_{2.5}$ mass concentration than to particle number, suggesting that reductions in UFPs were not the driving influence in improving vascular function.

As part of the study discussed earlier, Rundell and colleagues (2007) also assessed systemic vascular effects of exercise in outdoor environments with high versus low ambient particle counts. Higher particle count exposures were associated with markedly reduced systemic vascular function (measured by reduced flow-mediated dilatation of the forearm) and reduced reperfusion of small vessels of the forearm (measured by near-infrared spectrometry).

A recent study of cyclists in Ottawa, Canada found suggestive associations of total NC with reduced HRV parameters within four hours of the start of cycling, although some results were sensitive to outliers; select associations with other pollutants (e.g., BC, NO_2 , and O_3) were also observed with HRV endpoints (Weichenthal et al. 2011).

Laumbach and colleagues (2010) piloted a protocol for assessing the effects of car commuting on HRV measures. The authors recruited 21 subjects with type 2 diabetes to participate in 90- to 110-minute car rides on a busy



Figure 21. Endothelial function and exposure to diesel exhaust particulates after acetylcholine infusion. Forearm blood flow was measured in healthy subjects 6–8 hours after exposure to DE, filtered DE, or filtered air — either during acetylcholine infusion or without any infusion. Significant dose-dependent increases in blood flow were observed with infusion (P < 0.0001) versus without infusion. This effect was significantly attenuated with exposure to DE (panel A; P = 0.008), but not with filtered DE (panel B; P < 0.05). Results using carbon UFP were similar to those for filtered DE (not shown). (Source: Mills et al. 2011b, by permission of Oxford University Press.)

highway in New Jersey. Changes in HRV parameters (e.g., reduced high-frequency HRV) relative to pre-ride levels were observed post-ride and on the next day, and while not statistically significant, the authors linked these with invehicle pollutant concentrations (total NC, $PM_{2.5}$, CO, NO_2). Potential confounding by perceived stress and anxiety was considered in this study using a stress questionnaire that was administered at four time points during each sampling session. Observed results were independent of stress or anxiety in sensitivity analyses that excluded subjects with high stress or anxiety scores. Noise levels, however, were not measured in this study.

Summary of Human Cardiovascular Responses As noted for respiratory responses, there are a small number of studies examining cardiovascular endpoints, with differing approaches and a range of findings. Human exposure studies to carbon UFPs in Rochester, New York, suggested small, transient effects on both pulmonary and systemic vascular function. However, the study by Mills and colleagues (2011b) examining the components of DE found that inhalation of carbon UFPs did not reproduce the vascular effects of whole DE. Indoor air filtration in the homes of older subjects improved microvascular function, but the effect was more strongly related to PM mass than to particle number, implicating the larger particles in the indoor air mix.

With regard to cardiac ECG monitoring, both the Rochester and Chapel Hill studies, using carbon UFPs and concentrated UFPs respectively, suggested effects on cardiac repolarization and increases in high-frequency and low-frequency power, without substantial effects on time-domain variables of HRV. However, the study by Gong and colleagues (2003, 2008) in Southern California showed reductions, rather than increases, in low frequency power. Thus, the effects of UFP inhalation on cardiac autonomic function remain unclear.

There is evidence that carbon UFP exposure activates platelets in people with diabetes, and that concentrated ambient UFPs activate coagulation, an effect not seen with concentrated fine and coarse particles. However these studies require confirmation.

Thus, there is evidence in some studies for UFP effects on vascular function, HRV, cardiac repolarization, and coagulation, all findings that support adverse cardiovascular influences of exposures to UFPs, especially for people with underlying heart or vascular disease. However, there are inconsistencies among studies. There remains insufficient evidence from human studies for definitive conclusions about cardiovascular effects of inhalation exposure to UFPs.

Other Responses One study (Vinzents et al. 2005) looked at measures of oxidative and mutagenic activity (level of DNA purine oxidation and strand breaks) in 15 healthy nonsmoking subjects who bicycled in traffic on five occasions and in the laboratory on one occasion, with personal monitoring of total particle number counts. Cumulative outdoor and indoor exposures to UFPs were each independent predictors of the level of DNA purine oxidation, but not of strand breaks. Other outdoor pollutants, including PM_{10} , NO_x , CO, and urban background UFP concentrations were not significant predictors of oxidative or mutagenic activity.

Summary and Conclusions for Experimental Studies

UFPs have unique physical properties that determine their deposition and disposition in the respiratory tract. These characteristics indicate that, with repeated or prolonged exposures, UFPs have a greater potential than fine particles for retention in the lung. Studies in animals have suggested that UFPs can enter the blood and move beyond the lung, although the extent to which this happens in humans remains unknown. These properties indicate a potential for adverse effects in the lung and in other organs.

Animal studies indicate that inhalation of UFPs at concentrations relevant to ambient air does not cause substantial lung inflammation. However, UFPs have been shown to enhance responses to allergens in allergen-sensitized and challenged animals, increase the progression of atherosclerosis in susceptible animal models, and influence the autonomic control of the heart. UFPs have been shown to translocate from the nose to the brain via the olfactory nerve, and there is evidence for increased inflammatory markers in the brain of exposed rodents.

Human chamber studies with exposure to carbon UFPs and to concentrated ambient UFPs have been fairly consistent in finding no effects on lung function or airway inflammation. Some chamber studies found UFP effects on vascular function, cardiac repolarization, HRV, and blood coagulation, suggesting that UFPs may have effects outside the lung in the absence of lung inflammation. Other chamber studies, however, have shown conflicting data. One real-world study showed declines in lung function and increased airway inflammatory markers in subjects with asthma.

Both animal and human studies provide evidence for respiratory and cardiac effects, and animal exposure studies suggest the possibility of effects on the brain. However, the ability to draw definitive conclusions is limited by the absence of long-term animal exposure studies and by somewhat inconsistent findings in human clinical studies. Human clinical exposure studies remain limited by the technology available to generate exposures relevant to ambient UFPs: laboratory-generated particles are not completely representative of ambient UFPs, and concentrator studies are limited to the larger, quasi-UFPs.

Real-world ambient exposure studies arguably offer the most realistic exposures to ambient UFPs, and some adverse respiratory and cardiovascular responses associated with those exposures have been observed. However, such studies always involve exposures to complex mixtures, and even with appropriate study designs, with current statistical methods it is challenging to separate completely the effects of UFPs from those of other pollutants. The results of these studies contribute to the trafficrelated air pollution and health literature by suggesting high exposure to traffic-related pollutants (and associated factors, such as noise and stress or anxiety) during commuting may be relevant for human health.

Collectively, the studies reviewed in this chapter do not provide strong evidence that short-term exposures to UFPs have effects that are dramatically different from those of larger particles; the effects of long-term, repeated experimental exposures to UFPs are unknown.

CHAPTER 4. Do UFPs Affect Human Health at Environmental Concentrations? What Is the Evidence from Epidemiologic Studies?

In the previous chapters, we have explored the sources and environments which could result in human exposure to UFPs, and the evidence for possible health effects deriving from controlled animal and human exposure studies. In this section, we examine the evidence base from epidemiologic studies that attempt to address directly the most central question of this issue of HEI Perspectives: Does ambient UFP exposure have an adverse effect on human health?

Our evaluation of the literature in this chapter considers this question from two angles: 1) an evaluation of the evidence for specific endpoints, with an assessment of the consistency and coherence of observed associations, and 2) an evaluation of the evidence with respect to key study design and data issues, including UFP measurement, exposure assignment approaches, and consideration of potential copollutant confounding. In doing so, we attempt to address the overarching question as to whether any observed UFP effects are independent of those observed for other particle sizes or for other combustionor traffic-related pollutants.

The intent of this document is to provide a broad survey of the state of the science on UFPs by describing the nature and scope of the current body of evidence. This survey summarizes the human health outcomes, exposure assessment approaches, and the ways that the studies account for the complex multipollutant exposure environment that often accompanies UFPs. In this chapter, we do not give an intensive quantitative meta-analysis or a more systematic, in-depth literature review. Instead, we will use this survey of studies to identify areas of investigation that are needed to more fully understand the specific effects of UFPs on human health, if any, and to guide recommendations for future exploration of human health and UFP exposure.

EVIDENCE BASE

We searched for all articles published through December 2011 that examined associations between UFPs and health using online databases (Web of Science and PubMed) and personal article collections. The U.S. EPA's 2009 Integrated Science Assessment for Particulate Matter (PM ISA; U.S. EPA 2009) was also used as a source for relevant articles.

We included for consideration any epidemiologic (i.e., observational) study utilizing one or more relevant UFP metrics (i.e., number, mass, or surface area concentrations) as the measure of exposure in the health model. For number concentration, while we focused on articles that assessed UFP NC for particles < 100 nm, we included some articles that measured total NC, where the size range is unspecified or varies between 3 and 1000 nm (but where the study authors expect that the total number of particles will be dominated by the number of particles in the < 100 nm range). Studies that measured only particles > $0.3 \mu m$ were excluded, as were studies with no particle count or particle size measurements. For example, studies of traffic using only distance to roadway measures or pollutant measures that were not specific to UFPs, such as $PM_{2.5}$ EC, were excluded. We also focused on studies involving exposure to ambient UFPs or ambient UFP surrogates and therefore excluded articles focused on nanotechnology and exposures to workplace engineered nanoparticles.

The following groups of articles were identified: 1) 8 relevant reviews published in the years 2009–2011; 2) an expert elicitation conducted to assess causality and concentration–response functions for UFPs and health; and 3) over 75 articles presenting primary research on the relevant UFP metrics. Overall findings from the first two groups of articles, including the 2009 PM ISA (U.S. EPA 2009), were used as a starting point for examining whether ambient UFPs affect human health. Studies published since the 2009 PM ISA were evaluated together with prior evidence to determine whether recent findings further our understanding of whether ambient UFPs adversely affect human health.

PREVIOUS REVIEWS

The 2009 PM ISA reviewed over 40 primary research articles that examined the effects of UFPs on health (published in years 2000–2009) (U.S. EPA 2009). This report found that in a limited number of epidemiologic studies the investigators observed associations of UFPs and acute respiratory effects, such as respiratory symptoms in infants (Andersen et al. 2008a) and in adults with asthma (Von Klot et al. 2002) as well as hospitalizations and emergency department visits for asthma and pneumonia (Andersen et al. 2008b; Halonen et al. 2008). However, associations were not observed in all studies, such as in a study in Atlanta of emergency department visits and high UFP concentrations (38,000/cm³) (Peel et al. 2005). Similarly for cardiovascular outcomes, the 2009 PM ISA reviewed a small number of epidemiologic studies and found inconsistent evidence for an association between UFPs and CVD hospital admissions, although some positive associations for subclinical cardiovascular measures (i.e., arrhythmias and supraventricular beats) were cited. Taken together with toxicologic findings, the 2009 PM ISA concluded that evidence of a causal relationship between short-term UFP exposure and respiratory or cardiovascular effects is suggestive. Evidence of causal relationships between UFP exposure and other acute outcomes (e.g., mortality and central nervous system effects) and chronic outcomes (e.g., cardiovascular and respiratory effects, reproductive and developmental effects, cancer, genotoxicity, and mutagenicity) was deemed inadequate.

An expert elicitation was also recently conducted to specifically assess the likelihood of causal relationships (separately and independently from effects of coarser particle fractions or other components of the air pollution mix) and the likelihood of potential causal pathways for cardiac events (Knol et al. 2009). A panel of twelve European experts (epidemiologists, toxicologists, and clinicians) rated the causality of health effects of short-term UFP exposure as medium to very high for all-natural-cause mortality, low to high for cardiovascular and respiratory hospital admissions, very low to medium for cough, low to high for aggravation of symptoms in asthma patients, and low to very high for decrements in lung function. The experts rated the likelihood of a causal relationship between long-term UFP exposure and health effects as ranging from low to very high, with most evidence deemed as indirect. Of the causal pathways for cardiac events evaluated, the pathway involving respiratory inflammation and subsequent thrombotic effects was rated as most likely; the pathway with the lowest ratings involved translocation of particles affecting the autonomic nervous system and affecting HR, HRV, and arrhythmia endpoints. Lower ratings by experts were motivated by issues such as reliance of studies on limited UFP data, exposure misclassification, lack of evidence for the independent effects of UFPs, lack of correction for publication bias, and lack of data on long-term UFP exposures.

These and other review articles struck similar themes regarding the limitations of available epidemiologic evidence for the effect of ambient UFPs on human health:

- The reviews each noted an inadequate base of evidence with which to assess different facets of the UFP health effects field: that is, the lack of coherent studies assessing specific disease, organ or system-based endpoints that would lead to specific mechanistic hypotheses (Araujo and Nel 2009; Lotti et al. 2009; U.S. EPA 2009), limited knowledge of the effect of exposures in specific microenvironments such as those related to commuting (Knibbs et al. 2011) or in school environments (Mejia et al. 2011), and the lack of studies assessing the effects of long-term exposures (U.S. EPA 2009; Knol et al. 2009; Hoek et al. 2010).
- 2. Several of the reviews addressed the difficulties in assessing UFP effects in epidemiologic settings. UFP monitoring data are scarce. Also, assigning UFP exposures using data from single ambient monitoring sites results in a high likelihood of exposure misclassification due to the high spatial variability of UFP concentrations (Fanning et al. 2009; U.S. EPA 2009; Terzano et al. 2010). In the expert elicitation (Hoek et al. 2010), exposure misclassification was identified as a high source of uncertainty in epidemiologic investigations into UFPs.
- 3. High covariation of UFPs with other combustion-related pollutants, such as CO and NO_2 , makes it difficult to disentangle the independent effects of UFPs from these pollutants or the traffic-related mix in general (U.S. EPA 2009; Knol et al. 2009).

For this document, we evaluated the primary research articles based on their ability to provide evidence that would help address these limitations.

LONG-TERM EFFECTS

There have been no studies that have examined long-term UFP exposure and health with the kinds of retrospective or prospective cohort study designs that have played influential roles in characterizing the chronic cardiorespiratory effects of PM_{10} and $PM_{2.5}$. The absence of such studies is directly related to the limited monitoring of UFPs over time and with insufficient spatial resolution to identify the spatial contrasts in population exposures on which long-term studies typically rely.

Instead, investigators have attempted to use crosssectional study designs that assess the prevalence of chronic diseases in relation to concentrations of UFPs in different locations within a particular period (Lwebuga-Mukasa et al. 2005; Cahill et al. 2011; Kim et al. 2011). For example, Kim and colleagues (2011) reported results of a survey of over 1900 schoolchildren from 12 schools in Korea for which indoor and outdoor school environmental measurements were conducted over a 7-day period in winter 2004. This study found associations of wheeze and asthma (in the previous year) with a number of home environment factors, such as water damage, visible mold growth, and indoor dampness. Among the outdoor measurements, associations of wheeze with NO₂ and current asthma with total NC were observed. Lwebuga-Mukasa and colleagues (2005) examined total NC in relation to asthma prevalence among Buffalo, New York, neighborhoods using a cross-sectional survey of over 1600 households. The authors found that total NC was highest in neighborhoods downwind of the Peace Bridge Complex, a commercial truck traffic corridor on Buffalo's west side; these neighborhoods also had the highest asthma prevalence in the study.

The cross-sectional design of these studies is a limitation. With no information on long-term personal exposures, it is not possible to attribute exposures to disease onset or progression. Nor is it possible to tease apart the potential effects of particle number from those of other potential causal factors (i.e., confounders) for asthma that may covary spatially with particle number. For example, in the article by Lwebuga-Mukasa and colleagues (2005), NCs were examined as one of several potential factors contributing to asthma prevalence or exacerbation; trafficrelated pollution, distance to source, home environmental conditions, and socioeconomic differences among neighborhoods in this region.

SHORT-TERM EFFECTS

Studies of the health effects associated with short-term exposures (e.g., hourly or daily) are easier to conduct and are far more common. We reviewed over 75 articles and reports for studies assessing the short-term health effects of ambient UFPs, including over 25 articles from the 2009– 2012 period that were not included in the 2009 PM ISA (U.S. EPA 2009). In this section the articles reviewed presented UFP exposure that was represented by number concentration data. A small number of studies that evaluated UFP mass concentration data are considered separately in a later section, *Epidemiologic Studies Using Measures of UFP Mass.*

Figure 22 provides an overview of the geographic distribution of the studies reviewed. It indicates that the large majority of the short-term studies reviewed to date have been conducted in Europe and that several European cities (e.g., Erfurt and Copenhagen) have been studied repeatedly in the literature. Moreover, European research activity on UFPs is concentrated primarily in western European countries. More details on the study design features of the primary short-term research studies reviewed, organized by the health endpoints and geographic location, are provided in Appendix Table B.1. The table includes more specificity on the UFP measurement methods and metrics used in the study (i.e., sized-differentiated number concentrations) than provided in this chapter. For simplicity, we have



Figure 22. Geographic locations of epidemiologic investigations of the short-term exposures to UFPs discussed in this chapter. A number of populations in a given location have been the subject of multiple articles.

reported results for total NC when that was the only measure of UFP provided, and for UFP NC, representing the NC size fraction closest to the definition used in this document (e.g., NC < 100 nm).

Mortality

Several population-based studies examining the association between short-term exposure to total or UFP NC and mortality that were conducted over the past 10 years compared ambient central site particle measurements to various mortality outcomes (e.g., all-natural-cause, respiratory, cardiovascular, or stroke mortality) using time-series or casecrossover approaches (Appendix Table B.1). Most of these studies considered various particle and gaseous pollutant metrics in association with mortality. In general, associations of NC and mortality were not consistently observed across the studies. Strong (significant) associations with NC were noted in the studies conducted in Erfurt, London, Rome, and Beijing and weak or no associations were observed in studies conducted in Helsinki and Prague. Closer evaluation of study designs and exposure characteristics in different study locations would be necessary to better understand sources of variability in the findings. The strongest associations with NC reported in most studies were largely for cardiovascular causes of death. Breitner and colleagues (2011) assessed associations of various particle size metrics (number, mass, and surface area concentrations) and cause-specific cardiovascular deaths in Beijing and reported that the strongest associations were between all cardiovascular and ischemic heart disease mortality and UFP NC, with weaker or null associations with larger size fractions and with other UFP metrics (see Figure 23).

In general, however, differences in the specific mortality outcomes and in the lag structure of the associations examined in the various studies make it difficult to fully assess the consistency of the results. In Erfurt 1995–2001, for example, robust associations between UFP NC and total mortality and a combined cardiovascular and respiratory mortality grouping were found for a 4-day lag (other lags in the 0–5 day range were not significant) (Stölzel et al. 2007) as well as for longer averaging periods (i.e., 6- and 15-day polynomial distributed lag models) (Breitner et al. 2009). In contrast, among the other studies that found associations with NC, strongest associations were observed for shorter lags (lag 0, 1, or 2) (Forastiere et al. 2005; Kettunen et al. 2007; Atkinson et al. 2010; Breitner et al. 2011).



Figure 23. Percentage change (and 95% confidence interval) in cause- and age-specific cardiovascular mortality per an interquartile increase in particle metrics in Beijing, from March 2004 to August 2005. The investigators measured particle number concentrations (NC) which they then converted into surface area concentrations (SC) and mass concentrations (MC) for the specific size ranges indicated (using assumptions found in the published paper). The interquartile ranges for the different metrics were: nucleation mode (NC < 0.03 µm), 10,203/cm³; Aitken mode (NC 0.03–0.1 µm), 6,250/cm³; NC < 0.8 µm, 13,790/cm³; SC 0.1–0.3 µm, 469.9 µm²/cm³; SC 0.3–0.8, 486.7 µm²/cm³; SC < 0.8 µm, 973.7 µm²/cm³; MC 0.1–0.3 µm, 24.0 µg/m³; MC 0.3–0.8 µm, 57.9 µg/m³; and MC < 0.8 µm, 81.8 µg/m³. (Source: Breitner et al. 2011, Figure 2, reprinted with permission from Elsevier.)

Moderate to high correlations between UFP NC and mobile-source related gases (CO, NO₂) have also made it difficult to implicate UFPs, as opposed to more general mobile-source emissions, in observed associations with health endpoints (U.S. EPA 2009). Indeed, in most papers, authors implicate combustion sources, as opposed to UFPs specifically, as affecting mortality. For example, while the authors did not specifically assess the effects of trafficrelated gases or particle components, Atkinson and colleagues (2010) indicated that NC in London is largely influenced by nucleation-mode particles from diesel traffic that have a high OC component. In their analysis, NO_x data were used to apportion PM measures into primary and nonprimary components; the authors observed a high correlation (r = 0.77) between total NC and primary PM₁₀. In Rome, observed associations were strong for both total NC and CO, with a high correlation (r = 0.89) observed between these pollutants (Forastiere et al. 2005). In Helsinki, associations were suggestive for UFP NC and CO in the warm, but not the cold, season (UFP NC–CO warm season correlation, r = 0.39) (Kettunen et al. 2007). Breitner and colleagues (2011) did not include CO and NO_x in the Beijing analysis, but the authors indicated that UFP NC (30-100 nm) was highly influenced by local traffic emissions.

Some studies have investigated multipollutant models of UFP NC with other particle or gaseous copollutants as a means of controlling for potential copollutant confounding. In Erfurt, associations of UFP NC and total mortality for the 1995-2001 period appeared to be independent of those for mobile-source-related gases (CO, NO, NO₂) (i.e., in two-pollutant models, the effects of UFP NC adjusted for individual gases were even slightly higher than those for UFP NC alone, as indicated in Figure 24, but did not change the interpretation of the UFP NC results); twopollutant models of UFP NC and mass concentration metrics (i.e., PM_{2.5} and PM₁₀, which showed no association with mortality in this study) were not reported (Stölzel et al. 2007). In Erfurt, daily mean UFP NC were moderately correlated with daily mean CO (r = 0.57), NO₂ (r = 0.65), $PM_{2.5}$ (*r* = 0.51), and PM_{10} (*r* = 0.56). In Beijing, associations of UFP NC and mortality were unchanged in models that included other NC or mass concentration metrics (e.g., nucleation mode NC in the 10-30 nm size range and accumulation mode mass concentration in the 100-800 nm size range); this study however did not consider CO or NO₂ (Breitner et al. 2011).

Cardiorespiratory Acute Morbidity

Population-based studies assessing associations of NC and acute morbidity (using data on emergency department visits, hospital admissions, physicians' visits, or emergency service calls) have mainly been conducted in Europe



Figure 24. Relative risk of mortality per interquartile range of UFP NC (NC 0.01–0.1µm), adjusted for gaseous pollutants in two-pollutant models. Erfurt, Germany, September 1995 to August 2001. These results indicate that accounting for the presence of gaseous copollutants did not substantially change the UFP NC relative risks of mortality 4 days post exposure (lag 4) in this study. Significant associations of UFP NC with mortality were found at this lag but not generally with shorter lags in this study (Source: Stölzel et al. 2007, Figure 3, reprinted with permission from Macmillan Publishers Ltd: Journal of Exposure Science and Environmental Epidemiology.)

(Appendix Table B.1). Similar to studies assessing mortality, daily ambient central site particle measurements were used in these studies in relation to measures of morbidity using time-series or case-crossover approaches. Most of these studies considered a variety of particle and gaseous pollutant metrics. A number of studies have been published since the 2009 PM ISA (U.S. EPA 2009); however, it is not clear that these later studies provide additional insight into the consistency of associations for specific endpoints or clarity on the potential for UFPs to exert effects independent of copollutants.

While many studies reported associations of morbidity with NC, observed associations within studies were generally outcome dependent, age-group dependent, or both, making it difficult to assess consistency across studies. For example, Andersen and colleagues (2008b) observed significant associations of 5-day mean total NC (6–700 nm) and respiratory disease admissions, but not CVD admissions, in subjects 65 years or older in Copenhagen; associations with total NC were nonsignificant for pediatric asthma in this analysis and remained nonsignificant and weaker than associations with other PM and gaseous pollutants in a follow-up study in Copenhagen that added four additional years of data (Iskandar et al. 2012). In a study in Helsinki, investigators also observed associations of 5-day mean UFP NC (30–100 nm) with pneumonia and a combination of other respiratory disease admissions in the population 65 years or older, but they found no associations with asthma or COPD admissions (Halonen et al. 2009) or for emergency department visits (Halonen et al. 2008) in this age group. Pediatric asthma emergency department visits, however, were associated with UFP NC (30–100 nm) at 3–5 day lags in Helsinki (Halonen et al. 2008). Differing study period lengths, daily outcome counts, and modeling choices also likely affect the pattern of observed associations by study and preclude attempts at meta-analysis.

Moreover, for most studies in which UFP effects were observed, associations for other particle measures (PM mass concentration and/or accumulation mode NC) or gaseous copollutants were also reported. Observed NC effects remained after controlling for PM₁₀ or PM_{2.5} in several studies (Von Klot et al. 2005; Lanki et al. 2006; Belleudi et al. 2010), however, they were diminished in other studies (Andersen et al. 2008b). Only two studies found independent associations with NC after controlling for CO, NO₂, or both (Andersen et al. 2010; Leitte et al. 2011). Leitte and colleagues (2011), however, indicated that NC associations were generally stronger when controlled for NO₂ than when controlled for other pollutants in two-pollutant models. In other studies, adjustment for the gases reduced the observed NC effect (Andersen et al. 2008b; Halonen et al. 2008), or the gases were not included with NC in twopollutant models (Von Klot et al. 2005; Lanki et al. 2006; Halonen et al. 2009). In some studies, traffic-related copollutants were not considered at all, making it difficult to draw conclusions about the independence of the observed NC effects (Atkinson et al. 2010; Belleudi et al. 2010; Braniš et al. 2010; Franck et al. 2011). Overall, similar to the acute mortality studies, most authors implicate particles from traffic-related sources in the observed NC effects (Von Klot et al. 2005; Lanki et al. 2006; Andersen et al. 2008b, 2010; Halonen et al. 2008; Atkinson et al. 2010).

Respiratory Effects

In addition to population-based studies assessing mortality and morbidity outcomes, numerous panel-based and individual-level studies have been conducted that examine associations of UFPs and cardiorespiratory health endpoints. In general, specific endpoints in these studies have been chosen to reflect and support hypothesized biological mechanisms of UFPs, as described in Chapter 3. In this section, we consider observational studies investigating respiratory-related endpoints; these have included measurements of respiratory symptoms (obtained through questionnaires) and pulmonary function (obtained through spirometry).

Respiratory Symptoms Studies of respiratory symptoms (e.g., wheeze, cough, phlegm, shortness of breath) in relation to total and UFP NC have been conducted in Europe (Appendix Table B.1). Study populations have varied by age group (children, adults, older adults) and pre-existing disease (asthma, other chronic conditions). Overall, the results of these studies are inconsistent, with some studies reporting significant NC effects and some not. Similar to the previous sections, in those studies in which NC effects were observed, most of these effects were not independent from those of other particle or gaseous measurements (Peters et al. 1997; Von Klot et al. 2002; Andersen et al. 2008a).

Pulmonary Function A number of studies, also largely conducted in Europe, have assessed pulmonary function in relation to total and UFP NC (Appendix Table B.1). The results of these studies have again been inconsistent, with a majority of studies observing few effects of UFP NC specifically. Many of these studies were conducted in the 1990s and considered ambient fixed-site particle measurements in relation to subjects' daily measurements of PEF, a measure of airflow obstruction during exhalation.

One recent study adds to the literature in terms of the exposure assessment approach utilized, with effects estimated using several different exposure metrics. De Hartog and colleagues (2010) conducted a multicity study in four European cities (Helsinki, Finland; Athens, Greece; Amsterdam, the Netherlands; Birmingham, UK) with 135 subjects with mild to moderate asthma or COPD. This study was specifically designed to assess the effect of exposure assignment approaches on observed epidemiologic results. The authors compared associations of total NC and lung function among different exposure metrics (ambient central site, home outdoor, and home indoor). Overall, no consistent associations were observed for any particle metric with lung function, even after various modeling specifications and controlling for medications use and lung function measurement time-of-day, or restricting the analysis to people with asthma. Furthermore, associations were not stronger when exposures measured in the subjects' homes were used in the analyses. The authors cite several potential explanations for the null findings in this study; for example, stated limitations of this study were the 1-week design (each subject monitored over just a 1-week period), which did not allow for assessment of lags longer than 2 days for the home-based metrics, and that 94% of subjects used respiratory medication. However, the detailed exposure characterization in this study mostly rules out exposure measurement error as a contributing factor to the null findings. Ultimately, the observed results could also be due to lack of a true association between UFPs and lung function.

Allergy and Atopy While allergy is often considered in the context of respiratory responses, Song and colleagues (2011) recently conducted a panel study in Incheon to examine associations of UFPs and other pollutants, with atopic symptoms in 41 elementary school children (ages 8-12) with atopic dermatitis. The authors reported a strong UFP NC effect that appeared to be independent of other pollutants: 1-day lagged UFP NC was associated with skin itching, but other pollutants were not (PM₁, PM_{2.5}, PM₁₀, NO₂, SO₂, and O₃). Associations remained in two-pollutant models with PM mass metrics. A limitation of this study was that PM measurements were conducted on the roof of the school while the gaseous pollutants were collected at an ambient fixed site located two kilometers from the school. Because the UFPs were measured closer to the school, it is possible that there was less exposure misclassification for UFPs compared with the gases in this study. Distance from the collection site may be a reason for the lack of effects observed with the gaseous pollutants.

Cardiovascular Effects

In addition to respiratory endpoints, panel-based and individual-level studies have considered a number of cardiovascular endpoints, such as ECG-related outcomes (e.g., HRV, arrhythmias, ischemia), vascular reactivity, BP, and soluble blood (and urinary) markers of systemic inflammation, coagulation, and oxidative stress. These endpoints broadly relate to autonomic nervous system, inflammatory, and oxidative stress-related pathways hypothesized as routes of action for ultrafine as well as for other particle size fractions.

Heart-Rate Variability A number of studies were found that examined associations of total NC or UFP NC with HRV, including time-domain (e.g., SDNN, r-MSSD) and frequency-domain (e.g., low-frequency [LF], high-frequency [HF], LF:HF ratio) endpoints obtained from analysis of ECG measurements (Appendix Table B.1). These studies were conducted in cities across Europe, North America, and Asia and were largely repeated-measures panel studies in which multiple measurements were taken in the same individual over time. The one exception was an analysis conducted as part of the Normative Aging Study cohort for years 2000–2003, which provided outcome data

for one sampling session per subject throughout the study period (Park et al. 2005).

These six studies considered a range of different exposure measurements. Five studies used ambient fixed-site NC measurements (Park et al. 2005; Timonen et al. 2006; Barclay et al. 2009; Schneider et al. 2010; Rich et al. 2012). One study incorporated personal total NC (20–1000 nm range) monitoring (Chan et al. 2004), and the study by Barclay and colleagues (2009) estimated personal UFP NC exposures based on ambient fixed-site measurements.

Associations of HRV effects with NC were not observed consistently: associations were observed in only three of the six studies (Chan et al. 2004; Timonen et al. 2006; Rich et al. 2012). A variety of factors may have contributed to the positive associations observed in these studies, such as the reduced uncertainty in exposure assignment due to the personal exposure characterization by Chan and colleagues (2004) and the relatively high statistical power of the multicity study design (the ULTRA study) by Timonen and colleagues (2006). It should be noted, however, that these studies also reported similar effects associated with exposures to other particle size fractions and gases.

Arrhythmia and Related Endpoints Studies of arrhythmias and total and UFP NC have also largely been conducted in Europe, with studies in Aberdeen, Augsburg, Erfurt, and London and one study conducted in the United States, in Boston (Appendix Table B.1). These studies have considered arrhythmias detected in individuals with implanted cardioverter defibrillators (ICDs), as well as arrhythmias detected in panel study subjects through analysis of ECG recordings.

The ICD studies focus on serious ventricular arrhythmias (e.g., ventricular tachycardia, ventricular fibrillation) that ICDs are designed to detect and treat (such as with pacing or shock). In a follow-up study of ICD patients living in eastern Massachusetts, Dockery and colleagues (2005a,b) found associations between ventricular arrhythmias and 2-day mean PM_{2.5}, BC, CO, NO₂, and SO₂, but not for total NC or O₃, with significant associations observed only when restricting analyses to arrhythmias occurring within three days of a previous arrhythmia. It should be noted that NC measurements were only available for one third of the study period; consequently, the null associations with NC may have been a function of limited data (and lower statistical power). The authors implicated traffic pollution in this study. In London, traffic-related pollutants (total NC, black smoke, CO, NO₂) assessed at 0-5 day lags were not associated with ICD-detected arrhythmias, whereas there were associations with secondary pollutants (e.g., sulfate) (Anderson et al. 2010). Null results for UFP NC and other pollutants, assessed at 0-2 day lags, were also found in the Aberdeen panel study, in which ventricular and supraventricular arrhythmias were assessed through ECG recordings (Barclay et al. 2009). In Erfurt, Berger and colleagues (2006) observed associations of arrhythmias with 5-day mean (and shorter single-day lags of) UFP NC, but also with accumulation mode NC, PM_{2.5}, CO, and NO₂.

Two other panel studies conducted in Germany assessed ECG-derived measures of repolarization abnormalities (e.g., QT duration, T-wave complexity, T-wave amplitude, T-wave amplitude variability), which may be linked with the onset of arrhythmias (Henneberger et al. 2005; Yue et al. 2007; Hampel et al. 2010). The results of these studies have been mixed. No associations with total NC at 0-5 day lags were found in a panel of myocardial infarction survivors in Augsburg (Hampel et al. 2010). Henneberger and colleagues (2005) observed significant decreases in T-wave amplitude with exposures to UFP NC, accumulation mode NC, and PM2.5 in the previous 5 and 23 hours of ECG measurements in a panel study of males with ischemic heart disease in Erfurt. Further analysis of these data suggested that local traffic-related UFP NC and diesel traffic-related source factors showed the strongest associations with repolarization parameters (Yue et al. 2007).

Ischemia Several studies have been conducted to examine the effects of particle exposure on myocardial ischemia using ST-segment changes from ECG recordings (Appendix Table B.1). These were conducted in panels of subjects with coronary artery disease in Helsinki and in Los Angeles. The two Helsinki publications focused largely on the same subject data, but assessed different lag structures: 1) Pekkanen and colleagues (2002) observed strong associations with 2-day mean UFP NC (and accumulation mode NC, PM₁, PM_{2.5}, CO, NO₂) and ST-segment depression during exercise tests; and 2) in an assessment of subdaily exposures, Lanki and colleagues (2008) observed associations with 1- to 4-hour lagged $PM_{2.5}$, but not with UFP NC. In Los Angeles, Delfino and colleagues (2011) found STsegment depression associated with home outdoor trafficrelated pollutant measures (including PM_{2.5}, PM_{0.25}, BC, primary OC, CO, NO₂) at various lags (including 1-hr, 8-hr, and 1–3 day means), but not for total NC (r = 0.36 between total NC and PM_{0.25}) in a panel of older subjects living in a retirement community.

Vascular Reactivity Only two studies have considered vascular reactivity in relation to exposure to particle count measures (Appendix Table B.1), and both studies have found little association with NC. In a panel of subjects with diabetes or at risk for diabetes in Boston, O'Neill and

colleagues (2005) examined associations of ambient particle concentrations and two measures of vascular reactivity: non-endothelium-dependent nitroglycerin-mediated reactivity and endothelium-dependent flow-mediated reactivity. Total NC was associated with nonsignificant decreases in both measures, while other particle metrics ($PM_{2.5}$, BC, and sulfate) showed significant inverse associations. Dales and colleagues conducted a *bus stop study* in a panel of healthy subjects who were asked to sit at one of two bus stops in Ottawa, Canada, for two hours (Dales et al. 2007). Associations between $PM_{2.5}$ and flow-mediated dilatation were observed, but not with NO_2 , total NC ($\leq 1 \mu m$), or traffic density.

Blood Pressure A limited number of studies have considered changes in HR and BP, which both reflect changes in autonomic tone, in relation to NC measures. These studies have been conducted in Europe as part of the multicity ULTRA study (Amsterdam, Helsinki, and Erfurt), in Los Angeles, in Ottawa as part of the bus stop study, and in Taipei, Taiwan (Appendix Table B.1). The results to date have been inconsistent in regard to both the direction and significance of observed associations.

In a small panel of subjects with lung function impairments, Chuang and colleagues (2005) observed positive associations between personal exposures to total NC (20-1000 nm) and HR and both systolic and diastolic BP in the 1-3 hours before the BP measurement. In contrast, in the ULTRA multicity study of adults with coronary heart disease, small but significant inverse associations between particle measures (PM_{2.5}, NC_{0.01-0.1}, and accumulation mode particles $[NC_{0,1-1,0}]$) at 0–2 day lags and systolic and diastolic BP were found in pooled analyses (Ibald-Mulli et al. 2004). In the panel study of older subjects with coronary heart disease living in retirement homes in Los Angeles, outdoor home measurements of total NC, PM_{2.5}, OC, BC, and gases were conducted (Delfino et al. 2010c). In overall analyses, the authors observed positive associations of systolic and diastolic BP with all particle measures, at 3–9 day mean concentrations, except for total NC. In effect modification analyses, associations with 1-8 hour lagged total NC were observed when subjects reported moderate to strenuous physical activity in the hour preceding the BP measurement. In the bus stop study, only NO2 was associated (positively) with 2-hour bus stop exposures (Dales et al. 2007).

Soluble Markers A relatively large number of articles (n = 16) from eight different studies have considered total or UFP NC in relation to various blood (and urinary) markers of systemic inflammation, coagulation, and oxidative stress, pathways that are linked with processes of CVD and atherosclerosis (Appendix Table B.1). The large range

of specific markers examined across these studies makes it difficult to fully assess the consistency of effects. However, there have been a handful of common markers used across studies, including C-reactive protein (CRP) and interleukin-6 (IL-6) as markers of systemic inflammation, and fibrinogen as a marker of coagulation.

Of the eight studies assessing endpoints for CRP, IL-6, or both, four studies observed associations with various measures of number concentration: 1) a population-based cohort study in Germany (Hertel et al. 2010); 2) a European multicity study of myocardial infarction survivors in Athens, Greece; Augsburg, Germany; Barcelona, Spain; Helsinki, Finland; Rome, Italy; and Stockholm, Sweden (as part of the AIRGENE study) (Rückerl et al. 2007a); and two well-characterized panel studies in 3) Erfurt (Rückerl et al. 2006; Rückerl et al. 2007b; Yue et al. 2007) and in 4) Los Angeles (Delfino et al. 2008, 2009). The first of these studies (Hertel et al. 2010) used a dispersion and chemical transport model to determine spatiotemporally-resolved total NC, but did not consider pollutants other than NC, PM_{2.5}, and PM₁₀ in the analysis. The other three studies, however, in addition to observed associations with NC metrics, found associations of CRP, IL-6, or both with accumulation mode NC, CO, NO2, or other traffic-related pollutants (such as EC in the Los Angeles study). Twopollutant models of NC and other pollutants were not examined in these studies.

The four studies in which no significant associations were found between NC metrics and CRP or IL-6 included panel studies of: 1) adults with a previous myocardial infarction in Augsburg (Kraus et al. 2011) (although this study did find an association with plasma Lp-PLA2, a marker of vascular inflammation [Brüske et al. 2011]); 2) adult men with chronic pulmonary disease in Erfurt (Hildebrandt et al. 2009); 3) adults with stable chronic heart failure in Aberdeen (Barclay et al. 2009); and 4) adult men in the VA Normative Aging Study in Massachusetts (Zeka et al. 2006). It is difficult to generalize the differences between the positive and null studies here. However, the studies with null results for CRP were largely null not only with respect to particle number, but also with other pollutants examined. Such findings suggest that overall study design factors such as sample size or the exposure contrasts that were considered may have made detection of any underlying effect difficult.

Seven of the eight studies also assessed fibrinogen levels or other blood markers of coagulation in relation to various NC measures. The results for these endpoints have been very inconsistent; only two studies observed associations between NC and fibrinogen effects (Zeka et al. 2006; Hildebrandt et al. 2009), and studies that evaluated a range of coagulation endpoints have generally observed mixed results among the endpoints examined (Rückerl et al. 2006; Delfino et al. 2008; Hildebrandt et al. 2009).

In the Erfurt panel study, Rückerl and colleagues (2006) examined multiple blood biomarkers in a panel of coronary heart disease patients and observed associations between ambient fixed-site PM₁₀, UFP NC, accumulation mode NC, CO, and NO₂, and increased markers of inflammation (CRP) and adhesion (ICAM-1) above the 90th percentile, with the strongest associations using 2-day lagged pollutant measures; results for markers of coagulation were inconsistent (e.g., factor VII, fibrinogen, D-dimer) (Rückerl et al. 2006). In a further analysis of this panel, the authors observed associations between 0-23 hours mean UFP NC and accumulation mode NC with plasma levels of sCD40L, a marker for platelet activation (Rückerl et al. 2007b). In contrast, however, associations of 0-5 day lagged UFP NC with CRP and ICAM-1 were weak in another Erfurt panel study of male chronic pulmonary disease patients (Hildebrandt et al. 2009); in this study, associations of ambient UFP NC at 1- and 3-day lags and 5-day mean concentrations were observed for fibrinogen. The differences in findings between these two studies may be due in part to the differences in underlying health status of the patient populations examined, and may also point to limitations in the generalizability of results of small panel studies.

Delfino and colleagues have published several articles describing associations of total NC and other pollutants on blood markers and other endpoints in a panel of subjects with a history of coronary artery disease recruited from four retirement homes in the Los Angeles air basin (Delfino et al. 2008, 2009, 2010c). Overall, this study has included the most detailed health and exposure characterization of the studies published to date. The health endpoints characterized have included biomarkers of inflammation, coagulation, and oxidative stress, and the exposure characterization has included unusually detailed home outdoor and indoor measurements of UFPs, including total NC, PM_{0.25}, and PM_{0.25} components. (See section entitled Epidemiologic Studies Using Measures of UFP Mass for a discussion of PM_{0.25} epidemiologic results.) The results from this study have pointed to consistent associations between UFPs measured as total NC and inflammatory markers (IL-6, CRP, and sP-selectin); however, associations have also been strong for other traffic-related pollutants and components (e.g., EC, primary OC, CO, NO₂), and the authors implicate traffic emissions in their findings (Delfino et al. 2008, 2009). In contrast to the observed strong associations with the inflammatory markers, however, the results for markers of coagulation have been mixed, and the results for a measure of oxidative stress

(Cu, Zn-super oxide dismutase) have been internally inconsistent. For example, the authors observed negative and positive responder groups, with negative associations observed for certain subjects and positive associations observed for others (Delfino et al. 2008, 2009).

EXPOSURE ASSESSMENT CONSIDERATIONS

Our review of the epidemiologic literature identified several issues affecting the assessment of exposure to UFPs that are likely to have contributed to inconsistencies in observed results between studies as well as to uncertainties and limitations in assessing more specifically the contribution of UFPs to adverse health outcomes.

Copollutant Confounding

Previous reviews (e.g., U.S. EPA 2009), and many primary research articles examined as part of this evaluation, indicate that high covariation of UFPs with other combustion-related pollutants, such as CO and NO₂, makes it difficult to disentangle the independent effects of UFPs from these pollutants or the traffic-related mix in general. To illustrate this point with the currently available literature, of 42 published articles that cited any significant NC-health associations, 37 articles also noted significant effects for other particle or traffic-related pollutants and 10 articles did not consider traffic-related gases at all in the analysis (see Appendix Table B.1). Two-pollutant models were considered in some studies to assess the independence of UFP effects from other pollutants. Observed effects of NC measures did hold in two-pollutant models with other particle measures (e.g., accumulation mode NC or PM) and CO or NO₂ in some studies (8 and 3 studies, respectively). However, 5 studies reported that observed effects of NC measures did not hold in two-pollutant models with copollutants, and 23 of these studies did not consider NCs in two-pollutant models with copollutants. Multipollutant models are inherently difficult to interpret and may not be appropriate to implement with pollutants that are considered surrogates of the same source (e.g., traffic in this case). Thus, in most analyses to date, the independence of UFP effects cannot be clearly determined in the available observational study designs.

Exposure Measurement Error

Most studies assessing UFP health effects, especially the population-based studies of acute mortality and morbidity as well as many panel-based studies assessing clinical and subclinical endpoints, have utilized ambient fixed-site measurements of UFPs to represent exposures to individual study subjects. Depending on the study design and the nature of spatial and temporal variability in UFP concentrations in a study area, this approach may lead to varying degrees of error in how well individual exposures are represented (see related discussion in Chapter 2). These types of error in exposure measurement can, in turn, affect the ability of epidemiologic studies to detect associations with health outcomes.

Population-Based (Time-Series) Studies For populationbased studies of short-term exposures as in daily time-series studies, exposure contrasts are temporal and assessment of short-term average population exposure for the entire study area is necessary. In these studies, ambient fixed-site monitoring data may be adequate as the measure of exposure if the temporal variability in UFP concentrations at the fixed site represents the temporal variability of concentrations over the study area. However, it is important that assessments of both spatial and temporal variability of UFPs at a range of locations within the study area be provided to assess the suitability of reliance on the fixed-site monitor for this purpose.

How well exposures are characterized for the individuals in the study can affect the strength, and in some cases the direction, of the associations that can be observed in a study. Error in exposure estimates can often bias associations between exposures and health outcomes to the null and can be one explanation for lack of observed positive associations in a study (Atkinson et al. 2010). In other cases, exposure measurement error can lead to spurious associations.

Studies indicating strong temporal correlation among monitoring sites for NC, despite differences in absolute concentrations among sites, might reassure us that the use of ambient fixed-site monitoring data in these studies is adequate for detecting underlying associations in time-series epidemiologic studies. For example, Cyrys and colleagues (2008) documented strong site-to-site temporal correlations (r > 0.80) for ambient UFP NC in Erfurt, Germany. However, study locations likely differ in the level of spatiotemporal variability of UFPs, and need to be evaluated directly for each study location and period. In studies that observe associations only with larger particle measures, it is possible that null findings for UFPs may be due to greater exposure error for the UFP measurements than for other size fractions (Pekkanen and Kulmala 2004).

Panel Studies Assessment of exposure in panel studies that rely on ambient fixed-site monitoring data can be more complex than for population-based studies. Ambient fixed-site measurements in these studies are less likely to accurately describe the temporal variability of exposures

for each individual, depending on their time–activity patterns and proximity to local sources, and more precise measures of exposure may be needed. For $PM_{2.5}$, results of detailed exposure assessment studies suggest that for many individuals an ambient monitoring site can adequately represent temporal variability in exposures to ambient $PM_{2.5}$ (Ebelt et al. 2000; Janssen et al. 2000; Sarnat et al. 2000, 2006b). However, there have been no studies that have directly assessed the relationship between personal exposures and ambient concentrations for UFPs.

Hoek and colleagues (2008) have approached this question in their investigation of ambient (central site), outdoor home, and indoor home concentrations of total NC, $PM_{2.5}$, PM_{10} , soot, and sulfate for over 150 homes across four European cities. In their analysis of 24-hour average data, correlations of central site to indoor concentrations were lower for total NC (r = 0.16-0.45) than for $PM_{2.5}$ (r = 0.40-0.80) and sulfate (r = 0.91-0.99). These analyses suggest that ambient fixed-site monitoring data for NCs may be less representative of individuals' exposures than they are for $PM_{2.5}$.

Addressing Potential Exposure Error To overcome some of the concerns of exposure error in both types of studies, some investigators have considered *capture area analyses* (O'Neill et al. 2005; Stölzel et al. 2007; Andersen et al. 2008a; Belleudi et al. 2010), where associations are reassessed in only the population that resides within a limited distance from the monitoring site. This restriction made a difference in the study by Andersen and colleagues (2008a): total NC was not significantly associated with daily wheezing in infants in the overall analyses (i.e., including children living within a 15-km radius of the monitor), but associations were significant when the analysis was limited to children living within a 5-kilometer radius of the monitor (Andersen et al. 2008a).

In panel studies, populations have also been chosen on the basis of their residence near the fixed-site monitor of interest — within 2 kilometers (Penttinen et al. 2001a,b; Song et al. 2011), 5 kilometers (Pekkanen et al. 2002; Lanki et al. 2008), or 10 kilometers (Anderson et al. 2010) — of the ambient monitoring site. It is difficult to assess the contribution of these geographical restrictions to detection of epidemiologic associations, however. Effects with NC measures were observed in only three of the six cited studies; these variable findings may be attributable to a range of factors other than exposure error (such as study population and outcome investigated).

Other efforts have been made to improve UFP field monitoring campaigns for panel studies. Several of the more recent panel studies have utilized more comprehensive exposure characterization, including school-based outdoor measurements (Song et al. 2011), home outdoor and indoor measurements (Delfino et al. 2008, 2009, 2010b,c; de Hartog et al. 2010), as well as personal exposure monitoring of NCs (Chan et al. 2004; Chuang et al. 2005). Many, but not all, of these studies reported associations between NC measures and the respective outcomes of interest.

Several of these articles allow for a comparison of epidemiologic results obtained from different exposure assessment approaches. In a European multicity study, de Hartog and colleagues (2010) found no consistent associations between total NC and pulmonary function, regardless of whether central site, home outdoor, or home indoor measurements were applied. Delfino and colleagues (2008), in their study of retirement home subjects, found stronger associations between total NC and markers such as IL-6 and CRP when using outdoor compared with indoor concentrations (note that indoor UFPs may have different size and chemical composition than outdoor UFPs). Associations using estimated indoor NC of outdoor origin, however, were more similar to those observed for outdoor concentrations. Figure 25 compares, for example, the results for IL-6 using outdoor and indoor NC (including indoor NC of ambient origin). The authors suggest that measurements recorded at outdoor home locations may be adequate to capture outdoor air pollution-cardiovascular health associations.

Epidemiologic Studies Using Measures of UFP Mass

Our review has until this point focused on studies that evaluated associations between UFPs characterized by number concentration measurements and health outcomes. As discussed in the Chapter 2 of this document, scientists have also characterized UFP concentrations using different measures of UFP mass. Measurements of reconstructed mass (PM_{0.1}) that more specifically target UFPs in the size range of interest for this document have been used to study source-apportionment of UFPs. However, these methods are only just beginning to be used in epidemiologic studies, and results have not yet been published. Some of the most detailed work done to date with PM mass has involved a larger particle size cut point, 250 nm, or PM_{0.25} which is included in the class of particles commonly referred to as quasi-UFPs. Although these studies are not ideal for shedding light on UFPs (< 100 nm), we discuss them here in part because the investigators have made reasonable efforts to include data on particle number count and other air pollutants in their analyses. This work is also notable for providing some data on the composition of these smaller particle size fractions.



Figure 25. Comparison of the relationships of IL-6 (pg/mL), a biomarker of inflammation, to outdoor and indoor air pollutants. The figure plots the estimated change (adjusted coefficient and 95% CI) in IL-6 corresponding to an interquartile range change in the average air pollutant concentration for the previous day (lag 0), or for the previous several days preceding the blood draw). (Est Coef signifies estimated coefficient. For indoor measurements of EC, primary OC (OCpri), secondary organic aerosol (SOA), and PN, the symbol o_o signifies indoor concentrations of outdoor origin). (Source: Delfino et al. 2008, Figures 1 and 2, reproduced with permission from Environmental Health Perspectives.) (*Figure 25 continues on next page.*)

Pollutant	Lag	Ν	Est Coe	f Indoor Measurements
PM _{0.25}	Lag 0 4 d ave.	184 197	0.02 - -0.09 -	
PM _{0.25-2.5}	Lag 0 4 d ave.	186 197	-0.12 -0.46	
PM _{2.5-10}	Lag 0 4 d ave.	241 226	-0.05 -	
EC	Lag 0 3 d ave. 9 d ave.	195 187 167	-0.02 - 0.30 - 0.59 -	
EC o_o	Lag 0 3 d ave. 9 d ave.	195 231 193	0.20 - 0.37 - 0.76 -	
OC	Lag 0 3 d ave. 9 d ave.	195 187 167	-0.03 - 0.10 - -0.40 -	
OC _{pri} o_o	Lag 0 3 d ave. 9 d ave.	207 241 226	0.20 - 0.30 - 0.83 -	
SOA o_o	Lag 0 3 d ave. 9 d ave.	207 241 226	-0.07 -0.12 -0.17	
PN	Lag 0 3 d ave. 9 d ave.	205 205 194	0.31 - 0.28 - 0.29 -	
PN o_o	Lag 0 3 d ave. 9 d ave.	181 175 154	0.50 - 0.48 - 0.72 -	
NO ₂	Lag 0 3 d ave. 9 d ave.	241 226 226	0.40 - 0.42 - 0.43 -	
со	Lag 0 3 d ave. 9 d ave.	241 241 241	0.54 - 0.47 - 0.77 -	
			-1.	5 –1.0 –0.5 0.0 0.5 1.0 1.5 2.0

Figure 25 (Continued).
Two panel studies have assessed PM_{0.25} mass concentration, either alone or alongside number concentration or other PM data. One is a study of patients with a previous myocardial infarction in Italy in which negative correlation was found between HRV and exposure to PM_{0.25} in a group of patients not taking beta-blockers. More severe ventricular arrhythmias were observed at the highest concentrations of PM₁₀ and PM_{2.5}. Indexes of inflammation in either breath condensate or blood did not correlate with PM exposures (Folino et al. 2009). The other is a series of panel studies of retirement home subjects with a history of coronary artery disease by Delfino and colleagues in Los Angeles, which has included perhaps the most detailed work to date on quasi-UFPs in an epidemiologic setting (Delfino et al. 2008; 2009; 2010a,b; 2011). Overall, the authors have observed associations between outdoor and indoor home PM_{0.25} and ECG and blood marker outcomes. The group's recent analyses examined outdoor and indoor home measurements of $PM_{0.25}$ and $PM_{0.25}$ components (PAHs, hopanes, n-alkanes, organic acids, water-soluble

OC, and transition metals) (Delfino et al. 2010b). The authors reported that strong associations with inflammatory markers (IL-6 and sTNF-RII) were observed with PAHs, and that associations of total $PM_{0.25}$ were confounded by $PM_{0.25}$ PAHs (see Figures 26 and 27). In a further analysis, the authors found both IL-6 and exhaled NO, a marker of pulmonary inflammation, were associated with $PM_{0.25}$ oxidative potential, which was assessed via reactive oxygen species generation in the in vitro cellular assays (Delfino et al. 2010a).

This study also included measurements of total NC, and therefore affords a comparison of results between the different UFP metrics. For example, in examination of the inflammatory markers the authors found associations with outdoor home measurements of both total NC and $PM_{0.25}$ (Delfino et al. 2009). In another analysis, however, the authors found ST-segment depression associated with outdoor home measurements of $PM_{0.25}$ (and other trafficrelated pollutant measurements), but not with measurements of total NC (Delfino et al. 2011).



Figure 26. Associations of biomarkers of inflammation with 5-day average outdoor and indoor concentrations of PM_{0.25} mass, and markers of primary organic aerosols (POAs) and secondary organic aerosols (SOAs) in subjects from a retirement community in Los Angeles, CA. The figure plots the expected change (adjusted coefficient and 95% CI) in A: IL-6 and B: sTNF-RII, corresponding to an IQR increase in the air pollutant concentration, adjusted for temperature. (Source: Delfino et al. 2010b, Figure 1, reproduced with permission from Environmental Health Perspectives.)



Figure 27. Associations of circulating biomarkers of inflammation with outdoor $PM_{0.25}$ mass coregressed with outdoor total PAHs and hopanes in $PM_{0.25}$ for subjects in a retirement community in Los Angeles, CA. A: IL-6, PAHs, and $PM_{0.25}$. B: sTNF-RII, PAHs, and $PM_{0.25}$. C: sTNF-RII, hopanes, and $PM_{0.25}$. Estimated change in the biomarker (adjusted coefficient and 95% CI) corresponds to an IQR increase in the air pollutant concentration, adjusted for temperature. (Source: Delfino et al 2010b, Figure 2, reproduced with permission from Environmental Health Perspectives.)

SUMMARY AND CONCLUSIONS FROM EPIDEMIOLOGY

For this issue of HEI Perspectives on UFPs, we reviewed both older summary reviews of the UFP epidemiologic literature and relevant primary research articles that have been published in the interval after the compilation of the 2009 EPA PM ISA (U.S. EPA 2009). A growing number of studies have attempted to assess the health effects of UFPs, either as their main focus or as one of several pollutants of interest. However, for reasons summarized below, we have found that the evidence to date continues to lack consistency and coherence with regard to our overarching question of whether ambient UFPs affect human health differently or independently from the effects of other particle or gaseous copollutants.

Inconsistency of Results by Endpoint

Previous review articles have noted that the current evidence base lacks a coherent set of studies designed to address specific hypotheses about the specific health endpoints (Araujo and Nel 2009; Lotti et al. 2009). While a growing number of studies have considered the effects of short-term UFP exposure, the consistency of effects for any one endpoint is still lacking. For both respiratory and cardiovascular outcomes that are assessed here, studies continue to show inconsistent results, with some studies reporting associations with UFP exposure (e.g., Von Klot et al. 2002; Andersen et al. 2008a; Song et al. 2011) while others do not (e.g., de Hartog et al. 2003, 2010; Timonen et al. 2004).

The inconsistencies in observed associations and lag structures may be due to a number of factors, including differences in study designs, populations examined, data availability and UFP metric utilized, differential measurement error across studies, different model strategies and confounder control (e.g., weather), and possibly differences in pollutant composition, concentration, or a combination of composition and concentration that might influence health risk. Studies based on small panels of subjects are limited in their generalizability, which likely also contributes to the lack of consistent effects across the small but growing numbers of epidemiologic studies of UFPs. Ultimately, as in any study, one explanation that must be considered is that a true underlying association does not exist. However, the meta-analysis necessary to more fully test this hypothesis would be difficult to implement, given the current study design differences across the available epidemiologic literature.

Exposure Assessment

While research on UFPs and human health effects appears to be improving over time with regard to the quality of exposure measurements (e.g., improved measurement equipment, exposure assessment with multiple monitors), most studies lacked significant conclusions regarding the potential effect of exposure measurement error on study results. UFP concentrations are known to be highly spatially variable within cities, yet many city-wide assessments of UFPs do not account for this high variability. Short-term studies of UFPs and health effects may avoid the spatial error component by analyzing temporal variations, but the assumption that temporal variations within a city are spatially uniform is not explicitly evaluated. As such, there is a concern that null findings for health effects of UFPs may be the result of exposure measurement error. In one study where investigators specifically attempted to improve the accuracy of the NC exposure assessment, however, associations between total NC and human health effects were still not observed, possibly due to other study design considerations (de Hartog et al. 2010).

Assessment of the Independence of UFP Effects

Where positive associations have been observed with UFPs, studies have not generally shown independent effects of UFPs, either in the absence of other exposures or in models adjusting for expected copollutant effects. While associations between UFPs and human health effects were observed for some outcomes, many studies did not account or adjust for the potential associations with gases or other particle metrics (even when effects of those other particle metrics or gases were observed), or potential copollutant exposures were not addressed or even included in the analysis. While proximity to traffic and other markers of traffic exposures imply exposure to UFPs, it is very challenging to separate the potential health effects of UFP exposure from the potential health effects of other exposures associated with traffic such as NO_2 , CO, or noise.

This survey of the literature was intended to assess the state of the literature with the regard to the health effects that are potentially associated with UFP exposure. Despite a growing evidence base of observational studies of UFPs and improving measurement and exposure assessment approaches, there remain inconsistencies in reported results between studies of the same or similar health endpoints and suggestive, but not definitive, research findings. Research on the long-term exposure effects of UFPs is particularly absent from the literature. Given the emerging understanding of spatial and temporal exposure variation, the potential role of copollutants, evolving measurement methods and technology, and unclear physiologic mechanisms of action, the epidemiologic findings do not identify definitive, reproducible human health effects that are uniquely associated with UFP exposure.

CHAPTER 5. Summary and Conclusions

SUMMARY

Ever since the hazards of air pollution were first identified, scientists and policy makers have sought to identify those constituents of the air pollution mixture that might explain, in whole or in part, the adverse effects that have been observed. Over two decades ago, epidemiologic studies began to find that PM was associated with increased mortality and morbidity, but the underlying biological mechanisms for such relationships were the subject of much speculation. About the same time, researchers hypothesized that the components of PM, including UFPs, could be responsible for the adverse effects of PM and of the air pollution mixture in general. The motivation for research on these hypotheses was not just to understand the underlying mechanisms but to help ensure that efforts to control exposures to air pollution were targeted effectively on those components of most relevance to public health.

A substantial body of literature has now been published on the sources and generation of UFPs, their spatial and temporal distribution in ambient air, their inhalation and fate in the body, their mechanisms of toxicity, and their adverse effects in animals and in humans. The purpose of this issue of HEI Perspectives on UFPs has been to provide a broad assessment of what has been learned and what remains poorly understood. We structured our assessment of this literature and its ability to inform an answer to the overall objective as responses to three questions:

1. Ambient UFPs — sources, emissions, and exposures: To what extent do motor vehicles contribute?

As products of combustion and secondary atmospheric transformations, ambient UFPs have multiple sources whose relative contributions to ambient concentrations varies with location, season, and time-of-day. However, in urban areas, particularly in proximity to major roads, motor vehicle exhaust can be identified as the major contributor to UFP concentrations. Diesel vehicles have been found to contribute substantially, sometimes in disproportion to their numbers in the vehicle fleet. However, the absolute and relative contributions of different vehicle types to motor vehicle emissions is changing rapidly with changes in fuels, engine, and exhaust aftertreatment technology. The collective effect of all these changes has not been thoroughly explored and is likely to vary regionally, depending on the rate and extent to which they are deployed in different parts of the world.

It has been more challenging to characterize human exposure to ambient UFPs than to the more regionally dispersed and routinely monitored pollutants, such as PM_{2.5}. UFP concentrations are not routinely monitored, and most monitoring in studies relies primarily on measures of total NC and to a lesser extent on size-differentiated number concentrations. Particle number counts tell us little about other characteristics of UFPs, such as surface area, surface reactivity, or chemical composition, which may be of interest in understanding health effects. In addition, high covariation exists between UFPs and other combustionrelated pollutants, such as CO and NO_x, near sources such as traffic. Furthermore, UFP NCs often differ substantially from one location to another in the same city. Consequently, reliance on measurements at central site monitors to represent broad population exposure, for example, across an entire metropolitan area — a central feature of epidemiologic of studies of long-term exposures to PM_{2.5} and other pollutants — is more likely to lead to misclassification or errors in determining UFP exposure.

However, UFP NCs measured at multiple locations within cities do tend to vary temporally in similar patterns over the course of a day. Moderately good temporal correlations between UFP concentrations at central monitors, outdoors at residences, and even indoors at residences have been observed in some but not all cities. The correlations are not always as strong as those observed for $PM_{2.5}$, but in some locations they can be sufficient to support epidemiologic studies of the effects of short-term variations of NCs on human health, with study designs that have been useful in studies of larger size fractions. However, the temporal variability in UFP NC is likely to be similar to that of other PM size fractions and gaseous pollutants, making it difficult to differentiate the effects of UFP NC in such study designs.

2. Do UFPs affect health? What is the evidence from experimental studies in animals and humans?

Experimental studies have provided evidence to indicate that, as a result of their physical characteristics, inhaled UFPs differ from larger particles in their deposition patterns in the lung, their clearance mechanisms, and in their potential for translocation from the lung to other tissues in the body. Some animal studies have also demonstrated translocation of UFPs via the olfactory nerve to the brain. Taken together, these findings provide a rationale for the hypothesis that the adverse health effects of exposure to UFPs differ from those of larger particles.

Both animal and human studies provide evidence for respiratory and cardiovascular effects associated with exposure to UFPs. Observed effects in selected studies include lung function changes, airway inflammation, enhanced allergic responses, vascular thrombogenic effects, altered endothelial function, altered heart rate and heart rate variability, accelerated atherosclerosis, and increased markers of brain inflammation. With the exception of brain effects, the findings are largely similar to those observed for exposures to fine particles.

There are limitations and inconsistencies in the findings on UFP health effects. There are no long-term animal exposure studies of UFP health effects. Relatively few studies have directly compared UFPs with other particle size fractions. The somewhat inconsistent findings in human controlled exposure (chamber) and real-world studies discussed in Chapter 3 likely result in part from differing outcome measures, as well as limitations in measurements, study designs, and statistical power. Furthermore, clinical studies of exposure to UFP proxies, such as laboratory-generated UFP or concentrated ambient UFP, may not accurately reflect the effects of exposure to actual ambient UFP under real-life conditions. On the other hand, the real-world studies of exposure to ambient UFPs face the challenges of disentangling the health effects of UFPs from other traffic-related pollutants.

While selected studies show evidence for UFP effects, the current evidence, when considered together, is not sufficiently strong to conclude that short-term exposures to UFPs have effects that are dramatically different from those of larger particles. The limitations of the experimental data, and the absence of long-term exposure studies in animals or humans, constrain our ability to draw definitive conclusions about the consequences of exposure to UFPs.

3. Do UFPs affect human health at environmental concentrations? What is the evidence from epidemiologic studies?

Epidemiologic studies have provided suggestive, but often inconsistent, evidence of adverse effects of shortterm exposures to ambient UFPs on acute mortality and morbidity from respiratory and cardiovascular disease. One explanation that must be considered for the results to date is weakness in the true underlying relationship between UFP exposures and adverse effects — that the null hypothesis being tested by these studies is true. However, limitations of the current studies are likely to play a role: UFPs have not been assessed routinely in larger epidemiologic studies of air pollution health effects, in part because ambient monitoring of UFPs is not conducted in most locations; UFPs have been defined and measured in different ways; and the greater exposure measurement

error for UFPs relative to PM_{2.5} and other pollutants makes it difficult to design epidemiologic studies with sufficient statistical power to test confidently for what may be small, but important health outcomes. The available observational study designs have also not been able to clearly determine whether UFPs have effects independent of those for related pollutants. Where studies have measured UFPs, few have actually assessed whether the effects associated with UFPs are independent of other pollutants. When they have, the effects of UFPs have not been consistently discernible from those of other pollutants with which they often occur or share similar sources (e.g., traffic). Of 42 published articles that cited any significant health associations with UFPs measured as NC, 37 articles also noted significant effects for other particle size fractions or traffic-related pollutants, and 10 articles did not consider any traffic-related gases in the analysis. It should be noted that multipollutant models are inherently difficult to interpret and may not be appropriate to implement with pollutants that are considered surrogates of the same source (e.g., traffic in this case).

No epidemiologic studies of long-term exposures to ambient UFPs have been conducted, as the most common epidemiologic study designs are dependent on spatial contrasts that are far more difficult to characterize for UFPs than for $PM_{2.5}$.

CONCLUSIONS

Airborne PM has been the focus of extensive research and debate in the United States and around the world. At this point, considerable evidence from a broad array of experimental and epidemiologic studies has led to strong scientific consensus on the independent associations of airborne PM, in particular $PM_{2.5}$ and PM_{10} , with adverse respiratory and cardiovascular effects on human health (U.S. EPA 2009; Brook et al. 2010; CASAC 2010). This evidence has provided the foundation for many regulatory decisions to limit both PM emissions, particularly from motor vehicles, and ambient PM concentrations to which people might be exposed.

What role have ambient concentrations of UFPs played in the adverse effects that have been observed in human populations exposed to ambient air pollution?

In the years since investigators first became concerned about the potential adverse effects of exposure to the smallest of airborne particles, a considerable body of research has been conducted on the emissions, exposures, and health effects of UFPs. Several factors — the unique physical properties of UFPs, their interactions with tissues and cells, their potential for translocation beyond the lung — have led scientists to expect that UFPs may have specific or enhanced toxicity relative to other particle size fractions and may contribute to effects beyond the respiratory system. However, toxicologic studies in animals, human exposure studies, and epidemiologic studies to date have not provided consistent findings of such effects with exposures to ambient levels of UFPs, particularly in human populations. The evidence also does not support a conclusion that exposures to UFPs alone can account in substantial ways for the adverse effects that have also been associated with other ambient pollutants such as $PM_{2.5}$.

That the current database of experimental and epidemiologic studies does not support strong and consistent conclusions about the independent effects of UFPs on human health does not mean that such effects can be ruled out. The limitations in the evidence base are attributable to underlying gaps in exposure data, to numerous challenges to comparison and synthesis of existing studies, and to the inherent complexity of the scientific task scientists have set out to accomplish. Similar kinds of issues face ongoing efforts to tease out the health significance of other components of the PM mixture (Brunekreef 2010; Bell 2012; Lippmann et al. in press; Vedal et al. in press). Fortunately, and irrespective of evidence for a specific role for UFPs, recent PM regulatory decisions affecting fuels, engine designs and exhaust aftertreatment in the United States, Europe, and Japan will result in the significant reduction in emissions of both fine and ultrafine particles.

Where Do We Go From Here?

There are many considerations beyond the scientific opinions expressed in this issue of HEI Perspectives that inform the level of confidence in the evidence necessary for policy makers to "ensure that resources spent in the future on control technology and regulatory compliance will have a reasonable probability of success" (U.S. National Research Council 1998). Among them is the need to weigh carefully the value to scientific understanding and to regulatory decisions of continuing to treat UFPs as an individual pollutant versus alternative approaches that focus on the health effects of exposure to traffic or to the broader air pollution mixture.

As part of this discussion, however, steps to address some of the limitations of the current evidence on UFPs should be considered.

Experimental Studies

Even in the absence of broad-scale epidemiologic evidence, insight into the potential toxicologic implications of differences in the deposition and retention of inhaled UFPs may still be possible with well-designed experimental studies of controlled exposures to UFPs and related copollutants. Examples include:

- Further animal studies of the potential for, and health effects of, translocation and accumulation of UFPs in tissues beyond the lung, including the central nervous system. This work should be extended to additional animal species and to models of human disease.
- Animal inhalation studies of long-term exposures to UFPs. Virtually all of the work to date has been with short-term exposures; the kind of evidence that has been so important for understanding the effects of long-term exposures to PM_{2.5} and PM₁₀ does not exist for UFPs.
- Further human studies of UFP health effects and mechanisms. Such studies should include both controlled laboratory exposures and real-world panel studies that target UFPs of various sources and chemical composition but also involve comparisons with various PM size fractions and copollutants.

Epidemiologic Studies

- Studies of long-term exposure to ambient UFPs. The kinds of data that have provided broad support for epidemiologic investigations of the public health implications of long-term exposure to $PM_{2.5}$ and PM_{10} multiple years of monitoring data, using consistent methods, in major urban areas representing millions of people have simply not existed for UFPs. Different approaches to characterizing exposure (discussed below in recommendations for characterization of ambient UFP exposures) need to be considered for long-term studies of UFP exposures to be possible.
- Targeted study designs with sufficient contrasts in UFP exposure, but that improve the ability to characterize the independent effects of exposure to UFPs. These might include scripted activities, measurements in environments with unique UFP exposure features, and studies of interventions that are specifically designed to control exposures to UFPs. Intervention study designs where investigators filter out exposures to UFPs or to particles of various size fractions, but not gases, may be informative here.
- More consistent and comparable study designs. One of the factors that has limited comparisons and interpretation of the epidemiologic studies conducted to date on the effects of short-term exposures to ambient UFPs is the variability in study designs, both in exposure methods and measurements (including copollutants) and in the health outcomes across individual studies and cities. The kinds of meta-analyses that have been used

successfully to strengthen inferences from short-term studies of $PM_{2.5}$ and PM_{10} in the United States, Europe, Asia, and Latin America are consequently challenging and, to our knowledge, have not been conducted. Even the more consistent multicity study designs desirable for meta-analyses, and that are beginning to be applied to the study of UFPs, may still yield equivocal results if they must rely primarily on central monitors or do not account appropriately for copollutants.

Better Characterization of Ambient UFP Exposures

Many of the underlying challenges posed by the existing evidence on ambient UFPs relate to differences in how they are measured across studies and how much data are available to assess exposures. In part, variation in approaches reflects the exploratory stage of efforts to identify size and other characteristics that might predict toxicity of UFPs. However, it is one of the factors limiting comparison and synthesis of the studies that have been done to date.

- Find ways to exploit the high spatial variability of ambient UFPs in health studies. As UFPs do show gradients in concentrations within urban areas that are related to traffic sources, spatial modeling approaches that have been used to assess health effects related to traffic pollution may be applicable to UFPs, although characterizing the role of UFPs within the traffic mixture will remain challenging.
- Explore UFP metrics other than NC for these applications. For example, UFP mass and chemical composition are more difficult to measure at an individual location, but population exposure to these metrics may be easier to predict using a combination of statistical models and reactive chemical-transport models. Linked with source-apportionment methods, such data could assist in identifying sources and allow monitoring of how changes to those sources would affect these temporal and spatial patterns in the future.
- Consider the growing literature on new statistical and other analytic methods aimed at disentangling the sources, exposures, and health implications of PM components and other copollutants. Although not discussed in this document, these methods are addressing many of the same basic issues as those faced in the study of UFPs.

Ultimately, it will be important to monitor the effect on ambient UFPs of actions taken that target the emissions of PM and other pollutants that may directly or indirectly affect ambient UFP concentrations. For motor vehicle sources, the primary focus of this document, the recent developments in motor vehicle technologies — exhaust aftertreatment systems and the development of new fuels — already suggest major changes in the absolute and relative importance of UFP and other pollutant emissions. The time course and impact of regulatory and technological changes on ambient concentrations and on human exposures will depend on a number of factors, including shifts in the size, age, and composition of the vehicle fleet in particular regions. However, without ongoing monitoring and evaluation in the years to come, questions about whether or not these changes have addressed the most important characteristics of the PM mixture will remain.

REFERENCES

Abernethy R. 2012. A land use regression model for ultrafine particles in Vancouver. Master's thesis. Occupational and Environmental Hygiene, University of British Columbia, Canada. Available from: *https://circle* .ubc.ca/handle/2429/41895.

Abt E, Suh HH, Allen G, Koutrakis P. 2000. Characterization of indoor particle sources: A study conducted in the metropolitan Boston area. Environ Health Perspect 108:35–44.

Advanced Collaborative Emissions Study (ACES). 2012. Advanced Collaborative Emissions Study (ACES) Subchronic Exposure Results: Biologic Responses in Rats and Mice and Assessment of Genotoxicity. Research Report 166. Health Effects Institute, Boston, MA.

Alessandrini F, Beck-Speier I, Krappmann D, Weichenmeier I, Takenaka S, Karg E, Kloo B, Schulz H, Jakob T, Mempel M, Behrendt H. 2009. Role of oxidative stress in ultrafine particle-induced exacerbation of allergic lung inflammation. Am J Respir Crit Care Med 179:984–991.

Alessandrini F, Schulz H, Takenaka S, Lentner B, Karg E, Behrendt H, Jakob T. 2006. Effects of ultrafine carbon particle inhalation on allergic inflammation of the lung. J Allergy Clin Immunol 117:824–830.

Alessandrini F, Semmler-Behnke M, Jakob T, Schulz H, Behrendt H, Kreyling W. 2008. Total and regional deposition of ultrafine particles in a mouse model of allergic inflammation of the lung. Inhal Toxicol 20:585–593.

American Thoracic Society. 2000. What constitutes an adverse health effect of air pollution? Am J Respir Crit Care Med 161:665–673.

Andersen ZJ, Loft S, Ketzel M, Stage M, Scheike T, Mette MN, Bisgaard H. 2008a. Ambient air pollution triggers wheezing symptoms in infants. Thorax 63:710–716. Andersen ZJ, Olsen TS, Andersen KK, Loft S, Ketzel M, Raaschou-Nielsen O. 2010. Association between shortterm exposure to ultrafine particles and hospital admissions for stroke in Copenhagen, Denmark. Eur Heart J 31:2034–2040.

Andersen ZJ, Wahlin P, Raaschou-Nielsen O, Ketzel M, Scheike T, Loft S. 2008b. Size distribution and total number concentration of ultrafine and accumulation mode particles and hospital admissions in children and the elderly in Copenhagen, Denmark. Occup Environ Med 65:458–466.

Anderson HR, Armstrong B, Hajat S, Harrison R, Monk V, Poloniecki J, Timmis A, Wilkinson P. 2010. Air pollution and activation of implantable cardioverter defibrillators in London. Epidemiology 21:405–413.

Anderson HR, Spix C, Medina S, Schouten JP, Castellsague J, Rossi G, Zmirou D, Touloumi G, Wojtyniak B, Ponka A, Bacharova L, Schwartz J, Katsouyanni K. 1997. Air pollution and daily admissions for chronic obstructive pulmonary disease in 6 European cities: Results from the APHEA project. Eur Respir J 10:1064–1071.

Andre E, Stoeger T, Takenaka S, Bahnweg M, Ritter B, Karg E, Lentner B, Reinhard C, Schulz H, Wjst M. 2006. Inhalation of ultrafine carbon particles triggers biphasic proinflammatory response in the mouse lung. Eur Respir J 28:275–285.

Apte JS, Kirchstetter TW, Reich AH, Deshpande SJ, Kaushik G, Chel A, Marshall JD, Nazaroff WW. 2011. Concentrations of fine, ultrafine, and black carbon particles in auto-rickshaws in New Delhi, India. Atmos Environ 45:4470–4480.

Araujo JA, Barajas B, Kleinman M, Wang X, Bennett BJ, Gong KW, Navab M, Harkema J, Sioutas C, Lusis AJ, Nel AE. 2008. Ambient particulate pollutants in the ultrafine range promote early atherosclerosis and systemic oxidative stress. Circ Res 102:589–596.

Araujo JA, Nel AE. 2009. Particulate matter and atherosclerosis: Role of particle size, composition and oxidative stress. Part Fibre Toxicol 6:24.

Atkinson RW, Fuller GW, Anderson HR, Harrison RM, Armstrong B. 2010. Urban ambient particle metrics and health: A time-series analysis. Epidemiology 21:501–511.

Auerbach O, Conston AS, Garfinkel L, Parks VR, Kaslow HD, Hammond EC. 1980. Presence of asbestos bodies in organs other than the lung. Chest 77:133–137.

Barclay JL, Miller BG, Dick S, Dennekamp M, Ford I, Hillis GS, Ayres JG, Seaton A. 2009. A panel study of air pollution

in subjects with heart failure: Negative results in treated patients. Occup Environ Med 66:325–334.

Beckett WS, Chalupa DF, Pauly-Brown A, Speers DM, Stewart JC, Frampton MW, Utell MJ, Huang LS, Cox C, Zareba W, Oberdörster G. 2005. Comparing inhaled ultrafine versus fine zinc oxide particles in healthy adults: A human inhalation study. Am J Respir Crit Care Med 171:1129–1135.

Begum BA, Biswas SK, Hopke PK. 2006. Impact of banning of two-stroke engines on airborne particulate matter concentrations in Dhaka, Bangladesh. J Air Waste Manage Assoc 56:85–89.

Bein KJ, Zhao Y, Wexler AS, Johnston MV. 2005. Speciation of size-resolved individual ultrafine particles in Pittsburgh, Pennsylvania. J Geophys Res 110:D07S05.

Bell ML. 2012. Assessment of the Health Impacts of Particulate Matter Characteristics. Research Report 161. Health Effects Institute, Boston, MA.

Belleudi V, Faustini A, Stafoggia M, Cattani G, Marconi A, Perucci CA, Forastiere F. 2010. Impact of fine and ultrafine particles on emergency hospital admissions for cardiac and respiratory diseases. Epidemiology 21:414–423.

Berger A, Zareba W, Schneider A, Rückerl R, Ibald-Mulli A, Cyrys J, Wichmann HE, Peters A. 2006. Runs of ventricular and supraventricular tachycardia triggered by air pollution in patients with coronary heart disease. JOM 48:1149– 1158.

Bergmann M, Kirchner U, Vogt R, Benter T. 2009. On-road and laboratory investigation of low-level PM emissions of a modern diesel particulate filter equipped diesel passenger car. Atmos Environ 43:1908–1916.

Berry JP, Arnoux B, Stanislas G, Galle P, Chretien J. 1977. A microanalytic study of particles transport across the alveoli: Role of blood platelets. Biomedicine 27:354–357.

Biswas S, Verma V, Schauer JJ, Sioutas C. 2009. Chemical speciation of PM emissions from heavy-duty diesel vehicles equipped with diesel particulate filter (DPF) and selective catalytic reduction (SCR) retrofits. Atmos Environ 43:1917–1925.

Braniš M, Vyskovska J, Maly M, Hovorka J. 2010. Association of size-resolved number concentrations of particulate matter with cardiovascular and respiratory hospital admissions and mortality in Prague, Czech Republic. Inhal Toxicol 22:21–28.

Brauner EV, Forchhammer L, Møller P, Barregard L, Gunnarsen L, Afshari A, Wåhlin P, Glasius M, Dragsted LO, Basu S, Raaschou-Nielsen O, Loft S. 2008. Indoor particles affect vascular function in the aged: An air filtration-based intervention study. Am J Respir Crit Care Med 177:419–425.

Breitner S, Liu L, Cyrys J, Brüske I, Franck U, Schlink U, Leitte AM, Herbarth O, Wiedensohler A, Wehner B, Hu M, Pan X-C, Wichmann HE, Peters A. 2011. Sub-micrometer particulate air pollution and cardiovascular mortality in Beijing, China. Sci Total Environ 409:5196–5204.

Breitner S, Stölzel M, Cyrys J, Pitz M, Wölke G, Kreyling W, Küchenhoff H, Heinrich J, Wichmann H, Peters A. 2009. Short-term mortality rates during a decade of improved air quality in Erfurt, Germany. Environ Health Perspect 117:448–454.

Brook RD, Rajagopalan S, Pope CA 3rd, Brook JR, Bhatnagar A, Diez-Roux AV, Holguin F, Hong Y, Luepker RV, Mittleman MA, Peters A, Siscovick D, Smith SC Jr, Whitsel L, Kaufman JD. 2010. Particulate matter air pollution and cardiovascular disease: An update to the scientific statement from the American Heart Association. Circulation 121:2331–2378.

Brown JS, Zeman KL, Bennett WD. 2002. Ultrafine particle deposition and clearance in the healthy and obstructed lung. Am J Respir Crit Care Med 166:1240–1247.

Brunekreef B. 2010. The color of smoke. Epidemiology 21:903–904.

Brüske I, Hampel R, Baumgärtner Z, Rückerl R, Greven S, Koenig W, Peters A, Schneider A. 2011. Ambient air pollution and lipoprotein-associated phospholipase A_2 in survivors of myocardial infarction. Environ Health Perspect 119:921–926.

Brüske I, Hampel R, Socher MM, Rückerl R, Schneider A, Heinrich J, Oberdörster G, Wichmann HE, Peters A. 2010. Impact of ambient air pollution on the differential white blood cell count in patients with chronic pulmonary disease. Inhal Toxicol 22:245–252.

Burtscher H, Loretz S, Keller A, Mayer A, Kasper M, Artley RJ, Strasser R, Czerwinski J. 2008. Nanoparticle Filtration for Vehicle Cabins. SAE Technical Paper 2008-01-0827. SAE International.

Cahill TA, Barnes DE, Spada NJ, Lawton JA, Cahill TM. 2011. Very fine and ultrafine metals and ischemic heart disease in the California Central Valley 1: 2003–2007. Aerosol Sci Technol 45:1123–1134.

Campbell A, Araujo JA, Li H, Sioutas C, Kleinman M. 2009. Particulate matter induced enhancement of inflammatory markers in the brains of apolipoprotein E knockout mice. J Nanosci Nanotechnol 9:5099–5104.

Campbell A, Oldham M, Becaria A, Bondy SC, Meacher D, Sioutas C, Misra C, Mendez LB, Kleinman M. 2005. Particulate matter in polluted air may increase biomarkers of inflammation in mouse brain. Neurotoxicology 26:133– 140.

Cass GR, Hughes LA, Bhave P, Kleeman MJ, Allen JO, Salmon LG. 2000. The chemical composition of atmospheric ultrafine particles. Philos Trans R Soc Lond Ser A-Math Phys Eng Sci 358:2581–2592.

Cedervall T, Lynch I, Lindman S, Berggård T, Thulin E, Nilsson H, Dawson KA, Linse S. 2007. Understanding the nanoparticle-protein corona using methods to quantify exchange rates and affinities of proteins for nanoparticles. Proc Natl Acad Sci USA 104:2050–2055.

Chalupa D, Gibb FR, Morrow PE, Oberdörster G, Riesenfeld E, Gelein R, Utell MJ, Frampton MW. 2002. A facility for controlled human exposures to ultrafines particles (Heinrich U and Mohr U, eds). ILSI Press, Washington, D.C.

Chalupa DC, Morrow PE, Oberdörster G, Utell MJ, Frampton MW. 2004. Ultrafine particle deposition in subjects with asthma. Environ Health Perspect 112:879–882.

Chan CC, Chuang KJ, Shiao GM, Lin LY. 2004. Personal exposure to submicrometer particles and heart rate variability in human subjects. Environ Health Perspect 112:1063–1067.

Chang MC, Sioutas C, Kim S, Gong H Jr, Linn WS. 2000. Reduction of nitrate losses from filter and impactor samplers by means of concentration enrichment. Atmos Environ 34:85–98.

Charron A, Harrison RM. 2003. Primary particle formation from vehicle emissions during exhaust dilution in the roadside atmosphere. Atmos Environ 37:4109–4119.

Chen J, Tan M, Nemmar A, Song W, Dong M, Zhang G, Li Y. 2006. Quantification of extrapulmonary translocation of intratracheal-instilled particles in vivo in rats: effect of lipopolysaccharide. Toxicology 222:195–201.

Cheung HC, Morawska L, Ristovski ZD, Wainwright D. 2012. Influence of medium range transport of particles from nucleation burst on particle number concentration within the urban airshed. Atmos Chem Phys 12:4951–4962.

Cheung KL, Ntziachristos L, Tzamkiozis T, Schauer JJ, Samaras Z, Moore KF, Sioutas C. 2010. Emissions of particulate trace elements, metals and organic species from gasoline, diesel, and biodiesel passenger vehicles and their relation to oxidative potential. Aerosol Sci Technol 44:500–513.

Chuang KJ, Chan CC, Shiao GM, Su TC. 2005. Associations between submicrometer particles exposures and blood pressure and heart rate in patients with lung function impairments. JOM 47:1093–1098.

Chuepeng S, Xu HM, Tsolakis A, Wyszynski M, Price P. 2011. Particulate matter size distribution in the exhaust gas of a modern diesel engine fuelled with a biodiesel blend. Biomass Bioenerg 35:4280–4289.

Chung A, Herner JD, Kleeman MJ. 2001. Detection of alkaline ultrafine atmospheric particles at Bakersfield, California. Environ Sci Technol 35:2184–2190.

Clean Air Scientific Advisory Committee (CASAC). 2010. Review of Policy Assessment for the Review of the PM NAAQS: Second External Review Draft. EPA-CASAC-10-015. Research Triangle Park, NC. Available from www.yosemite.epa.gov/sab/sabproduct.nsf/264cb122 7d55e02c85257402007446a4/8bee96ad3228eabe8525760 400702786!OpenDocument&TableRow=2.3#2.

Cohen BS, Heikkinen MSA, Hazi Y, Gao H, Peters P, Lippman M. 2004. Field Evaluation of Nanofilm Detectors for Measuring Acidic Particles in Indoor and Outdoor Air. Research Report 121. Health Effects Institute, Boston, MA.

Coordinating Research Council. 2007. Heavy-Duty Vehicle Chassis Dynamometer Testing for Emissions Inventory, Air Quality Modeling, Source Apportionment and Air Toxics Emissions Inventory. E-55/E-59. CRC. Alpharetta, GA. Available from www.crcao.org/reports/recentstudies2007/ E-55-59/E-55_59_Final_Report_23AUG2007.pdf. Posted August 2007. Accessed 10/25/2012.

Coordinating Research Council. 2009. Phase 1 of the Advanced Collaborative Emissions Study. SWRI-03.13062. CRC. Alpharetta, GA. Available from www.crcao.org/ reports/recentstudies2009/ACES% 20Phase%201/ACES% 20Phase1%20Final%20Report%2015JUN2009.pdf. Posted June 2009. Accessed 10/24/2012.

Cyrys J, Pitz M, Heinrich J, Wichmann H, Peters A. 2008. Spatial and temporal variation of particle number concentration in Augsburg, Germany. Sci Total Environ 401:168–175.

Dahl A, Gharibi A, Swietlicki E, Gudmundsson A, Bohgard M, Ljungman A, Blomqvist G, Gustafsson M. 2006. Trafficgenerated emissions of ultrafine particles from pavementtire interface. Atmos Environ 40:1314–1323. Daigle CC, Chalupa DC, Gibb FR, Morrow PE, Oberdörster G, Utell MJ, Frampton MW. 2003. Ultrafine particle deposition in humans during rest and exercise. Inhal Toxicol 15:539–552.

Dales R, Liu L, Szyszkowicz M, Dalipaj M, Willey J, Kulka R, Ruddy TD. 2007. Particulate air pollution and vascular reactivity: The bus stop study. Int Arch Occup Environ Health 81:159–164.

De Filippo A, Maricq MM. 2008. Diesel nucleation mode particles: Semivolatile or solid? Environ Sci Technol 42:7957–7962.

de Hartog JJ, Ayres JG, Karakatsani A, Analitis A, Brink HT, Hameri K, Harrison R, Katsouyanni K, Kotronarou A, Kavouras I, Meddings C, Pekkanen J, Hoek G. 2010. Lung function and indicators of exposure to indoor and outdoor particulate matter among asthma and COPD patients. Occup Environ Med 67:2–10.

de Hartog JJ, Hoek G, Peters A, Timonen KL, Ibald-Mulli A, Brunekreef B, Heinrich J, Tiittanen P, Van Wijnen JH, Kreyling W, Kulmala M, Pekkanen J. 2003. Effects of fine and ultrafine particles on cardiorespiratory symptoms in elderly subjects with coronary heart disease: The ULTRA study. Am J Epidemiol 157:613–623.

Delfino RJ, Gillen DL, Tjoa T, Staimer N, Polidori A, Arhami M, Sioutas C, Longhurst J. 2011. Electrocardiographic ST-segment depression and exposure to trafficrelated aerosols in elderly subjects with coronary artery disease. Environ Health Perspect 119:196–202.

Delfino RJ, Staimer N, Tjoa T, Arhami M, Polidori A, Gillen DL, George SC, Shafer MM, Schauer JJ, Sioutas C. 2010a. Associations of primary and secondary organic aerosols with airway and systemic inflammation in an elderly panel cohort. Epidemiology 21:892–902.

Delfino RJ, Staimer N, Tjoa T, Arhami M, Polidori A, Gillen DL, Kleinman MT, Schauer JJ, Sioutas C. 2010b. Association of biomarkers of systemic inflammation with organic components and source tracers in quasi-ultrafine particles. Environ Health Perspect 118:756–762.

Delfino RJ, Staimer N, Tjoa T, Gillen DL, Polidori A, Arhami M, Kleinman MT, Vaziri ND, Longhurst J, Sioutas C. 2009. Air pollution exposures and circulating biomarkers of effect in a susceptible population: Clues to potential causal component mixtures and mechanisms. Environ Health Perspect 117:1232–1238.

Delfino RJ, Staimer N, Tjoa T, Polidori A, Arhami M, Gillen DL, Kleinman MT, Vaziri ND, Longhurst J, Zaldivar F, Sioutas C. 2008. Circulating biomarkers of inflammation,

antioxidant activity, and platelet activation are associated with primary combustion aerosols in subjects with coronary artery disease. Environ Health Perspect 116:898–906.

Delfino RJ, Tjoa T, Gillen DL, Staimer N, Polidori A, Arhami M, Jamner L, Sioutas C, Longhurst J. 2010c. Traffic-related air pollution and blood pressure in elderly subjects with coronary artery disease. Epidemiology 21:396–404.

Denier van der Gon HAC, Visschedijk A, Johansson C, Ntziachristos L, Harrison R. 2010. Size-resolved pan-European anthropogenic particle number inventory. Abstract submitted to International Aerosol Conference, August 29–September 3, 2010, Helsinki, Finland.

Diapouli E, Chaloulakou A, Mihalopoulos N, Spyrellis N. 2008. Indoor and outdoor PM mass and number concentrations at schools in the Athens area. Environ Monit Assess 136:13–20.

Dockery DW, Luttmann-Gibson H, Rich DQ, Link MS, Mittleman MA, Gold DR, Koutrakis P, Schwartz JD, Verrier RL. 2005a. Association of air pollution with increased incidence of ventricular tachyarrhythmias recorded by implanted cardioverter defibrillators. Environ Health Perspect 113:670–674.

Dockery DW, Luttman-Gibson H, Rich DQ, Link MS, Schwartz JD, Gold DR, Koutrakis P, Verrier RL, Mittleman MA. 2005b. Part II. Association of air pollution with confirmed arrhythmias recorded by implanted defibrillators. In: Particulate Air Pollution and Nonfatal Cardiac Events. Research Report 124. Health Effects Institute, Boston, MA.

Dockery DW, Pope CA, Xu X, Spengler JD, Ware JH, Fay ME, Ferris BG Jr, Speizer FE. 1993. An association between air pollution and mortality in six US cities. N Engl JMed 329:1753–1759.

Donaldson K, Tran L, Jimenez LA, Duffin R, Newby DE, Mills N, MacNee W, Stone V. 2005. Combustion-derived nanoparticles: a review of their toxicology following inhalation exposure. Part Fibre Toxicol 2:10.

Ebelt ST, Petkau AJ, Vedal S, Fisher TV, Brauer M. 2000. Exposure of chronic obstructive pulmonary disease patients to particulate matter: Relationships between personal and ambient air concentrations. J Air Waste Manag Assoc 50:1081–1094.

Elder A, Couderc JP, Gelein R, Eberly S, Cox C, Xia X, Zareba W, Hopke P, Watts W, Kittelson D, Frampton M, Utell M, Oberdörster G. 2007. Effects of on-road highway aerosol exposures on autonomic responses in aged, spontaneously hypertensive rats. Inhal Toxicol 19:1–12. Elder A, Gelein R, Finkelstein J, Phipps R, Frampton M, Utell M, Kittelson DB, Watts WF, Hopke P, Jeong CH, Kim E, Liu W, Zhao W, Zhuo L, Vincent R, Kumarathasan P, Oberdörster G. 2004a. On-road exposure to highway aerosols. 2. Exposures of aged, compromised rats. Inhal Toxicol 16 Suppl 1:41–53.

Elder A, Oberdörster G. 2006. Translocation and effects of ultrafine particles outside of the lung. Clin Occup Environ Med 5:785–796.

Elder AC, Gelein R, Azadniv M, Frampton M, Finkelstein J, Oberdörster G. 2004b. Systemic effects of inhaled ultrafine particles in two compromised, aged rat strains. Inhal Toxicol 16:461–471.

Elder AC, Gelein R, Finkelstein JN, Cox C, Oberdörster G. 2000a. Pulmonary inflammatory response to inhaled ultrafine particles is modified by age, ozone exposure, and bacterial toxin. Inhal Toxicol 12 Suppl 4:227–246.

Elder ACP, Finkelstein J, Johnston C, Gelein R, Oberdörster G. 2000b. Induction of adaptation to inhaled lipopolysaccharide in young and old rats and mice. Inhal Toxicol 12:225–243.

Elder ACP, Gelein R, Finkelstein JN, Cox C, Oberdörster G. 2000c. Endotoxin priming affects the lung response to ultrafine particles and ozone in young and old rats. Inhal Toxicol 12:85–98.

Elleman RA, Covert DS. 2009. Aerosol size distribution modeling with the Community Multiscale Air Quality modeling system in the Pacific Northwest: 1. Model comparison to observations. J Geophys Res 114:D11206.

Elleman RA, Covert DS. 2010. Aerosol size distribution modeling with the Community Multiscale Air Quality modeling system in the Pacific Northwest: 3. Size distribution of particles emitted into a mesoscale model. J Geophys Res-Atmos 115:D03204.

Fanning EW, Froines JR, Utell MJ, Lippmann M, Oberdörster G, Frampton M, Godleski J, Larson TV. 2009. Particulate matter (PM) research centers (1999–2005) and the role of interdisciplinary center-based research. Environ Health Perspect 117:167–174.

Ferin J, Oberdörster G, Penney DP. 1992. Pulmonary retention of ultrafine and fine particles in rats. Am J Respir Cell Mol Biol 6:535–542.

Folino AF, Scapellato ML, Canova C, Maestrelli P, Bertorelli G, Simonato L, Iliceto S, Lotti M. 2009. Individual exposure to particulate matter and the short-term arrhythmic and autonomic profiles in patients with myocardial infarction. Eur Heart J 30:1614–1620. Fontaras G, Kousoulidou M, Karavalakis G, Tzamkiozis T, Pistikopoulos P, Ntziachristos L, Bakeas E, Stournas S, Samaras Z. 2010. Effects of low concentration biodiesel blend application on modern passenger cars. Part 1: Feedstock impact on regulated pollutants, fuel consumption and particle emissions. Environ Pollut 158:1451–1460.

Forastiere F, Stafoggia M, Picciotto S, Bellander T, D'Ippoliti D, Lanki T, Von Klot S, Nyberg F, Paatero P, Peters A, Pekkanen J, Sunyer J, Perucci CA. 2005. A casecrossover analysis of out-of-hospital coronary deaths and air pollution in Rome, Italy. Am J Respir Crit Care Med 172:1549–1555.

Frampton MW. 2001. Systemic and cardiovascular effects of airway injury and inflammation: Ultrafine particle exposure in humans. Environ Health Persp 109:529–532.

Frampton MW, Stewart JC, Oberdörster G, Morrow PE, Chalupa D, Pietropaoli AP, Frasier LM, Speers DM, Cox C, Huang LS, Utell MJ. 2006. Inhalation of ultrafine particles alters blood leukocyte expression of adhesion molecules in humans. Environ Health Perspect 114: 51–58.

Frampton MW, Utell MJ, Samet JM. 2000. Cardiopulmonary consequences of particle inhalation. In: Particle-Lung Interactions (Gehr P, Heyder J eds) pp 653–670. Marcel Dekker, New York, NY.

Frampton MW, Utell MJ, Zareba W, Oberdörster G, Cox C, Huang LS, Morrow PE, Lee FE-H, Chalupa D, Fraiser LM, Speers DM, Stewart J. 2004. Effects of Exposure to Ultrafine Carbon Particles in Healthy Subjects and Subjects with Asthma. Research Report 126. Health Effects Institute, Boston, MA.

Franck U, Odeh S, Wiedensohler A, Wehner B, Herbarth O. 2011. The effect of particle size on cardiovascular disorders: The smaller the worse. Sci Total Environ 409:4217–4221.

Fruin S, Westerdahl D, Sax T, Sioutas C, Fine PM. 2008. Measurements and predictors of on-road ultrafine particle concentrations and associated pollutants in Los Angeles. Atmos Environ 42:207–219.

Fujimaki H, Saneyoshi K, Shiraishi F, Imai T, Endo T. 1997. Inhalation of diesel exhaust enhances antigen-specific IgE antibody production in mice. Toxicology 116:227–233.

Gäggeler HW, Baltensperger U, Emmenegger M, Jost DT, Schmidtott A, Haller P, Hofmann M. 1989. The epiphaniometer, a new device for continuous aerosol monitoring. J Aerosol Sci 20:557–564. Geiser M, Casaulta M, Kupferschmid B, Schulz H, Semmler-Behnke M, Kreyling W. 2008. The role of macrophages in the clearance of inhaled ultrafine titanium dioxide particles. Am J Respir Cell Mol Biol 38:371–376.

Geiser M, Kreyling WG. 2010. Deposition and biokinetics of inhaled nanoparticles. Part Fibre Toxicol 7:2.

Geiser M, Rothen-Rutishauser B, Kapp N, Schurch S, Kreyling W, Schulz H, Semmler M, Im Hof V, Heyder J, Gehr P. 2005. Ultrafine particles cross cellular membranes by nonphagocytic mechanisms in lungs and in cultured cells. Environ Health Perspect 113:1555–1560.

Geller MD, Ntziachristos L, Mamakos A, Samaras Z, Schmitz DA, Froines JR, Sioutas C. 2006. Physicochemical and redox characteristics of particulate matter (PM) emitted from gasoline and diesel passenger cars. Atmos Environ 40:6988–7004.

Giechaskiel B, Alfoldy B, Drossinos Y. 2009. A metric for health effects studies of diesel exhaust particles. J Aerosol Sci 40:639–651.

Giechaskiel B, Ntziachristos L, Samaras Z, Scheer V, Casati R, Vogt R. 2005. Formation potential of vehicle exhaust nucleation mode particles on-road and in the laboratory. Atmos Environ 39:3191–3198.

Gilmour PS, Ziesenis A, Morrison ER, Vickers MA, Drost EM, Ford I, Karg E, Mossa C, Schroeppel A, Ferron GA, Heyder J, Greaves M, MacNee W, Donaldson K. 2004. Pulmonary and systemic effects of short-term inhalation exposure to ultrafine carbon black particles. Toxicol Appl Pharmacol 195:35–44.

Gojova A, Guo B, Kota RS, Rutledge JC, Kennedy IM, Barakat AI. 2007. Induction of inflammation in vascular endothelial cells by metal oxide nanoparticles: Effect of particle composition. Environ Health Persp 115:403–409.

Gojova A, Lee JT, Jung HS, Guo B, Barakat AI, Kennedy IM. 2009. Effect of cerium oxide nanoparticles on inflammation in vascular endothelial cells. Inhal Toxicol 21 Suppl 1:123–130.

Gong H, Sioutas C, Linn WS, Clark KW, Terrell SL, Terrell LL, Anderson KR, Kim S, Chang M-C. 2000. Controlled human exposures to concentrated ambient fine particles in metropolitan Los Angeles: Methodology and Preliminary Health-Effect Findings. Inhal Toxicol 12:107–119.

Gong H Jr, Linn WS, Clark KW, Anderson KR, Sioutas C, Alexis NE, Cascio WE, Devlin RB. 2008. Exposures of healthy and asthmatic volunteers to concentrated ambient ultrafine particles in Los Angeles. Inhal Toxicol 20:533– 545. Gong H Jr, Linn WS, Terrell SL, Anderson KR, Clark KW, Terrell LL, Sioutas C. 2002. Controlled exposures of asthmatic and healthy volunteers to concentrated ambient fine particles in Los Angeles (abstract). Am J Respir Crit Care Med 165:A214.

Gong H Jr, Sioutas C, Linn WS. 2003. Controlled Exposures of Healthy and Asthmatic Volunteers to Concentrated Ambient Particles in Metropolitan Los Angeles. Research Report 118. Health Effects Institute, Boston, MA.

Gong L, Xu B, Zhu Y. 2009. Ultrafine particles deposition inside passenger vehicles. Aerosol Sci Technol 43:544– 553.

Guo H, Morawska L, He C, Gilbert D. 2008. Impact of ventilation scenario on air exchange rates and on indoor particle number concentrations in an air-conditioned classroom. Atmos Environ 42:757–768.

Guo H, Morawska L, He C, Zhang Y, Ayoko G, Cao M. 2010. Characterization of particle number concentrations and $PM_{2.5}$ in a school: Influence of outdoor air pollution on indoor air. Environ Sci Pollut Res Int 17:1268–1278.

Hahn FF, Barr EB, Ménache MG, Segrave J. 2005. Particle Size and Composition Related to Adverse Health Effects in Aged, Sensitive Rats. Research Report 129. Health Effects Institute, Boston, MA.

Halonen JI, Lanki T, Yli-Tuomi T, Kulmala M, Tiittanen P, Pekkanen J. 2008. Urban air pollution, and asthma and COPD hospital emergency room visits. Thorax 63:635– 641.

Halonen JI, Lanki T, Yli-Tuomi T, Tiittanen P, Kulmala M, Pekkanen J. 2009. Particulate air pollution and acute cardiorespiratory hospital admissions and mortality among the elderly. Epidemiology 20:143–153.

Ham WA, Kleeman MJ. 2011. Size-resolved source apportionment of carbonaceous particulate matter in urban and rural sites in central California. Atmos Environ 45:3988– 3995.

Hampel R, Schneider A, Brüske I, Zareba W, Cyrys J, Rückerl R, Breitner S, Korb H, Sunyer J, Wichmann HE, Peters A. 2010. Altered cardiac repolarization in association with air pollution and air temperature among myocardial infarction survivors. Environ Health Perspect 118:1755–1761.

Harder V, Gilmour P, Lentner B, Karg E, Takenaka S, Ziesenis A, Stampfl A, Kodavanti U, Heyder J, Schulz H. 2005. Cardiovascular responses in unrestrained WKY rats to inhaled ultrafine carbon particles. Inhal Toxicol 17:29–42. Harris SJ, Maricq MM. 2001. Signature size distributions for diesel and gasoline engine exhaust particulate matter. J Aerosol Sci 32:749–764.

Hazi Y, Heikkinen MSA, Cohen BS. 2003. Size distribution of acidic sulfate ions in fine ambient particulate matter and assessment of source region effect. Atmos Environ 37:5403–5413.

Health Effects Institute. 2003. Revised Analyses of Time-Series Studies of Air Pollution and Health. Special Report. Health Effects Institute, Boston, MA.

Health Effects Institute. 2010. Traffic-Related Air Pollution: A Critical Review of the Literature on Emissions, Exposure, and Health Effects. Special Report 17. Health Effects Institute, Boston, MA.

Health Effects Institute. 2011. The Future of Vehicle Fuels and Technologies: Anticipating Health Benefits and Challenges. Communication 16. Health Effects Institute, Boston, MA.

Heikkila J, Virtanen A, Ronkko T, Keskinen J, Aakko-Saksa P, Murtonen T. 2009. Nanoparticle emissions from a heavyduty engine running on alternative diesel fuels. Environ Sci Technol 43:9501–9506.

Henneberger A, Zareba W, Ibald-Mulli A, Rückerl R, Cyrys J, Couderc JP, Mykins B, Woelke G, Wichmann HE, Peters A. 2005. Repolarization changes induced by air pollution in ischemic heart disease patients. Environ Health Perspect 113:440–446.

Herner JD, Aw J, Gao O, Chang DP, Kleeman MJ. 2005. Size and composition distribution of airborne particulate matter in northern California: I-particulate mass, carbon, and water-soluble ions. J Air Waste Manage Assoc 55:30– 51.

Herner JD, Hu SH, Robertson WH, Huai T, Chang MCO, Rieger P, Ayala A. 2011. Effect of advanced aftertreatment for PM and NO(x) reduction on heavy-duty diesel engine ultrafine particle emissions. Environ Sci Technol 45:2413– 2419.

Herner JD, Ying Q, Aw J, Gao O, Chang DPY, Kleeman MJ. 2006. Dominant mechanisms that shape the airborne particle size and composition distribution in central California. Aerosol Sci Technol 40:827–844.

Hertel S, Viehmann A, Moebus S, Mann K, Brocker-Preuss M, Mohlenkamp S, Nonnemacher M, Erbel R, Jakobs H, Memmesheimer M, Jockel KH, Hoffmann B. 2010. Influence of short-term exposure to ultrafine and fine particles on systemic inflammation. Eur J Epidemiol 25:581–592.

Hesterberg TW, Long CM, Lapin CA, Hamade AK, Valberg PA. 2010. Diesel exhaust particulate (DEP) and nanoparticle exposures: What do DEP human clinical studies tell us about potential human health hazards of nanoparticles? Inhal Toxicol 22:679–694.

Hesterberg TW, Long CM, Sax SN, Lapin CA, McClellan RO, Bunn WB, Valberg PA. 2011. Particulate matter in new technology diesel exhaust (NTDE) is quantitatively and qualitatively very different from that found in traditional diesel exhaust (TDE). J Air Waste Manage Assoc 61:894–913.

Hildebrandt K, Rückerl R, Koenig W, Schneider A, Pitz M, Heinrich J, Marder V, Frampton M, Oberdörster G, Wichmann HE, Peters A. 2009. Short-term effects of air pollution: A panel study of blood markers in patients with chronic pulmonary disease. Part Fibre Toxicol 6:25.

Hoek G, Beelen R, Kos G, Dijkema M, van der Zee SC, Fischer PH, Brunekreef B. 2011. Land use regression model for ultrafine particles in Amsterdam. Environ Sci Technol 45:622–628.

Hoek G, Boogaard H, Knol A, De Hartog J, Slottje P, Ayres JG, Borm P, Brunekreef B, Donaldson K, Forastiere F, Holgate S, Kreyling WG, Nemery B, Pekkanen J, Stone V, Wichmann HE, Van der Sluijs J. 2010. Concentration response functions for ultrafine particles and all-cause mortality and hospital admissions: Results of a European expert panel elicitation. Environ Sci Technol 44:476–482.

Hoek G, Kos G, Harrison RM, de Hartog J, Meliefste K, ten Brink H, Katsouyanni K, Karakatsani A, Lianou M, Kotronarou A, Kavouras I, Pekkanen J, Vallius M, Kulmala M, Puustinen A, Thomas S, Meddings C, Ayres J, van Wijnen J, Hameri K. 2008. Indoor-outdoor relationships of particle number and mass in four European cities. Atmos Environ 42:156–169.

Hoekman SK, Gertler A, Brosch A, Robbins C. 2009. Investigation of biodistillates as potential blendstocks for transportation fuels. Final Report, Project No. AVFL-17. Coordinating Research Council, Alpharetta, GA.

Holmes NS, Morawska L. 2006. A review of dispersion modelling and its application to the dispersion of particles: An overview of different dispersion models available. Atmos Environ 40:5902–5928.

Hovorka J, Braniš M. 2011. New particle formation and condensational growth in a large indoor space. Atmos Environ 45:2736–2749.

Hu SH, Herner JD, Shafer M, Robertson W, Schauer JJ, Dwyer H, Collins J, Huai T, Ayala A. 2009. Metals emitted

from heavy-duty diesel vehicles equipped with advanced PM and NO(X) emission controls. Atmos Environ 43:2950–2959.

Ibald-Mulli A, Timonen KL, Peters A, Heinrich J, Wölke G, Lanki T, Buzorius G, Kreyling WG, De Hartog J, Hoek G, Ten Brink HM, Pekkanen J. 2004. Effects of particulate air pollution on blood pressure and heart rate in subjects with cardiovascular disease: A multicenter approach. Environ Health Perspect 112:369–377.

International Commission on Radiological Protection (ICRP). 1994. Human Respiratory Tract Model for Radiological Protection. Pergamon Press, Oxford, England.

Iskandar A, Andersen ZJ, Bonnelykke K, Ellermann T, Andersen KK, Bisgaard H. 2012. Coarse and fine particles but not ultrafine particles in urban air trigger hospital admission for asthma in children. Thorax 67:252–257.

Jacobs L, Nawrot TS, de Geus B, Meeusen R, Degraeuwe B, Bernard A, Sughis M, Nemery B, Panis LI. 2010. Subclinical responses in healthy cyclists briefly exposed to trafficrelated air pollution: An intervention study. Environ Health 9:64.

Janssen NA, de Hartog JJ, Hoek G, Brunekreef B, Lanki T, Timonen KL, Pekkanen J. 2000. Personal exposure to fine particulate matter in elderly subjects: Relation between personal, indoor, and outdoor concentrations. J Air Waste Manag Assoc 50:1133–1143.

Jayaratne ER, Meyer NK, Ristovski ZD, Morawska L, Miljevic B. 2010. Critical analysis of high particle number emissions from accelerating compressed natural gas buses. Environ Sci Technol 44:3724–3731.

Jayaratne ER, Ristovski ZD, Meyer N, Morawska L. 2009. Particle and gaseous emissions from compressed natural gas and ultralow sulphur diesel-fuelled buses at four steady engine loads. Sci Total Environ 407:2845–2852.

Johnston CJ, Finkelstein JN, Gelein R, Oberdörster G. 1998. Pulmonary cytokine and chemokine mRNA levels after inhalation of lipopolysaccharide in C57BL/6 mice. Toxicol Sci 46:300–307.

Johnston CJ, Finkelstein JN, Mercer P, Corson N, Gelein R, Oberdörster G. 2000a. Pulmonary effects induced by ultrafine PTFE particles. Toxicol Appl Pharmacol 168:208–215.

Johnston CJ, Finkelstein JN, Oberdörster G, Reynolds SD, Stripp BR. 1999. Clara cell secretory protein-deficient mice differ from wild-type mice in inflammatory chemokine expression to oxygen and ozone, but not to endotoxin. Exp Lung Res 25:7–21. Johnston CJ, Oberdörster G, Gelein R, Finkelstein JN. 2000b. Newborn mice differ from adult mice in chemokine and cytokine expression to ozone, but not to endotoxin. Inhal Toxicol 12:205–224.

Johnston MV, Klems, JP, Zordan CA, Pennington MR, Smith JN. In press. Selective Detection and Characterization of Nanoparticles from Motor Vehicles. Research Report 173. Health Effects Institute, Boston, MA.

Jung HJ, Kittelson DB, Zachariah MR. 2006. Characteristics of SME biodiesel-fueled diesel particle emissions and the kinetics of oxidation. Environ Sci Technol 40:4949–4955.

Kapp N, Kreyling W, Schulz H, Im Hof V, Gehr P, Semmler M, Geiser M. 2004. Electron energy loss spectroscopy for analysis of inhaled ultrafine particles in rat lungs. Microsc Res Tech 63:298–305.

Karner AA, Eisinger DS, Niemeier DA. 2010. Nearroadway air quality: Synthesizing the findings from realworld data. Environ Sci Technol 44:5334–5344.

Kaur S, Nieuwenhuijsen M, Colvile R. 2005. Personal exposure of street canyon intersection users to $PM_{2.5}$, ultrafine particle counts and carbon monoxide in central London, UK. Atmos Environ 39:3629–3641.

Kelly JT, Avise J, Cai C, Kaduwela AP. 2011. Simulating particle size distributions over California and impact on lung deposition fraction. Aerosol Sci Technol 45:148–162.

Kennedy IM, Wilson D, Barakat AI. 2009. Uptake and Inflammatory Effects of Nanoparticles in a Human Vascular Endothelial Cell Line. Research Report 136. Health Effects Institute, Boston, MA.

Keogh DU, Ferreira L, Morawska L. 2009. Development of a particle number and particle mass vehicle emissions inventory for an urban fleet. Environ Modell Softw 24:1323–1331.

Kettunen J, Lanki T, Tiittanen P, Aalto PP, Koskentalo T, Kulmala M, Salomaa V, Pekkanen J. 2007. Associations of fine and ultrafine particulate air pollution with stroke mortality in an area of low air pollution levels. Stroke 38:918–922.

Khalek IA, Bougher TL, Merritt PM. 2009. Phase 1 of the Advanced Collaborative Emissions Study. Coordinating Research Council, Alpharetta, GA.

Khalek IA, Bougher TL, Merritt PM, Zielinska B. 2011. Regulated and unregulated emissions from highway heavy-duty diesel engines complying with U.S. Environmental Protection Agency 2007 emissions standards. J Air Waste Manage Assoc 61:427–442. Khandoga A, Stoeger T, Khandoga AG, Bihari P, Karg E, Ettehadieh D, Lakatos S, Fent J, Schulz H, Krombach F. 2010. Platelet adhesion and fibrinogen deposition in murine microvessels upon inhalation of nanosized carbon particles. J Thromb Haemost 8:1632–1640.

Kim E, Hopke PK, Larson TV, Covert DS. 2004. Analysis of ambient particle size distributions using unmix and positive matrix factorization. Environ Sci Technol 38:202–209.

Kim JL, Elfman L, Wieslander G, Ferm M, Toren K, Norback D. 2011. Respiratory health among Korean pupils in relation to home, school and outdoor environment. J Korean Med Sci 26:166–173.

Kim S, Sioutas C, Chang MC, Gong H Jr. 2000. Factors affecting the stability of the performance of ambient fine-particle concentrators. Inhal Toxicol 12 Suppl 4:281–298.

Kittelson DB. 1998. Engines and nanoparticles: A review. J Aerosol Sci 29:575–588.

Kittelson DB, Watts W, Johnson J. 2002. Diesel aerosol sampling methodology. CRC E-43 Final Report. Available from: www.crcao.org/reports/recentstudies00-02/E-43%20Final %20Report.pdf.

Kittelson DB, Watts WF, Johnson JP. 2006. On-road and laboratory evaluation of combustion aerosols — Part 1: Summary of diesel engine results. J Aerosol Sci 37:913– 930.

Kleeman MJ, Riddle SG, Jakober CA. 2008a. Size distribution of particle-phase molecular markers during a severe winter pollution episode. Environ Sci Technol 42:6469– 6475.

Kleeman MJ, Riddle SG, Robert MA, Jakober CA. 2008b. Lubricating oil and fuel contributions to particulate matter emissions from light-duty gasoline and heavy-duty diesel vehicles. Environ Sci Technol 42:235–242.

Kleeman MJ, Riddle SG, Robert MA, Jakober CA, Fine PM, Hays MD, Schauer JJ, Hannigan MP. 2009. Source apportionment of fine (PM1.8) and ultrafine (PM0.1) airborne particulate matter during a severe winter pollution episode. Environ Sci Technol 43:272–279.

Kleinman MT, Araujo JA, Nel A, Sioutas C, Campbell A, Cong PQ, Li H, Bondy SC. 2008. Inhaled ultrafine particulate matter affects CNS inflammatory processes and may act via MAP kinase signaling pathways. Toxicol Lett 178:127–130.

Kleinman MT, Hamade A, Meacher D, Oldham M, Sioutas C, Chakrabarti B, Stram D, Froines JR, Cho AK. 2005. Inhalation of concentrated ambient particulate matter near a heavily trafficked road stimulates antigen-induced airway responses in mice. J Air Waste Manage Assoc 55:1277– 1288.

Kleinman MT, Sioutas C, Froines JR, Fanning E, Hamade A, Mendez L, Meacher D, Oldham M. 2007. Inhalation of concentrated ambient particulate matter near a heavily trafficked road stimulates antigen-induced airway responses in mice. Inhal Toxicol 19 Suppl 1:117–126.

Klems JP, Pennington MR, Zordan CA, Johnston MV. 2010. Ultrafine particles near a roadway intersection: Origin and apportionment of fast changes in concentration. Environ Sci Technol 44:7903–7907.

Klems JP, Pennington MR, Zordan CA, McFadden L, Johnston MV. 2011. Apportionment of motor vehicle emissions from fast changes in number concentration and chemical composition of ultrafine particles near a roadway intersection. Environ Sci Technol 45:5637–5643.

Klepczynska Nyström A, Svartengren M, Grunewald J, Pousette C, Rödin I, Lundin A, Sköld CM, Eklund A, Larsson BM. 2010. Health effects of a subway environment in healthy volunteers. Eur Respir J 36:240–248.

Knibbs LD, Cole-Hunter T, Morawska L. 2011. A review of commuter exposure to ultrafine particles and its health effects. Atmos Environ 45:2611–2622.

Knol AB, de Hartog JJ, Boogaard H, Slottje P, van der Sluijs JP, Lebret E, Cassee FR, Wardekker A, Ayres JG, Borm PJ, Brunekreef B, Donaldson K, Forastiere F, Holgate ST, Kreyling WG, Nemery B, Pekkanen J, Stone V, Wichmann HE, Hoek G. 2009. Expert elicitation on ultrafine particles: likelihood of health effects and causal pathways. Part Fibre Toxicol 6:19.

Koponen IK, Asmi A, Keronen P, Puhto K, Kulmala M. 2001. Indoor air measurement campaign in Helsinki, Finland 1999: The effect of outdoor air pollution on indoor air. Atmos Environ 35:1465–1477.

Kraus U, Breitner S, Schnelle-Kreis J, Cyrys J, Lanki T, Rückerl R, Schneider A, Brüske I, Gu JW, Devlin R, Wichmann HE, Zimmermann R, Peters A. 2011. Particleassociated organic compounds and symptoms in myocardial infarction survivors. Inhal Toxicol 23:431–447.

Kreyling W, Semmler-Behnke M, Möller W. 2006a. Health implications of nanoparticles. J Nanopart Res 8:543–562.

Kreyling WG, Dirscherl P, Ferron GA, Heilmann P, Josten M, Miaskowski U, Neuner M, Reitmeir P, Ruprecht L, Schumann G, Takenaka S, Ziesenis A, Heyder J. 1999. Health effects of sulfur-related environmental air pollution. III. Nonspecific respiratory defense capacities. Inhal Toxicol 11:391–422.

Kreyling WG, Möller W, Semmler-Behnke M, Oberdörster G. 2007. Particle dosimetry: Deposition and clearance from the respiratory tract and translocation towards extra-pulmonary sites. In: Particle Toxicology. (Donaldson K, Borm P, eds.). CRC Press, Taylor & Francis Group, Boca Raton.

Kreyling WG, Semmler M, Erbe F, Mayer P, Takenaka S, Schulz H, Oberdörster G, Ziesenis A. 2002. Translocation of ultrafine insoluble iridium particles from lung epithelium to extrapulmonary organs is size dependent but very low. J Toxicol Environ Health A 65:1513–1530.

Kreyling WG, Semmler-Behnke M, Möller W. 2006b. Ultrafine particle-lung interactions: Does size matter? J Aerosol Med 19:74–83.

Kreyling WG, Semmler-Behnke M, Seitz J, Scymczak W, Wenk A, Mayer P, Takenaka S, Oberdörster G. 2009. Size dependence of the translocation of inhaled iridium and carbon nanoparticle aggregates from the lung of rats to the blood and secondary target organs. Inhal Toxicol 21 Suppl 1:55–60.

Kuhlbusch TAJ, Asbach C. 2011. Particle Characterization (Cassee FR, Mills NL, Newby DE, eds). John Wiley & Sons, Hoboken, NJ.

Kuhn T, Biswas S, Sioutas C. 2005. Diurnal and seasonal characteristics of particle volatility and chemical composition in the vicinity of a light-duty vehicle freeway. Atmos Environ 39:7154–7166.

Kukutschova J, Moravec P, Tomasek V, Matejka V, Smolik J, Schwarz J, Seidlerova J, Safarova K, Filip P. 2011. On airborne nano/micro-sized wear particles released from lowmetallic automotive brakes. Environ Pollut 159:998–1006.

Kulmala M, Asmi A, Lappalainen H K, Baltensperger U, Brenguier J-L, Facchini MC, Hansson H-C, Hov Ø, O'Dowd CD, Pöschl U, Wiedensohler A, Boers R, Boucher O, de Leeuw G, Denier van den Gon H, Feichter J, Krejci R, Laj P, Lihavainen H, Lohmann U, McFiggans G, Mentel T, Pilinis C, Riipinen I, Schulz M, Stohl A, Swietlicki E, Vignati E, Amann M, Ammann M, Alves C, Arabas S, Artaxo P, Beddows DCS, Bergström R, Beukes JP, Bilde M, Burkhart JF, Canonaco F, Clegg S, Coe H, Crumeyrolle S, D'Anna B, Decesari S, Gilardoni S, Fischer M, Fjæraa AM, Fountoukis C, George C, Gomes L, Halloran P, Hamburger T, Harrison RM, Herrmann H, Hoffmann T, Hoose C, Hu M, Hõrrak U, Iinuma Y, Iversen T, Josipovic M, Kanakidou M, Kiendler-Scharr A, Kirkevåg A, Kiss G, Klimont Z, Kolmonen P, Komppula M, Kristjánsson J-E, Laakso L, Laaksonen A, Labonnote L, Lanz VA, Lehtinen KEJ,

Makkonen R, McMeeking G, Merikanto J, Minikin A, Mirme S, Morgan WT, Nemitz E, O'Donnell D, Panwar TS, Pawlowska H, Petzold A, Pienaar JJ, Pio C, Plass-Duelmer C, Prévôt ASH, Pryor S, Reddington CL, Roberts G, Rosenfeld D, Schwarz J, Seland Ø, Sellegri K, Shen XJ, Shiraiwa M, Siebert H, Sierau B, Simpson D, Sun JY, Topping D, Tunved P, Vaattovaara P, Vakkari V, Veefkind JP, Visschedijk A, Vuollekoski H, Vuolo R, Wehner B, Wildt J, Woodward S, Worsnop DR, van Zadelhoff G-J, Zardini AA, Zhang K, van Zyl PG, Kerminen V-M, Carslaw KS, Pandis SN. 2011. General overview: European Integrated project on Aerosol Cloud Climate and Air Quality interactions (EUCAARI): Iintegrating aerosol research from nano to global scales. Atmos Chem Phys Discuss 11:17941–18160.

Lahde T, Ronkko T, Virtanen A, Solla A, Kyto M, Soderstrom C, Keskinen J. 2010. Dependence between nonvolatile nucleation mode particle and soot number concentrations in an EGR equipped heavy-duty diesel engine exhaust. Environ Sci Technol 44:3175–3180.

Langrish JP, Li X, Wang S, Lee MM, Barnes GD, Miller MR, Cassee FR, Boon NA, Donaldson K, Li J, Li L, Mills NL, Newby DE, Jiang L. 2011. Reducing personal exposure to particulate air pollution improves cardiovascular health in patients with coronary heart disease. Environ Health Perspect 120:367–372.

Lanki T, Hoek G, Timonen KL, Peters A, Tiittanen P, Vanninen E, Pekkanen J. 2008. Hourly variation in fine particle exposure is associated with transiently increased risk of ST segment depression. Occup Environ Med 65:782–786.

Lanki T, Pekkanen J, Aalto P, Elosua R, Berglind N, D'Ippoliti D, Kulmala M, Nyberg F, Peters A, Picciotto S, Salomaa V, Sunyer J, Tiittanen P, Von Klot S, Forastiere F, for the HSG. 2006. Associations of traffic-related air pollutants with hospitalisation for first acute myocardial infarction: The HEAPSS study. Occup Environ Med 63:844–851.

Laumbach RJ, Rich DQ, Gandhi S, Amorosa L, Schneider S, Zhang JF, Ohman-Strickland P, Gong J, Lelyanov O, Kipen HM. 2010. Acute changes in heart rate variability in subjects with diabetes following a highway traffic exposure. JOM 52:324–331.

Lee D, Wallis C, Wexler AS, Schelegle ES, Van Winkle LS, Plopper CG, Fanucchi MV, Kumfer B, Kennedy IM, Chan JK. 2010. Small particles disrupt postnatal airway development. J Appl Physiol 109:1115–1124.

Lee H, Myung CL, Park S. 2009. Time-resolved particle emission and size distribution characteristics during dynamic engine operation conditions with ethanolblended fuels. Fuel 88:1680–1686. LeFevre ME, Green FH, Joel DD, Laqueur W. 1982. Frequency of black pigment in livers and spleens of coal workers: Correlation with pulmonary pathology and occupational information. Hum Pathol 13:1121–1126.

Leitte AM, Schlink U, Herbarth O, Wiedensohler A, Pan XC, Hu M, Richter M, Wehner B, Tuch T, Wu ZJ, Yang MJ, Liu LQ, Breitner S, Cyrys J, Peters A, Wichmann HE, Franck U. 2011. Size-segregated particle number concentrations and respiratory emergency room visits in Beijing, China. Environ Health Perspect 119:508–513.

Li N, Harkema JR, Lewandowski RP, Wang M, Bramble LA, Gookin GR, Ning Z, Kleinman MT, Sioutas C, Nel AE. 2010. Ambient ultrafine particles provide a strong adjuvant effect in the secondary immune response: Implication for traffic-related asthma flares. Am J Physiol Lung Cell Mol Physiol 299:L374–383.

Lippmann M, Chen L-C. 2009. Health effects of concentrated ambient air particulate matter (CAPs) and its components. Critical Reviews in Toxicology 39:865–913.

Lippmann M, Chen L-C, Gordon T, Kazuhiko I, Thurston GD. In press. Characteristics of Particulate Matter Associated with Health Effects. Research Report 177. Health Effects Institute, Boston, MA.

Ljungman P, Bellander T, Schneider A, Breitner S, Forastiere F, Hampel R, Illig T, Jacquemin B, Katsouyanni K, von Klot S, Koenig W, Lanki T, Nyberg F, Pekkanen J, Pistelli R, Pitsavos C, Rosenqvist M, Sunyer J, Peters A. 2009. Modification of the interleukin-6 response to air pollution by interleukin-6 and fibrinogen polymorphisms. Environ Health Perspect 117:1373–1379.

Lonati G, Crippa M, Gianelle V, Van Dingenen R. 2011. Daily patterns of the multi-modal structure of the particle number size distribution in Milan, Italy. Atmos Environ 45:2434–2442.

Lotti M, Olivato I, Bergamo L. 2009. Inflammation and short-term cardiopulmonary effects of particulate matter. Nanotoxicology 3:27–32.

Lung SC, Kao HY, Peng CW. 2005. Pedestrians' exposure concentrations of $PM_{2.5}$, ultrafine particles and particulate polycyclic aromatic hydrocarbons in Taiwan at intersections with different surroundings. Unpublished manuscript. As cited in: Kaur S, Nieuwenhuijsen MJ, Colvile RN. 2007. Fine particulate matter and carbon monoxide exposure concentrations in urban street transport microenvironments. Atmos Environ 41:4781–4810.

Lwebuga-Mukasa JS, Oyana TJ, Johnson C. 2005. Local ecological factors, ultrafine particulate concentrations,

and asthma prevalence rates in Buffalo, New York, neighborhoods. J Asthma 42:337–348.

Lynch I, Cedervall T, Lundqvist M, Cabaleiro-Lago C, Linse S, Dawson KA. 2007. The nanoparticle-protein complex as a biological entity; a complex fluids and surface science challenge for the 21st century. Adv Colloid Interface Sci 134–135:167–174.

Maier KL, Alessandrini F, Beck-Speier I, Hofer TP, Diabate S, Bitterle E, Stoger T, Jakob T, Behrendt H, Horsch M, Beckers J, Ziesenis A, Hultner L, Frankenberger M, Krauss-Etschmann S, Schulz H. 2008. Health effects of ambient particulate matter — biological mechanisms and inflammatory responses to in vitro and in vivo particle exposures. Inhal Toxicol 20:319–337.

Maricq MM. 2007. Chemical characterization of particulate emissions from diesel engines: A review. J Aerosol Sci 38:1079–1118.

Martin CL, Longley ID, Dorsey JR, Thomas RM, Gallagher MW, Nemitz E. 2009. Ultrafine particle fluxes above four major European cities. Atmos Environ 43:4714–4721.

Mathis U, Mohr M, Forss AM. 2005. Comprehensive particle characterization of modern gasoline and diesel passenger cars at low ambient temperatures. Atmos Environ 39:107–117.

Mauderly JL, McDonald JD. 2012. Advanced Collaborative Emissions Study (ACES) Phase 3A: Characterization of U.S. 2007-Compliant Diesel Engine and Exposure System Operation. Communication 17. Health Effects Institute, Boston, MA.

McCreanor J, Cullinan P, Nieuwenhuijsen MJ, Stewart-Evans J, Malliarou E, Jarup L, Harrington R, Svartengren M, Han IK, Ohman-Strickland P, Chung KF, Zhang J. 2007. Respiratory effects of exposure to diesel traffic in persons with asthma. N Engl J Med 357:2348–2358.

Mejia JF, Choy SL, Mengersen K, Morawska L. 2011. Methodology for assessing exposure and impacts of air pollutants in school children: Data collection, analysis and health effects — A literature review. Atmos Environ 45:813–823.

Metzger KB, Tolbert PE, Klein M, Peel JL, Flanders WD, Todd KH, Mulholland JA, Ryan PB, Frumkin H. 2004. Ambient air pollution and cardiovascular emergency department visits. Epidemiology 15:46–56.

Mills NL, Amin N, Robinson SD, Anand A, Davies J, Patel D, de la Fuente JM, Cassee FR, Boon NA, Macnee W, Millar AM, Donaldson K, Newby DE. 2006. Do inhaled carbon

nanoparticles translocate directly into the circulation in humans? Am J Respir Crit Care Med 173:426–431.

Mills NL, Miller MR, Lucking AJ, Beveridge J, Flint L, Boere AJ, Fokkens PH, Boon NA, Sandstrom T, Blomberg A, Duffin R, Donaldson K, Hadoke PW, Cassee FR, Newby DE. 2011b. Combustion-derived nanoparticulate induces the adverse vascular effects of diesel exhaust inhalation. Eur Heart J 32:2660–2671.

Mills NL, Tornqvist H, Robinson SD, Gonzalez M, Darnley K, MacNee W, Boon NA, Donaldson K, Blomberg A, Sandstrom T, Newby DE. 2005. Diesel exhaust inhalation causes vascular dysfunction and impaired endogenous fibrinolysis. Circulation 112:3930–3936.

Minguillon MC, Arhami M, Schauer JJ, Sioutas C. 2008. Seasonal and spatial variations of sources of fine and quasi-ultrafine particulate matter in neighborhoods near the Los Angeles-Long Beach harbor. Atmos Environ 42:7317–7328.

Mo Y, Wan R, Chien S, Tollerud DJ, Zhang Q. 2009. Activation of endothelial cells after exposure to ambient ultrafine particles: The role of NADPH oxidase. Toxicol Appl Pharmacol 236:183–193.

Möller W, Felten K, Seitz J, Sommerer K, Takenaka S, Wiebert P, Philipson K, Svartengren M, Kreyling WG. 2006. A generator for the production of radiolabelled ultrafine carbonaceous particles for deposition and clearance studies in the respiratory tract. J Aerosol Sci 37:631–644.

Möller W, Felten K, Sommerer K, Scheuch G, Meyer G, Meyer P, Häussinger K, Kreyling WG. 2008. Deposition, retention, and translocation of ultrafine particles from the central airways and lung periphery. Am J Respir Crit Care Med 177: 426–432.

Moore K, Krudysz M, Pakbin P, Hudda N, Sioutas C. 2009. Intra-community variability in total particle number concentrations in the San Pedro Harbor Area (Los Angeles, California). Aerosol Sci Technol 43:587–603.

Morawska L, Ristovski Z, Jayaratne ER, Keogh DU, Ling X. 2008. Ambient nano and ultrafine particles from motor vehicle emissions: Characteristics, ambient processing and implications on human exposure. Atmos Environ 42:8113–8138.

Muranaka M, Suzuki S, Koizumi K, Takafuji S, Miyamoto T, Ikemori R, Tokiwa H. 1986. Adjuvant activity of diesel-exhaust particulates for the production of IgE antibody in mice. J Allergy Clin Immunol 77:616–623.

Nemmar A, Hoet PHM, Vanquickenborne B, Dinsdale D, Thomeer M, Hoylaerts MF, Vanbilloen H, Mortelmans L, Nemery B. 2002. Passage of inhaled particles into the blood circulation in humans. Circulation 105:411–414.

Ntziachristos L, Mamakos A, Samaras Z, Mathis U, Mohr M, Thompson N, Stradling R, Forti L, de Serves C. 2004. Overview of the European "Particulates" Project on the Characterization of Exhaust Particulate Emissions From Road Vehicles: Results for Light-Duty Vehicles. 2004-01-1985. SAE International, Dearborn, MI.

Ntziachristos L, Pistikopoulos P, Samaras Z. 2005. Particle characterization from two-stroke powered two-wheelers. Int J Engine Res 6:263–275.

Nurkiewicz TR, Porter DW, Hubbs AF, Cumpston JL, Chen BT, Frazer DG, Castranova V. 2008. Nanoparticle inhalation augments particle-dependent systemic microvascular dysfunction. Part Fibre Toxicol 5:1.

Nurkiewicz TR, Porter DW, Hubbs AF, Stone S, Chen BT, Frazer DG, Boegehold MA, Castranova V. 2009. Pulmonary nanoparticle exposure disrupts systemic microvascular nitric oxide signaling. Toxicol Sci 110:191–203.

Nurkiewicz TR, Porter DW, Hubbs AF, Stone S, Moseley AM, Cumpston JL, Goodwill AG, Frisbee SJ,Perrotta PL, Brock RW, Frisbee JC, Boegehold MA, Frazer DG, Chen BT, Castranova V. 2011. Pulmonary Particulate Matter and Systemic Microvascular Dysfunction. Research Report 164. Health Effects Institute, Boston, MA.

Nylund N-O, Erkkila K, Lappi M, Ikonen M. 2004. Transit bus emission study: Comparison of emissions from diesel and natural gas buses. Research Report PRO3/P5150/04. VTT Technical Research Centre of Finland, Espoo, Finland.

Oberdörster G. 2010. Safety assessment for nanotechnology and nanomedicine: Concepts of nanotoxicology. J Intern Med 267:89–105.

Oberdörster G, Ferin J, Finkelstein G, Wade P, Corson N. 1990. Increased pulmonary toxicity of ultrafine particles? II. Lung lavage studies. J Aerosol Sci 21:384–387.

Oberdörster G, Ferin J, Gelein R, Soderholm SC, Finkelstein J. 1992. Role of the alveolar macrophage in lung injury: Studies with ultrafine particles. Environ Health Perspect 97:193-199.

Oberdörster G, Finkelstein JN, Johnston C, Gelein R, Cox C, Baggs R, Elder ACP. 2000. Acute Pulmonary Effects of Ultrafine Particles in Rats and Mice. Research Report 96. Health Effects Institute, Cambridge, MA.

Oberdörster G, Gelein RM, Ferin J, Weiss B. 1995. Association of particulate air-pollution and acute mortality — Involvement of ultrafine particles. Inhal Toxicol 7:111– 124.

Oberdörster G, Oberdörster E, and Oberdörster J. 2005. Nanotoxicology: An emerging discipline evolving from studies of ultrafine particles. Environ Health Perspect 113:823–839.

Oberdörster G, Sharp Z, Atudorei V, Elder A, Gelein R, Lunts A, Kreyling W, Cox C. 2002. Extrapulmonary translocation of ultrafine carbon particles following wholebody inhalation exposure of rats. J Toxicol Environ Health A 65:1531–1543.

Ogulei D, Hopke PK, Chalupa DC, Utell MJ. 2007a. Modeling source contributions to submicron particle number concentrations measured in Rochester, New York. Aerosol Sci Technol 41:179–201.

Ogulei D, Hopke PK, Ferro AR, Jaques PA. 2007b. Factor analysis of submicron particle size distributions near a major United States-Canada trade bridge. J Air Waste Manage Assoc 57:190–203.

Ogulei D, Hopke PK, Zhou LM, Pancras JP, Nair N, Ondov JM. 2006. Source apportionment of Baltimore aerosol from combined size distribution and chemical composition data. Atmos Environ 40:S396–S410.

Okamoto RA, Kado NY, Kuzmicky PA, Ayala A, Kobayashi R. 2006. Unregulated emissions from compressed natural gas (CNG) transit buses configured with and without oxidation catalyst. Environ Sci Technol 40:332–341.

O'Neill MS, Veves A, Zanobetti A, Sarnat JA, Gold DR, Economides PA, Horton ES, Schwartz J. 2005. Diabetes enhances vulnerability to particulate air pollution-associated impairment in vascular reactivity and endothelial function. Circulation 111:2913–2920.

Osunsanya T, Prescott G, Seaton A. 2001. Acute respiratory effects of particles: Mass or number? Occup Environ Med 58:154–159.

Park SK, O'Neill MS, Vokonas PS, Sparrow D, Schwartz J. 2005. Effects of air pollution on heart rate variability: The VA normative aging study. Environ Health Perspect 113:304–309.

Parker JL, Larson RR, Eskelson E, Wood EM, Veranth JM. 2008. Particle size distribution and composition in a mechanically ventilated school building during air pollution episodes. Indoor Air 18:386–393.

Peel JL, Tolbert PE, Klein M, Metzger KB, Flanders WD, Knox T, Mulholland JA, Ryan PB, Frumkin H. 2005. Ambient air pollution and respiratory emergency department visits. Epidemiology 16:164–174.

Pekkanen J, Kulmala M. 2004. Exposure assessment of ultrafine particles in epidemiologic time-series studies. Scand J Work Environ Health 30 Suppl 2:9–18.

Pekkanen J, Peters A, Hoek G, Tiittanen P, Brunekreef B, de Hartog J, Heinrich J, Ibald-Mulli A, Kreyling WG, Lanki T, Timonen KL, Vanninen E. 2002. Particulate air pollution and risk of ST-segment depression during repeated submaximal exercise tests among subjects with coronary heart disease: The exposure and risk assessment for fine and ultrafine particles in ambient air (ULTRA) study. Circulation 106:933–938.

Pekkanen J, Timonen KL, Ruuskanen J, Reponen A, Mirme A. 1997. Effects of ultrafine and fine particles in urban air on peak expiratory flow among children with asthmatic symptoms. Environ Res 74:24–33.

Penttinen P, Timonen KL, Tiittanen P, Mirme A, Ruuskanen J, Pekkanen J. 2001a. Number concentration and size of particles in urban air: Effects on spirometric lung function in adult asthmatic subjects. Environ Health Perspect 109:319–323.

Penttinen P, Timonen KL, Tiittanen P, Mirme A, Ruuskanen J, Pekkanen J. 2001b. Ultrafine particles in urban air and respiratory health among adult asthmatics. Eur Respir J 17:428–435.

Péry AR, Brochot C, Hoet PH, Nemmar A, Bois FY. 2009. Development of a physiologically based kinetic model for 99m-technetium-labelled carbon nanoparticles inhaled by humans. Inhal Toxicol 21:1099–1107.

Peters A, Breitner S, Cyrys J, Stölzel M, Pitz M, Wölke G, Heinrich J, Kreyling W, Küchenhoff H, Wichmann H-E. 2009. The Influence of Improved Air Quality on Mortality Risks in Erfurt, Germany. Research Report 137. Health Effects Institute, Boston, MA.

Peters A, von Klot S, Heier M, Trentinaglia I, Cyrys J, Hörmann A, Hauptmann M, Wichmann HE, Löwell H. 2005. Part I. Air pollution, personal activities, and onset of myocardial infarction in a case–crossover study. In: Particulate Air Pollution and Nonfatal Cardiac Events. Research Report 124. Health Effects Institute, Boston, MA.

Peters A, von Klot S, Heier M, Trentinaglia I, Hormann A, Wichmann HE, Lowel H. 2004. Exposure to traffic and the onset of myocardial infarction. N Engl J Med 351:1721–1730.

Peters A, Wichmann HE, Tuch T, Heinrich J, Heyder J. 1997. Respiratory effects are associated with the number of ultrafine particles. Am J Respir Crit Care Med 155:1376–1383.

Pey J, Querol X, Alastuey A, Rodriguez S, Putaud JP, Van Dingenen R. 2009. Source apportionment of urban fine and ultra-fine particle number concentration in a Western Mediterranean city. Atmos Environ 43:4407–4415.

Pietropaoli AP, Frampton MW, Hyde RW, Morrow PE, Oberdörster G, Cox C, Speers DM, Frasier LM, Chalupa DC, Huang LS, Utell MJ. 2004a. Pulmonary function, diffusing capacity, and inflammation in healthy and asthmatic subjects exposed to ultrafine particles. Inhal Toxicol 16 Suppl 1:59–72.

Pietropaoli AP, Frampton MW, Oberdörster G, Cox C, Huang LS, Marder V, Utell MJ. 2004b. Blood markers of coagulation and inflammation in healthy human subjects exposed to carbon ultrafine particles. (Heinrich U, ed). INIS Monographs, Fraunhofer IRB Verlag, Stuttgart, Germany.

Pinkerton KE, Zhou Y, Teague SV, Peake JL, Walther RC, Kennedy IM, Leppert VJ, Aust AE. 2004. Reduced lung cell proliferation following short-term exposure to ultrafine soot and iron particles in neonatal rats: Key to impaired lung growth? Inhal Toxicol 1:73–81.

Pinkerton KE, Zhou Y, Zhong C, Smith KR, Teague SV, Kennedy IM, Ménache MG. 2008. Mechanisms of Particulate Matter Toxicity in Neonatal and Young Adult Rat Lungs. Research Report 135. Health Effects Institute, Boston, MA.

Polidori A, Arhami M, Sioutas C, Delfino RJ, Allen R. 2007. Indoor/outdoor relationships, trends, and carbonaceous content of fine particulate matter in retirement homes of the Los Angeles basin. J Air Waste Manage Assoc 57:366–379.

Pope CA 3rd, Dockery DW. 2006. Health effects of fine particulate air pollution: Lines that connect. J Air Waste Manag Assoc 56:709–742.

Pope CA 3rd, Schwartz J, Ransom MR. 1992. Daily mortality and $\rm PM_{10}$ pollution in Utah Valley. Arch Environ Health 47:211–217.

Prisby RD, Muller-Delp J, Delp MD, Nurkiewicz TR. 2008. Age, gender, and hormonal status modulate the vascular toxicity of the diesel exhaust extract phenanthraquinone. J Toxicol Environ Health A 71:464–470.

Pui DY, Qi C, Stanley N, Oberdörster G, Maynard A. 2008. Recirculating Air Filtration Significantly Reduces Exposure to Airborne Nanoparticles. Environ Health Perspect 116:863–866. Puustinen A, Hämeri K, Pekkanen J, Kulmala M, de Hartog J, Meliefste K, ten Brink H, Kos G, Katsouyanni K, Karakatsani A, Kotronarou A, Kavouras I, Meddings C, Thomas S, Harrison R, Ayres JG, van der Zee S, Hoek G. 2007. Spatial variation of particle number and mass over four European cities. Atmos Environ 41:6622–6636.

Qi C, Stanley N, Pui DY, Kuehn TH. 2008. Laboratory and on-road evaluations of cabin air filters using number and surface area concentration monitors. Environ Sci Technol 42:4128–4132.

Ren CZ, Fang SN, Wright RO, Suh H, Schwartz J. 2011. Urinary 8-hydroxy-2'-deoxyguanosine as a biomarker of oxidative DNA damage induced by ambient pollution in the Normative Aging Study. Occup Environ Med 68:562–569.

Rich DQ, Schwartz J, Mittleman MA, Link M, Luttmann-Gibson H, Catalano PJ, Speizer FE, Dockery DW. 2005. Association of short-term ambient air pollution concentrations and ventricular arrhythmias. Am J Epidemiol 161:1123–1132.

Rich DQ, Zareba W, Beckett W, Hopke PK, Oakes D, Frampton MW, Bisognano J, Chalupa D, Bausch J, O'Shea K, Wang Y, Utell MJ. 2012. Are ambient ultrafine, accumulation mode, and fine particles associated with adverse cardiac responses in patients undergoing cardiac rehabilitation? Environ Health Perspect 120:1162–1169.

Riddle SG, Robert MA, Jakober CA, Hannigan MP, Kleeman MJ. 2008. Size-resolved source apportionment of airborne particle mass in a roadside environment. Environ Sci Technol 42:6580–6586.

Rim D, Siegel J, Spinhirne J, Webb A, McDonald-Buller E. 2008. Characteristics of cabin air quality in school buses in Central Texas. Atmos Environ 42:6453–6464.

Robert MA, VanBergen S, Kleeman MJ, Jakober CA. 2007. Size and composition distributions of particulate matter emissions: Part 1 — Light-duty gasoline vehicles. J Air Waste Manage Assoc 57:1414–1428.

Rothen-Rutishauser B, Muhlfeld C, Blank F, Musso C, Gehr P. 2007. Translocation of particles and inflammatory responses after exposure to fine particles and nanoparticles in an epithelial airway model. Part Fibre Toxicol 4:9.

Rückerl R, Greven S, Ljungman P, Aalto P, Antoniades C, Bellander T, Berglind N, Chrysohoou C, Forastiere F, Jacquemin B, von Klot S, Koenig W, Kuchenhoff H, Lanki T, Pekkanen J, Perucci CA, Schneider A, Sunyer J, Peters A. 2007a. Air pollution and inflammation (interleukin-6, C-reactive protein, fibrinogen) in myocardial infarction survivors. Environ Health Perspect 115:1072–1080. Rückerl R, Ibald-Mulli A, Koenig WA, Schneider A, Woelke G, Cyrys J, Heinrich J, Marder V, Frampton M, Wichmann HE, Peters A. 2006. Air pollution and markers of inflammation and coagulation in patients with coronary heart disease. Am J Respir Crit Care Med 173:432–441.

Rückerl R, Phipps RP, Schneider A, Frampton M, Cyrys J, Oberdörster G, Wichmann HE, Peters A. 2007b. Ultrafine particles and platelet activation in patients with coronary heart disease — results from a prospective panel study. Part Fibre Toxicol 4:1.

Rundell KW, Hoffman JR, Caviston R, Bulbulian R, Hollenbach AM. 2007. Inhalation of ultrafine and fine particulate matter disrupts systemic vascular function. Inhal Toxicol 19:133–140.

Rundell KW, Slee JB, Caviston R, Hollenbach AM. 2008. Decreased lung function after inhalation of ultrafine and fine particulate matter during exercise is related to decreased total nitrate in exhaled breath condensate. Inhal Toxicol 20:1–9.

Samet JM, Dominici F, Zeger SL, Schwartz J, Dockery DW. 2000a. Part I. Methods and methodologic issues. In: The National Morbidity, Mortality, and Air Pollution Study. Research Report 94. Health Effects Institute, Cambridge, MA.

Samet JM, Graff D, Berntsen J, Ghio AJ, Huang YC, Devlin RB. 2007. A comparison of studies on the effects of controlled exposure to fine, coarse and ultrafine ambient particulate matter from a single location. Inhal Toxicol 19 Suppl 1:29–32.

Samet JM, Rappold A, Graff D, Cascio WE, Berntsen JH, Huang YC, Herbst M, Bassett M, Montilla T, Hazucha MJ, Bromberg PA, Devlin RB. 2009. Concentrated ambient ultrafine particle exposure induces cardiac changes in young healthy volunteers. Am J Respir Crit Care Med 179:1034–1042.

Samet JM, Zeger SL, Dominici F, Curriero F, Coursac I, Dockery DW, Schwartz J, Zanobetti A. 2000b. Part II. Morbidity and mortality from air pollution in the United States. In: The National Morbidity, Mortality, and Air Pollution Study. Research Report 94. Health Effects Institute, Cambridge, MA.

Sanders PG, Xu N, Dalka TM, Maricq MM. 2003. Airborne brake wear debris: Size distributions, composition, and a comparison of dynamometer and vehicle tests. Environ Sci Technol 37:4060–4069.

Sardar SB, Fine PM, Yoon H, Sioutas C. 2004. Associations between particle number and gaseous co-pollutant

concentrations in the Los Angeles basin. J Air Waste Manage Assoc 54:992–1005.

Sarnat JA, Koutrakis P, Suh HH. 2000. Assessing the relationship between personal particulate and gaseous exposures of senior citizens living in Baltimore, MD. J Air Waste Manag Assoc 50:1184–1198.

Sarnat SE, Coull BA, Ruiz PA, Koutrakis P, Suh HH. 2006a. The influences of ambient particle composition and size on particle infiltration in Los Angeles, CA residences. J Air Waste Manage Assoc 56:186–196.

Sarnat SE, Coull BA, Schwartz J, Gold DR, Suh HH. 2006b. Factors affecting the association between ambient concentrations and personal exposures to particles and gases. Environ Health Perspect 114:649–654.

Schneider A, Hampel R, Ibald-Mulli A, Zareba W, Schmidt G, Schneider R, Rückerl R, Couderc JP, Mykins B, Oberdörster G, Wölke G, Pitz M, Wichmann HE, Peters A. 2010. Changes in deceleration capacity of heart rate and heart rate variability induced by ambient air pollution in individuals with coronary artery disease. Part Fibre Toxicol 7:29.

Schürch S, Gehr P, Im Hof V, Geiser M, Green F. 1990. Surfactant displaces particles toward the epithelium in airways and alveoli. Respir Physiol 80:17–32.

Seaton A, Dennekamp M. 2003. Hypothesis: Ill health associated with low concentrations of nitrogen dioxide — an effect of ultrafine particles? Thorax 58:1012–1015.

Seaton A, MacNee W, Donaldson K, Godden D. 1995. Particulate air pollution and acute health effects. Lancet 345:176–178.

Seigneur C. 2009. Current understanding of ultrafine particulate matter emitted from mobile sources. J Air Waste Manage Assoc 59:3–17.

Semmler M, Seitz J, Erbe F, Mayer P, Heyder J, Oberdörster G, Kreyling WG. 2004. Long-term clearance kinetics of inhaled ultrafine insoluble iridium particles from the rat lung, including transient translocation into secondary organs. Inhal Toxicol 16:453–459.

Semmler-Behnke M, Takenaka S, Fertsch S, Wenk A, Seitz J, Mayer P, Oberdörster G, Kreyling WG. 2007. Efficient elimination of inhaled nanoparticles from the alveolar region: Evidence for interstitial uptake and subsequent reentrainment onto airways epithelium. Environ Health Perspect 115:728–733.

Sgro LA, Borghese A, Speranza L, Barone AC, Minutolo P, Bruno A, D'Anna A, D'Alessio A. 2008. Measurements of

nanoparticles of organic carbon and soot in flames and vehicle exhausts. Environ Sci Technol 42:859–863.

Shah AP, Pietropaoli AP, Frasier LM, Speers DM, Chalupa DC, Delehanty JM, Huang LS, Utell MJ, Frampton MW. 2008. Effect of inhaled carbon ultrafine particles on reactive hyperemia in healthy human subjects. Environ Health Perspect 116:375–380.

Sinclair AH, Tolsma D. 2004. Associations and lags between air pollution and acute respiratory visits in an ambulatory care setting: 25-month results from the aerosol research and inhalation epidemiological study. J Air Waste Manage Assoc 54:1212–1218.

Sioutas C, Delfino RJ, Singh M. 2005. Exposure assessment for atmospheric ultrafine particles (UFP) and implications in epidemiologic research. Environ Health Perspect 113:947–955.

Sioutas C, Kim S, Chang M, Terrell LL, Gong H Jr. 2000. Field evaluation of a modified DataRAM MIE scattering monitor for real-time $PM_{2.5}$ mass concentration measurements. Atmos Environ 34:4829–4838.

Song S, Lee K, Lee YM, Lee JH, Il Lee S, Yu SD, Paek D. 2011. Acute health effects of urban fine and ultrafine particles on children with atopic dermatitis. Environmental Research 111:394–399.

Stanier CO, Khlystov AY, Pandis SN. 2004. Nucleation events during the Pittsburgh air quality study: Description and relation to key meteorological, gas phase, and aerosol parameters. Aerosol Sci Technol 38:253-264.

Stanier CO, Lee SR. In progress. Development and Application of a Personal Exposure Screening Model for Size-Resolved Urban Aerosols. Research Report. Health Effects Institute, Boston, MA.

Stewart JC, Chalupa DC, Devlin RB, Frasier LM, Huang LS, Little EL, Lee SM, Phipps RP, Pietropaoli AP, Taubman MB, Utell MJ, Frampton MW. 2010. Vascular effects of ultrafine particles in persons with type 2 diabetes. Environ Health Perspect 118:1692–1698.

Stölzel M, Breitner S, Cyrys J, Pitz M, Wölke G, Kreyling W, Heinrich J, Wichmann HE, Peters A. 2007. Daily mortality and particulate matter in different size classes in Erfurt, Germany. J Expo Sci Environ Epidemiol 17:458– 467.

Stölzel M, Peters A, Wichmann HE. 2003. Daily mortality and fine and ultrafine particles in Erfurt, Germany. In: Revised Analyses of Time-Series Studies of Air Pollution and Health; Revised Analyses of Selected Time-Series Studies. Special Report. Health Effects Institute, Boston, MA.

Strak M, Boogaard H, Meliefste K, Oldenwening M, Zuurbier M, Brunekreef B, Hoek G. 2010. Respiratory health effects of ultrafine and fine particle exposure in cyclists. Occup Environ Med 67:118–124.

Takafuji S, Suzuki S, Koizumi K, Tadokoro K, Miyamoto T, Ikemori R, Muranaka M. 1987. Diesel-exhaust particulates inoculated by the intranasal route have an adjuvant activity for IgE production in mice. J Allergy Clin Immunol 79:639–645.

Takano H, Yoshikawa T, Ichinose T, Miyabara Y, Imaoka K, Sagai M. 1997. Diesel exhaust particles enhance antigeninduced airway inflammation and local cytokine expression in mice. Am J Respir Crit Care Med 156:36–42.

Takenaka S, Karg E, Kreyling W, Lentner B, Möller W, Behnke-Semmler M, Jennen L, Walch A, Michalke B, Schramel P. 2006. Distribution pattern of inhaled ultrafine gold particles in the rat lung. Inhal Toxicol 18:733–740.

Takenaka S, Karg E, Roth C, Schulz H, Ziesenis A, Heinzmann U, Schramel P, Heyder J. 2001. Pulmonary and systemic distribution of inhaled ultrafine silver particles in rats. Environ Health Perspect 109 Suppl 4: 547–551.

Takenaka S, Möller W, Semmler-Behnke M, Karg E, Wenk A, Schmid O, Stoeger T, Jennen L, Aichler M, Walch A, Pokhrel S, Mädler L, Eickelberg O, Kreyling WG. 2012. Efficient internalization and intracellular translocation of inhaled gold nanoparticles in rat alveolar macrophages. Nanomedicine (Lond) 7:855–865.

Tatli E, Clark NN. 2009. Crankcase particulate emissions from diesel engines. SAE Int J Fuels Lubr 1:1334–1344.

Terzano C, Di Stefano F, Conti V, Graziani E, Petroianni A. 2010. Air pollution ultrafine particles: Toxicity beyond the lung. Eur Rev Med Pharmacol Sci 14:809–821.

Timonen KL, Hoek G, Heinrich J, Bernard A, Brunekreef B, De Hartog J, Hameri K, Ibald-Mulli A, Mirme A, Peters A, Tiittanen P, Kreyling WG, Pekkanen J. 2004. Daily variation in fine and ultrafine particulate air pollution and urinary concentrations of lung Clara cell protein CC16. Occup Environ Med 61:908–914.

Timonen KL, Vanninen E, De Hartog J, Ibald-Mulli A, Brunekreef B, Gold DR, Henrich J, Hoek G, Lanki T, Peters A, Tarkiainen T, Tiittanen P, Kreyling W, Pekkanen J. 2006. Effects of ultrafine and fine particulate and gaseous air pollution on cardiac autonomic control in subjects with coronary artery disease: The ULTRA study. J Expo Sci Environ Epidemiol 16:332–341. Toner SM, Shields LG, Sodeman DA, Prather KA. 2008. Using mass spectral source signatures to apportion exhaust particles from gasoline and diesel powered vehicles in a freeway study using UF-ATOFMS. Atmos Environ 42:568–581.

Tuch TM, Herbarth O, Franck U, Peters A, Wehner B, Wiedensohler A, Heintzenberg J. 2006. Weak correlation of ultrafine aerosol particle concentrations < 800 nm between two sites within one city. J Expo Sci Environ Epidemiol 16:486–491.

Tzamkiozis T, Ntziachristos L, Samaras Z. 2010. Diesel passenger car PM emissions: From Euro 1 to Euro 4 with particle filter. Atmos Environ 44:909–916.

U.S. Environmental Protection Agency. 2009. Integrated Science Assessment for Particulate Matter: Final Report. EPA/600/R-08/139F. National Center for Environmental Assessment-RTP Division, Office of Research and Development, Research Triangle Park, NC. Available from www.epa.gov/ttn/naaqs/standards/pm/s_pm_2007_ isa.html.

U.S. Environmental Protection Agency. 2010. Renewable Fuel Standard Program (RFS2) Regulatory Impact Analysis. EPA-420-R-10-006. Assessment and Standards Division, Office of Transportation and Air Quality. Washington, D.C.

U.S. National Research Council. 1998. Research Priorities for Airborne Particulate Matter: I. Immediate Priorities and a Long-Range Research Portfolio. National Academy Press, Washington, D.C.

U.S. National Research Council. 1999. Research Priorities for Airborne Particulate Matter: II. Evaluating Research Progress and Updating the Portfolio. National Academy Press, Washington, D.C.

U.S. National Research Council. 2001. Research Priorities for Airborne Particulate Matter: III. Early Research Progress. National Academy Press, Washington, D.C.

U.S. National Research Council. 2004. Research Priorities for Airborne Particulate Matter: IV Continuing Research Progress. Committee on Research Priorities for Airborne Particulate Matter. National Academy Press, Washington, D.C.

Utell MJ, Frampton MW. 2000. Acute health effects of ambient air pollution: The ultrafine particle hypothesis. J Aerosol Med 13:355–359.

Utell MJ, Frampton MW, Zareba W, Devlin RB, Cascio WE. 2002. Cardiovascular effects associated with air pollution:

Potential mechanisms and methods of testing. Inhal Toxicol 14:1231–1247.

Vedal S, Campen MJ, Kaufman JD, Larson TV, Sampson PD, Sheppard EZ, Simpson CD, Szpiro AA, McDonald JD, Lund AK. In press. Integrated Epidemiologic and Toxicologic Cardiovascular Studies to Identify Toxic Components and Sources of Fine Particulate Matter. Research Report 178. Health Effects Institute, Boston, MA.

Venkatachari P, Hopke KP, Brune WH, Ren X, Lesher R, Mao J, Mitchell M. 2007. Characterization of wintertime reactive oxygen species concentrations in Flushing, New York. Aerosol Sci Technol 41:97–111.

Vinzents PS, Møller P, Sørensen M, Knudsen LE, Herte LQ, Jensen FP, Schibye B, Loft S. 2005. Personal exposure to ultrafine particles and oxidative DNA damage. Environ Health Perspect 113:1485–1490.

Von Klot S, Peters A, Aalto P, Bellander T, Berglind N, D'Ippoliti D, Elosua R, Hormann A, Kulmala M, Lanki T, Lowel H, Pekkanen J, Picciotto S, Sunyer J, Forastiere F. 2005. Ambient air pollution is associated with increased risk of hospital cardiac readmissions of myocardial infarction survivors in five European cities. Circulation 112:3073–3079.

Von Klot S, Wölke G, Tuch T, Heinrich J, Dockery DW, Schwartz J, Kreyling WG, Wichmann HE, Peters A. 2002. Increased asthma medication use in association with ambient fine and ultrafine particles. Eur Resp J 20:691– 702.

Wan R, Mo YQ, Zhang X, Chien SF, Tollerud DJ, Zhang QW. 2008. Matrix metalloproteinase-2 and-9 are induced differently by metal nanoparticles in human monocytes: The role of oxidative stress and protein tyrosine kinase activation. Toxicol Appl Pharmacol 233:276–285.

Wang YG, Hopke PK, Chalupa DC, Utell MJ. 2010. Longterm characterization of indoor and outdoor ultrafine particles at a commercial building. Environ Sci Technol 44:5775–5780.

Wang YG, Hopke PK, Chalupa DC, Utell MJ. 2011. Effect of the shutdown of a coal-fired power plant on urban ultrafine particles and other pollutants. Aerosol Sci Technol 45:1245–1249.

Weichenthal S, Dufresne A, Infante-Rivard C, Joseph L. 2008. Characterizing and predicting ultrafine particle counts in Canadian classrooms during the winter months: Model development and evaluation. Environ Res 106:349– 360. Weichenthal S, Kulka R, Dubeau A, Martin C, Wang D, Dales R. 2011. Traffic-related air pollution and acute changes in heart rate variability and respiratory function in urban cyclists. Environ Health Perspect 119:1373–1378.

Westerdahl D, Fruin S, Sax T, Fine PM, Sioutas C. 2005. Mobile platform measurements of ultrafine particles and associated pollutant concentrations on freeways and residential streets in Los Angeles. Atmos Environ 39:3597– 3610.

Whitlow TH, Hall A, Zhang KM, Anguita J. 2011. Impact of local traffic exclusion on near-road air quality: Findings from the New York City "Summer Streets" campaign. Environ Pollut 159:2016–2027.

Wichmann HE, Spix C, Tuch T, Wölke G, Peters A, Heinrich J, Kreyling WG, Heyder J. 2000. Part I: Role of particle number and particle mass. In: Daily Mortality and Fine and Ultrafine Particles in Erfurt, Germany. Research Report 98. Health Effects Institute, Cambridge, MA.

Wiebert P, Sanchez-Crespo A, Falk R, Philipson K, Lundin A, Larsson S, Möller W, Kreyling WG, Svartengren M. 2006a. No significant translocation of inhaled 35-nm carbon particles to the circulation in humans. Inhal Toxicol 18:741–747.

Wiebert P, Sanchez-Crespo A, Seitz J, Falk R, Philipson K, Kreyling WG, Möller W, Sommerer K, Larsson S, Svartengren M. 2006b. Negligible clearance of ultrafine particles retained in healthy and affected human lungs. Eur Resp J 28:286–290.

World Health Organization. 2006. WHO Air quality guidelines for particulate matter, ozone, nitrogen dioxide, and sulfur dioxide. Global Update 2005. Summary of risk assessment. Available from http://whqlibdoc.who.int/hq/ 2006/WHO_SDE_PHE_OEH_06.02_eng.pdf.

Yang GS, Teague S, Pinkerton K, Kennedy IM. 2001. Synthesis of an ultrafine iron and soot aerosol for the evaluation of particle toxicity. Aerosol Sci Technol 35:759–766.

Yue W, Schneider A, Stölzel M, Rückerl R, Cyrys J, Pan X, Zareba W, Koenig W, Wichmann HE, Peters A. 2007. Ambient source-specific particles are associated with prolonged repolarization and increased levels of inflammation in male coronary artery disease patients. Mutat Res 621:50–60.

Zareba W, Couderc JP, Oberdörster G, Chalupa D, Cox C, Huang LS, Peters A, Utell MJ, Frampton MW. 2009. ECG parameters and exposure to carbon ultrafine particles in young healthy subjects. Inhal Toxicol 21:223–233. Zeka A, Sullivan JR, Vokonas PS, Sparrow D, Schwartz J. 2006. Inflammatory markers and particulate air pollution: Characterizing the pathway to disease. Int J Epidemiol 35:1347–1354.

Zhang J, McCreanor JE, Cullinan P, Chung KF, Ohman-Strickland P, Han I-K, Järup L, Nieuwenhuijsen MJ. 2009. Health Effects of Real-World Exposure to Diesel Exhaust in Persons with Asthma. Research Report 138. Health Effects Institute, Boston, MA.

Zhang KM, Wexler AS, Zhu YF, Hinds WC, Sioutas C. 2004. Evolution of particle number distribution near roadways. Part II: the "road-to-ambient" process. Atmos Environ 38:6655–6665.

Zhang Q. In press. Activation of Endothelial cells and Gene Expression in Lungs Following Exposure to Ultrafine Particles. Research Report. Health Effects Institute, Boston, MA.

Zhang Q, Zhu Y. 2010. Measurements of ultrafine particles and other vehicular pollutants inside school buses in south Texas. Atmos Environ 44:253–261.

Zhong CY, Zhou YM, Smith KR, Kennedy IM, Chen CY, Aust AE, Pinkerton KE. 2010. Oxidative injury in the lungs of neonatal rats following short-term exposure to ultrafine iron and soot particles. J Toxicol Environ Health A 73:837– 847.

Zhou Y, Levy JI. 2007. Factors influencing the spatial extent of mobile source air pollution impacts: A meta-analysis. BMC Public Health 7:89.

Zhou YM, Zhong CY, Kennedy IM, Leppert VJ, Pinkerton KE. 2003a. Oxidative stress and NF kappa B activation in the lungs of rats: A synergistic interaction between soot and iron particles. Toxicol Appl Pharmacol 190:157–169.

Zhou YM, Zhong CY, Kennedy IM, Pinkerton KE. 2003b. Pulmonary responses of acute exposure to ultrafine iron particles in healthy adult rats. Environ Toxicol 18:227– 235.

Zhu Y. (in press). Assessing Children's Exposure to Ultrafine Particles from Vehicular Emissions. Research Report. Health Effects Institute, Boston, MA. Zhu Y, Eiguren-Fernandez A, Hinds WC, Miguel AH. 2007. In-cabin commuter exposure to ultrafine particles on Los Angeles freeways. Environ Sci Technol 41:2138–2145.

Zhu Y, Hinds WC, Kim S, Shen S, Sioutas C. 2002. Study of ultrafine particles near a major highway with heavy-duty diesel traffic. Atmos Environ 36:4323–4335.

Zhu Y, Hinds WC, Krudysz M, Kuhn T, Froines J, Sioutas C. 2005. Penetration of freeway ultrafine particles into indoor environments. J Aerosol Sci 36:303–322.

Zhu Y, Hinds WC, Shen S, Sioutas C. 2004. Seasonal trends of concentration and size distribution of ultrafine particles near major highways in Los Angeles. Aerosol Sci Technol 38:5–13.

Zhu Y, Kuhn T, Mayo P, Hinds WC. 2006. Comparison of daytime and nighttime concentration profiles and size distributions of ultrafine particles near a major highway. Environmental Science and Technology 39:2531–2536.

Zordan CA, Pennington MR, Johnston MV. 2010. Elemental composition of nanoparticles with the nano aerosol mass spectrometer. Anal Chem 82:8034–8038.

Zuurbier M, Hoek G, Oldenwening M, Lenters V, Meliefste K, van den Hazel P, Brunekreef B. 2010. Commuters' exposure to particulate matter air pollution is affected by mode of transport, fuel type, and route. Environ Health Perspect 118:783–789.

Zuurbier M, Hoek G, Oldenwening M, Meliefste K, Krop E, van den Hazel P, Brunekreef B. 2011a. In-traffic air pollution exposure and CC16, blood coagulation, and inflammation markers in healthy adults. Environ Health Perspect 119:1384–1389.

Zuurbier M, Hoek G, Oldenwening M, Meliefste K, van den Hazel P, Brunekreef B. 2011b. Respiratory effects of commuters' exposure to air pollution in traffic. Epidemiology 22:219–227.

Report or Study Title	Summary	Related Journal Publications
nt		
ld Evaluation of anofilm Detectors r Measuring cidic Particles in door and Outdoor ir	Dr. Cohen and her colleagues at New York University School of Medicine tested the performance of a novel iron nanofilm substrate in the laboratory and the field to collect and measure sulfuric acid ultrafine particles under a variety of temperature and humidity conditions. They had hypothesized that acidic ultrafine particles could be particularly damaging to the lung. Particles were predominantly in the 35–100 nm range but could range up to 300 nm. The method was partially successful but required further development	Hazi et al. 2003
ective Detection id haracterization of anoparticles from otor Vehicles	In this study, the authors field tested a Nanoaerosol Mass Spectrometer they had developed to provide real-time measurement and composition of nanoparticles particles sized 20–30 nm. They used the NAMS with other state-of-the art instruments to determine the composition of the nanoparticles and to identify the contribution of individual motor vehicles to ambient nanoparticles concentrations at a traffic intersection.	Klems et al. 2010; Klems et al. 2011; Zordan et al. 2010
evelopment and pplication of a arsonal exposure rreening model for ze-resolved urban srosols"	Dr. Stanier is constructing an exposure estimation model based on several existing models to predict ultrafine particle formation from secondary photochemical processes and from vehicle exhaust in large urban areas.	
ssessing children's posure to ltrafine particles om vehicular missions"	Dr. Zhu is measuring ultrafine particle levels in classrooms and school buses under a variety of conditions, including different diesel engine exhaust after treatment devices, traffic density, and ventilation.	Zhang and Zhu 2010
	Tabl	le continues next page

Appendix Table A	1.1 (Continued). Overview	/ of HEI Research Program on UFPs	
Research Report (RR) #; Author, Date	Report or Study Title	Summary	Related Journal Publications
Animal and in vit	tro Exposures		
RR 96; Oberdörster et al. 2000	Acute Pulmonary Effects of Ultrafine Particles in Rats and Mice	Dr. Oberdörster and colleagues at the University of Rochester School of Medicine and Dentistry hypothesized that inhaled ultrafine particles induce an inflammatory response in the airways of mice and rats and that animals with preexisting airway inflammatory conditions may be particularly vulnerable. The investigators focused on inhaled carbon and platinum particles because these elements are constituents of particles found in urban atmospheres. The investigators tested a small number of young and old mice and rats that were healthy or had preexisting airway inflammatory conditions. Pulmonary inflammation was evaluated by measurement of cellular and biochemical parameters in bronchoalveolar lavage fluid, focusing on increases in the percentage of neutrophils and production of reactive oxygen species, which appeared to be the most sensitive indicators of a response. The HEI Health Review Committee concluded that the study provided little evidence that inhaled ultrafine particles cause inflammation, a result it thought could be attributed to small numbers of animals and experiments, and uncertainties about the animal model, the relative toxicity of the particles studied. More research was advised.	Elder et al. 2000a; Elder et al. 2000b; Elder et al. 2000c; Johnston et al. 1999; Johnston et al. 2000a; 2000a; Johnston et al. 2000b
RR 129; Hahn et al. 2005	Particle Size and Composition Related to Adverse Health Effects in Aged, Sensitive Rats	Dr. Hahn and colleagues systematically examined lung inflammation in young adult and old rats after inhalation of fine particles (< 2.5 µm) and ultrafine particles (< 0.1 µm, ~median diameter 35 nm) of different composition: relatively inert carbon and vanadium pentoxide (V_2O_5) , which contains the transition metal vanadium, known to cause toxic effects upon inhalation in humans in occupational settings. In addition, they examined the effect of a short-term increase (spike) in particle exposure concentration on inflammatory response. They found greater retention of ultrafine V_2O_5 particles and more evidence of inflammation in aged rats compared with fine particles. Spikes in ultrafine and fine particle levels increased the inflammatory response at some, but not all	
RR 135; Pinkerton et al. 2008	Mechanisms of Particulate Matter Toxicity in Neonatal and Young Adult Rat Lungs	Dr. Pinkerton and colleagues designed a study to determine whether the biologic response to inhaled ultrafine particles depends on particle composition. Neonatal and young adult rats were exposed to laboratory-generated ultrafine metal particles (mean 70–80 nm), either alone or in combination with soot (60% EC, 40% OC), and their lungs examined for oxidative stress, inflammation, and injury. The results suggested that some markers of oxidative stress and inflammation were higher for exposures to combinations of iron and soot than to either particle type alone.	Pinkerton et al. 2004; Yang et al. 2001; Zhou et al. 2003a; Zhou et al. 2003b
		Tabl	e continues next page

87

Appendix Table 1	A.1 (Continued). Overview	r of HEI Research Program on UFPs	
Research Report (RR) #; Author, Date	Report or Study Title	Summary	Related Journal Publications
Animal and in vi	tro Exposures (<i>Continued</i>		
RR 136; Kennedy et al. 2009	Uptake and Inflammatory Effects of Nanoparticles in a Human Vascular Endothelial Cell Line	This report describes a one-year study to evaluate which physicochemical characteristics (size, shape, agglomeration, calculated surface area) of metal nanoparticles may contribute to their toxicity. Nanoparticles of oxides of iron, zinc, yttrium, and cerium were generated in the laboratory and characterized in detail, followed by preliminary studies of particle uptake and inflammatory effects in human endothelial cells exposed to these nanoparticles. The study found that different metal oxide particles induced different levels of biologic responses, but could not distinguish whether these effects were also attributable to differences in their physical properties. In particular, biological responses were inversely correlated with calculated surface area, an unexpected and unexplained result.	Gojova et al. 2007; Gojova et al. 2009
RR 164; Nurkiewicz et al. 2011	Pulmonary Particulate Matter and Systemic Microvascular Dysfunction	Dr. Nurkiewicz and colleagues evaluated whether exposure to fine or nano-sized titanium dioxide (TiO ₂) particles may affect cardiovascular endpoints, in particular endothelium-dependent vascular dilation. Rats were exposed via inhalation to 0.5 to 20 mg/m^3 TiO ₂ for up to 12 hours and evaluated for vascular dilation and for markers of oxidative stress, coagulation, and inflammation. The study demonstrated that the degree of systemic microvascular dysfunction associated with PM exposure is related to the size and lung burden of the particles, that the nanoparticles were more potent at an equivalent lung burden, and that was mediated by decreased NO production in endothelial cells. Whether this effect of nanoparticles was related to particle surface area could not be determined. Evidence of lung inflammation was weak and a role of systemic inflammatory mediators was not clearly established.	Nurkiewicz et al. 2008; Nurkiewicz et al. 2009; Prisby et al. 2008
In review; Q. Zhang et al.	"Activation of endothelial cells and gene expression in lungs following exposure to ultrafine particles"	Dr. Zhang is investigating effects of exposure to ultrafine particles on endothelial cells in the lungs of apolipoprotein E (ApoE) knockout mice and their healthy counterparts.	Mo et al. 2009; Wan et al. 2008
RR 177; Lippmann et al., in press	National Particle Component Toxicity Study: Health Effects of PM Components in Mice, and Cells In vitro	Using 3 different size fractions of concentrated ambient particles collected in 5 US locations representing diverse ambient air pollution conditions, the authors conducted subchronic inhalation studies in mice in order to identify the $PM_{2.5}$ constituents most responsible for acute and cumulative effects; They also investigated the role of chemical composition in $PM_{10-2.5}$, $PM_{2.5-0.2}$, and $PM_{<0.2}$ on in vitro and in vivo acute toxicity in the same five locations.	Lippmann and Chen 2009
		Tab	le continues next pag

Appendix Table A	A.1 (Continued). Overviev	v of HEI Research Program on UFPs	
Research Report (RR) #; Author, Date	Report or Study Title	Summary	Related Journal Publications
Human Controlle	d Exposures		
RR 118; Gong et al. 2003	Controlled Exposures of Health and Asthmatic Volunteers to Concentrated Ambient Particles in Metropolitan Los Angeles	Funded at the same time as the Frampton study, the Gong study focused on exposures to 12 healthy and 12 mildly asthmatic individuals but to 'worst case' concentrations of fine particles concentrated from ambient air in Los Angeles (mean 174 μ g/m ³). The two studies shared similar designs, exposure protocols, and health endpoints in order to provide a better basis for comparison of the results for fine and ultrafine particles. The study found that their CAPs exposures induced few cardiovascular, pulmonary function, or airway or systemic inflammatory effects in either healthy or asthmatic individuals. For the few significant changes observed, effects were small and demonstrated no consistent patterns. Sensitivity of the study for the most endpoints appeared sufficient.	Chang et al. 2000; Gong 2000; Gong et al. 2002; Kim et al. 2000; Sioutas et al. 2000
RR 126; Frampton et al. 2004	Effects of Exposure to Ultrafine Carbon Particles in Healthy Subjects and Subjects with Asthma	Dr. Frampton and his colleagues evaluated the effects of exposing 12 healthy and 16 mildly asthmatic men and women to laboratory-generated ultrafine carbon particles (elemental carbon, count median diameter is 25 nm). They hypothesized that ultrafine particle exposure would activate leukocytes and endothelial cells and lead to an inflammatory response in the airway and in the blood; and that it also might affect respiration and cardiac electrophysiologic function. The investigators evaluated markers of airway inflammation in sputum and in blood, and pulmonary and cardiac function before, during, and after a 2-hour exposure to concentrations of $10-25$ µg/m ³ particles while at rest or with intermittent exercise. The number and mass of inhaled particles that deposited in the lungs of the participants were calculated. The study found few ultrafine particle-associated airway, systemic, or cardiac electrophysiologic changes in either healthy or asthmatic individuals. The HEI Review Committee concluded that the clinical significance of the small changes that were observed was not clear and that additional study was needed to provide a stronger test of the hypothesis that ultrafine particles are more toxic than larger ones.	Chalupa et al. 2002; Chalupa et al. 2004; Daigle et al. 2003; Frampton 2001; Pietropaoli et al. 2000; Pietropaoli et al. 2004b; Utell, Frampton 2000; Utell et al. 2002
RR 138; J. Zhang et al. 2009	Health Effects of Real- World Exposure to Diesel Exhaust in Persons with Asthma	Dr. Zhang and colleagues evaluated how inhaling air with a high concentration of diesel exhaust from vehicular traffic while walking on a busy street in Central London might affect people who had either mild or moderate asthma. They evaluated pulmonary function parameters, bronchial reactivity, and markers of airway inflammation and oxidative stress in participants who walked for two hours along Oxford Street, where vehicles are predominantly diesel-powered, and in Hyde Park, where there is urban background air pollution and no traffic.	McCreanor et al. 2007
		1001	econtinues next page

89

Appendix Table /	A.1 (Continued). Overview	of HEI Research Program on UFPs	
Research Report (RR) #; Author, Date	Report or Study Title	Summary	Related Journal Publications
Epidemiology			
RR 98; Wichmann et al. 2000	Daily Mortality and Fine and Ultrafine Particles in Erfurt, Germany, Part 1: Role of Particle Number and Particle Mass	Dr. H-Erich Wichmann and colleagues at the National Research Center for Environment and Health in Germany prospectively studied the association of daily mortality data with size-fractionated mass and number concentrations of ultrafine and fine particles in Erfurt, Germany, using Poisson regression techniques. Concentrations were measured near a road and a time-series approach was used to look at short-term changes in particle concentration and concurrent deaths due to cardiovascular and respiratory causes over a period of 3.5 years. The study found similar associations between mortality and ultrafine and fine particulate size fractions. The HEI Review Committee however concluded the study did not provide support for a relative or temporal difference in the effects of ultrafine and fine particles.	
RR 124 Part I; Peters et al. 2005	Air pollution, personal activities, and onset of myocardial infarction in a case–crossover study. In: Particulate Air Pollution and Nonfatal Cardiac Events	Dr. Peters and her colleagues evaluated the association between nonfatal myocardial infarction (MI) and exposure to particulate matter just prior to the event. 691 patients in hospitals in Augsburg, Germany, who survived an MI were asked to provide hourly details about their activities 4 days before MI onset. The investigators used a case–crossover analysis to determine whether exposure to pollutants was associated with onset of MI. They measured levels of ultrafine particles, $PM_{2.5}$ and PM_{10} in ambient air and obtained information about levels of associations were found between elevated concentrations of particles. No associations were found between elevated concentrations of particles. No associations were found between elevated concentrations of particles and MI hours later, or between ultrafine particles and MI onset within 5 days.	Peters et al. 2004; Stölzel et al. 2007
		Tab	e continues next pag

Appendix Table A	A.1 (Continued). Overview	of HEI Research Program on UFPs	
Research Report (RR) #; Author, Date	Report or Study Title	Summary	Related Journal Publications
Epidemiology (Co	intinued)		
RR 124 Part II; Dockery et al. 2005b	Association of air pollution with confirmed arrhythmias recorded by implanted defibrillators. In: Particulate Air Pollution and Nonfatal Cardiac Events	Dr. Dockery and colleagues assessed the correlation between short-term increases in ambient concentrations of particulate matter and the risk of possibly life-threatening arrhythmias in patients with implanted cardioverter defibrillators (ICDs). An ICD is programmed to respond when the heart rate exceeds a preset number of beats per minute; it records and stores the heart rate exceeds a preset number of beats per minute; it records and stores the heart rate exceeds and responde and (if necessary) delivers an electrical stimulus to return the heart rate to a normal rhythm. The investigators studied 195 patients from Boston, MA, who had either single or dual-chamber ICD's and used logistic regression models to determine whether exposure to pollutants ($PM_{2,5}$, ultrafine particles, black carbon, sulfate, NO ₂ , CO, SO ₂ , and ozone) was associated with arrhythmias. In addition, they evaluated patients' clinical information in some analyses to determine whether specific characteristics (e.g., diagnosis at ICD implantation, medication use, occurrence of multiple arrhythmics) would modify a pollutant's effects. The study found that nonfatal arrhythmics (total particle number) and sulfate. Of all pollutants evaluated for ultrafines (total particle number) and sulfate. Of all pollutants evaluated. SO ₂ showed the strongest and most robust associations with anbient concentrations of $PM_{2,5}$ or black carbon up to 3 days before the event. Similar findings were reported for ultrafines (total particle number) and sulfate. Of all pollutants evaluated, SO_2 showed the strongest and most robust associations af though associations were also observed with NO_2 and OO .	Jockery et al. 2005a; Rich et al. 2005
Special Report 17 HEI 2010	Traffic-Related Air Pollution: A Critical Review of the Literature on Emissions, Exposure, and Health Effects	This report is the most comprehensive and systematic review to date of the scientific literature on emissions, exposure, and health effects from traffic-related air pollution. It includes conclusions about the populations exposed around major roads, the associations between exposure to air pollution from traffic and human health, and important remaining data gaps.	
Communication 16 HEI 2011	The Future of Vehicle Fuels and Technologies: Anticipating Health Benefits and Challenges	This report reviews new vehicle fuels and technologies that are likely to be commercially available within the next 10 years in the United States and other industrialized countries at a level that could result in significant population exposure. It highlights expected changes in emissions and other effects from the use of each technology and fuel, along with any life-cycle and regulatory issues. This Communication was prepared by the Special Committee on Emerging Technologies (SCET) whose 18 members include government, academic, industrial, and other experts from diverse fields, including automotive engineering and emissions, emissions control technologies, new fuels, electric vehicles, and environmental modeling. The report identifies many high-priority areas for research. In response, the HEI Research Committee has developed an Action Plan that describes the steps the Committee is taking, or plans to take, to address the high-priority issues identified by SCET.	

Appendix Tabl Location ^a	le B.1. Primary	r Research Art	ticles Present	ing Results of 7	Fotal and UF	'P NC Epidemio	logic Studies by Outco	ome and Geographic
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two- Pollutant Models? ^d
Mortality								
Prague, Czech Republic	Braniš et al. 2010	Time series	2006	NC _{0.05-0.2}	SMPS	Central P	VC ₃₂ , NC ₁₁₅ , VC ₃₆₅ , PM _{2.5}	No
London, England	Atkinson et al. 2010	Time series	2000–2005	Total NC	CPC	Central H	$M_{2.5}$	PM _{2.5}
Helsinki, Finland	Halonen et al. 2009	Time series	1998–2004	NC _{0.03-0.1}	DMPS	Central I r I	M _{2.5} , nucleation node, Aitken mode, .ccumulation mode, M _{2.5-10}	No
	Kettunen et al. 2007	Time series	1998–2004	NC _{<0.1}	DMPS	Central I	VO ₂ , O ₃ , CO, PM _{2.5} , M ₁₀ , PM _{2.5-10}	$PM_{2.5}$, CO
Erfurt, Germany	Breitner et al. 2009	Time series	1991–2002	NC _{0.01-0.1}	AS	Central (20, NO ₂ , PM _{2.5} , •M ₁₀	CO, NO ₂ , PM _{2.5} , PM ₁₀
	Peters et al. 2009	Time series	1991–2002	NC _{0.003} -0.064, NC _{0.01} -0.03, NC_{0.01}-0.1	AS	Central C	50, NO ₂ , SO ₂ , O ₃ , VC _{0.01-0.1} , NC _{0.01-2.5} , M _{2.5} , PM ₁₀	CO, NO ₂ , O ₃
	Stölzel et al. 2003	Time series	1995–1998	NC _{0.01-0.1}	MAS	Central	Vot reported	SO ₂ , NO ₂ , CO, PM _{2.5}
	Stölzel et al. 2007	Time series	1995–2001	NC _{0.01-0.1}	MAS	Central C	50, NO ₂ , NO, PM ₁₀ , VC, and MC size ractions	NO, NO ₂ , CO
	Wichmann et al. 2000	Time series	1995–1998	NC _{0.01-0.1}	MAS	Central (20, NO ₂ , SO ₂ , PM _{2.5}	CO, NO ₂ , SO ₂ , PM _{2.5}
							Tabl	le continues next page
a APC = aerosol pa = differential mol = polycyclic aron mobility particle	rticle counter; AS : bility particle sizer, attic hydrocarbons sizer; WSOC = wat	= aerosol spectror ; EAS = electric au ; PMF = positive) er soluble organic	meter; BC = blac] erosol spectromε matrix factorizat : carbon.	k carbon; BS = blacl ster; MAS = mobile ion; SMPS = scanni	k smoke; CPC = aerosol spectror ing mobility par	condensation partic neter; MC = mass co ticle sizer; SOC – se	le counter; DMA = differenti ncentration; NC = particle n condary organic carbon; TD	al mobility analyzer; DMPS umber concentration; PAHs MPS = twin differential
^b Study abbreviatio RUPIOH = Relatio Ultrafine Particle	ns: AIRGENE = Atl onship between Ul s in Air.	aens, Augsburg, Ba trafine and fine Pa	arcelona, Helsink articulate matter	ci, Rome, Stockholm in Indoor and Outc	ı; NAS = Normat door air and res _f	ive Aging Study; PE. piratory Health; ULT	ACE = Pollution Effects in As RA = Exposure and Risk Ase	thmatic Children in Europe; sessment for Fine and
c Diameter size ran < 0.100 µm in dia	iges are in microme imeter or UFP NC).	sters (μm). Severa	l articles conside	ered multiple size f	ractions. The bo	lded metric is the or	ie closest to the definition us	sed in this document (i.e.,
^d These are two-po	llutant models unl	ess otherwise ind	icated.					

Appendix Tab Geographic Lo	le B.1 (<i>Continued</i> cation ^a). Primary Re	search Article:	s Presenting Res	ults of Total a	nd UFP NC Epi	demiologic Studies by	y Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Mortality (Con	tinued)							
Rome, Italy	Forastiere et al. 2005	Case– crossover	1998–2000	Total NC	CPC	Central	CO, NO ₂ , SO ₂ , O ₃ , PM ₁₀	PM ₁₀
Beijing, China	Breitner et al. 2011	Time series	2004–2005	NC _{0.003} –0.03, NC _{0.03} –0.1 (also related MC and SC size fractions)	TDMPS	Central	Between all particle size fractions	Selected NC size fractions
Morbidity								
Prague, Czech Republic	Branis et al. 2010	Time series	2006	NC _{0.05-0.2}	SMPS	Central	PM _{2.5} , NC ₃₂ , 115, 365	No
Copenhagen, Denmark	Andersen et al. 2008b	Time series	2001–2004	NC _{0.06-0.7} NC _{<0.1}	DMPS	Central	NO ₂ , O ₃ , CO, other NC size fractions, PM ₁₀ , PM _{2.5}	$CO, NO_2, O_3, PM_{10}, PM_{2.5}$ Multi-pollutant models: NC size fractions, PM_{10}
	Andersen et al. 2010	Case– crossover	2003–2006	NC _{<0.1}	DMPS	Central	NO _x , CO, PM ₁₀	NO _x , CO, PM ₁₀
	Iskandar et al. 2012	Case– crossover	2001– 2008	NC _{0.01-0.7}	DMPS	Central	NO _x , NO ₂ , PM ₁₀ , PM _{2.5}	NO _X , NO ₂ , PM ₁₀ , PM _{2.5}
London, England	Atkinson et al. 2010	Time series	2000-2005	Total NC	CPC	Central	PM _{2.5} , PM ₁₀ , PM _{2.5-10} , NO ₃ ⁻ , SO ₄ ²⁻ , BS	Particles PM _{2.5} , PM ₁₀ , BS
							Π	able continues next page
a APC = aerosol p mobility particle hydrocarbons; P soluble organic c	urticle counter; AS = aer sizer; EAS = electric a MF = positive matrix fa arbon.	rosol spectromete erosol spectromet actorization; SMP?	r; BC = black carbo er; MAS = mobile S = scanning mobi	n; BS = black smoke aerosol spectrometer lity particle sizer; SC	; CPC = condensat r; MC = mass conc 0C – secondary org	ion particle counter entration; NC = par ganic carbon; TDMP	; DMA = differential mobilit ticle number concentration; S = twin differential mobilit	y analyzer; DMPS = differential PAHs = polycyclic aromatic ty particle sizer; WSOC = water
^b Study abbreviati Relationship bet	ons: AIRGENE = Athens ween Ultrafine and fine	s, Augsburg, Barce e Particulate matte	lona, Helsinki, Roı sr in Indoor and O	ne, Stockholm; NAS utdoor air and respir	= Normative Agin ratory Health; ULT	g Study; PEACE = P RA = Exposure and	ollution Effects in Asthmatic Risk Assessment for Fine a	. Children in Europe; RUPIOH = nd Ultrafine Particles in Air.
^c Diameter size rai diameter or UFP	ages are in micrometers NC).	s (μm). Several art	icles considered m	ultiple size fractions	s. The bolded metı	ric is the one closest	to the definition used in thi	s document (i.e., < 0.100 µm in

Appendix Tab Geographic Lc	le B.1 (Continue cation ^a	ed). Primary Res	search Article	s Presenting Resi	ults of Total a	nd UFP NC Epi	demiologic Studies by	y Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Morbidity (Co	ntinued)							
Helsinki, Finland	Halonen et al. 2008	Time series	1998–2004	NC _{<0.03} NC _{0.03-0.1}	DMPS	Central	VO ₂ , CO, PM _{2.5} , O ₃	NO ₂ , CO, PM _{2.5}
	Halonen et al. 2009	Time series	1998–2004	NC _{0.03-0.1}	DMPS	Central I	² M _{2.5} , nucleation node, Aitken mode, nccumulation mode, ² M _{2.5-10}	No
Leipzig, Germany	Franck et al. 2011	Panel (cross- sectional)	1997–2002	NC _{>0.3} NC _{>0.5}	APC	Personal I	$^{9}M_{1}$, $PM_{2.5}$	No
Rome, Italy	Belleudi et al. 2010	Case- crossover	2001–2005	Total NC	CPC	Central	$^{2}\mathrm{M}_{10},\mathrm{PM}_{2.5}$	PM ₁₀ , PM _{2.5}
5-Cities (HEAPPS)	Lanki et al. 2006	Time series	1992–2000	Total NC	CPC & retrospective estimation	Central	CO, NO ₂ , O ₃ , PM ₁₀	O ₃ , PM ₁₀
	Von Klot et al. 2005	Time series	1992–2001	Total NC	CPC & retrospective estimation	Central (30, NO ₂ , O ₃ , PM ₁₀	O ₃ , PM ₁₀
Atlanta, GA	Metzger et al. 2004	Time series	1993–2000	NC _{0.01-0.1}	SMPS	Central (D ₃ , NO ₂ , CO, SO ₂ , PM ₁₀ , PM _{2.5}	No
	Peel et al. 2005	Time series	1993–2000	NC _{0.01-0.1}	SMPS	Central (20, NO ₂ , PM _{2.5} , OC, PM _{2.5} , EC	No
	Sinclair and Tolsma 2004	Time series	1998–2000	PM 0.01–0.1 area	SMPS	Central	Йо	No
							L	able continues next page
a APC = aerosol p mobility particle hydrocarbons; P soluble organic e	article counter; AS = : • sizer; EAS = electric MF = positive matrix arbon.	aerosol spectrometer : aerosol spectromete : factorization; SMPS	r; BC = black carbc er; MAS = mobile) = scanning mobi	on; BS = black smoke; s aerosol spectrometer ility particle sizer; SO	CPC = condensat ; MC = mass conc .C – secondary or	ion particle counter; entration; NC = part ganic carbon; TDMP	DMA = differential mobilit iele number concentration; S = twin differential mobili	y analyzer; DMPS = differential PAHs = polycyclic aromatic ty particle sizer; WSOC = water
^b Study abbreviati Relationship bet ^c Diameter size raı	ons: AIRGENE = Athé ween Ultrafine and fi 1ges are in micromete	ans, Augsburg, Barcel ine Particulate matte 3rs (µm). Several arti	lona, Helsinki, Ro r in Indoor and C cles considered m	me, Stockholm; NAS : Jutdoor air and respir nultiple size fractions.	= Normative Agin atory Health; ULI . The bolded meti	g Study; PEACE = Pc TRA = Exposure and ric is the one closest	ullution Effects in Asthmatic Risk Assessment for Fine a to the definition used in thi	: Children in Europe; RUPIOH = nd Ultrafine Particles in Air. is document (i.e., < 0.100 µm in
diameter or UFF	NC).	,		4				-

Appendix Tab Geographic Lo	le B.1 (<i>Continue</i> cation ^a	:d). Primary Re	search Article	s Presenting Res	sults of Total a	nd UFP NC Ep	idemiologic Studies by	Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Morbidity (Co	ntinued)							
Beijing, China	Leitte et al. 2011	Time series	2004–2006	NC< 0.1	TDMPS	Central	NO ₂ , SO ₂ , NC size fractions	NO ₂ , PM ₁₀
Respiratory Sy	mptoms							
Copenhagen, Denmark	Andersen et al. 2008a	Panel (repeated measure)	1998–2004	NC _{0.1-0.7}	DMPS	Central	NO ₂ , NO _x , CO, O ₃ , PM _{2.5} , PM ₁₀	NO ₂ , NO _x , CO, PM ₁₀
Kuopio, Finland	Tiitanen et al. 1999	Panel (repeated measure)	1995	NC _{0.01-0.1}	EAS	Central	CO, SO ₂ , NO ₂ , O ₃ , PM ₁₀ , PM _{2.5} , PN _{0.1-1.0} , BC	PM _{2.5} , PM ₁₀
Helsinki, Finland	Penttinen et al. 2001b	Panel (repeated measure)	1996–1997	NC0.01-0.1, NC0.01-10	EAS + CPC	Central	CO, NO, NO2, PM ₁₀ , PM2,5-10, PM2,5, PM ₁ , NC0,01-0.1, NC0,1-1, NC0,01-10	CO, NO, NO ₂ (with NC _{0.01-10} only)
Erfurt, Germany	Peters et al. 1997	Panel (repeated measure)	1991–1992	NC _{0.01-2.5} , NC _{0.01-0.1} , NC _{0.01-0.5} , NC _{0.5-2.5} , (and related MC fractions)	Electrical mobility analyzer + CPC	Central	NC size fractions, MC fractions, PM ₁₀	PM ₁₀ , MC _{0.1-0.5}
	Von Klot et al. 2002	Panel (repeated measure)	1996–1997	NC 0.01–0.1 [,] NC _{0.1–0.5} , NC _{0.5–2.5} , (and related MC fractions)	DMA + CPC	Central	NO ₂ , CO, SO ₂ , NC _{0.1-0.5} , NC _{0.5-2.5} , MC _{0.1-0.5} , MC _{0.5-2.5} , PM _{2.5-10} , PM ₁₀	NO_2 , SO_2 , CO
							Tc	tble continues next page
a APC = aerosol pe mobility particle hydrocarbons; P? soluble organic c	urticle counter; AS = : sizer; EAS = electric MF = positive matrix arbon.	aerosol spectromete : aerosol spectromet : factorization; SMP;	rr; BC = black carbc ter; MAS = mobile S = scanning mobi	on; BS = black smoke aerosol spectromete ility particle sizer; SC	;; CPC = condensat r; MC = mass conc DC - secondary org	ion particle counte: :entration; NC = par ganic carbon; TDMI	r: DMA = differential mobility ticle number concentration; l SS = twin differential mobility	analyzer; DMPS = differential 2AHs = polycyclic aromatic y particle sizer; WSOC = water
^b Study abbreviatic Relationship bet	ons: AIRGENE = Athε veen Ultrafine and fi	ens, Augsburg, Barce ine Particulate matt	elona, Helsinki, Ro er in Indoor and C	me, Stockholm; NAS Jutdoor air and respii	= Normative Agin ratory Health; ULI	ig Study; PEACE = F TRA = Exposure and	ollution Effects in Asthmatic I Risk Assessment for Fine ar	Children in Europe; RUPIOH = .d Ultrafine Particles in Air.
^c Diameter size rar diameter or UFP	nges are in micromete NC).	ers (µm). Several art	icles considered n	nultiple size fraction:	s. The bolded metı	ric is the one closes	t to the definition used in this	i document (i.e., < 0.100 μm in

Appendix Tab Geographic Lo	le B.1 (<i>Continu</i> cation ^a	e d). Primary Re	search Article	s Presenting Res	ults of Total a	nd UFP NC E _l	videmiologic Studies by	Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	e Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Respiratory S	ymptoms (Conti	nued)						
Aberdeen, Scotland	Osunsanya et al. 2001	Panel (repeated measure)	NA	NC _{< 0.1}	SMPS	Central	PM_{10}	No
Amsterdam, Erfurt, Helsinki (ULTRA study)	de Hartog et al. 2003	Panel (repeated measure)	1998–1999	NC 0.01-0.1 Total NC	AS + CPC	Central	Total NC, PM _{2.5}	Yes, not reported
Incheon, South Korea	Song et al. 2011	Panel (repeated measure)	2009	NC0.01-0.1' NC _{0.11-1} ' NC _{0.01-1}	SMPS + CPC	Central	NO ₂ , SO ₂ , CO, PM ₁ , PM _{2.5} , PM ₁₀	PM ₁₀ , PM _{2.5} , PM ₁
Pulmonary Fu	nction							
Kuopio, Finland (PEACE study)	Pekkanen et al. 1997	Panel (repeated measure)	1994	$\begin{array}{l} NC_{0.010-0.032},\\ NC_{0.032-0.10},\\ NC_{0.10-0.32},\\ NC_{0.32-1.0},\\ NC_{1.0-3.2},\\ NC_{1.0-3.2},\\ NC_{3.2-10} \end{array}$	EAS	Central	NO, NO ₂ , CO, SO ₂ , black smoke, NC size fractions	NO ₂ , NO, SO ₂
	Tiitanen et al. 1999	Panel (repeated measure)	1995	NC _{0.01-0.1}	EAS	Central	CO, SO ₂ , NO ₂ , O ₃ , PM ₁₀ , PM _{2.5} , NC _{0.1-1.0} , BC	PM _{2.5} , PM ₁₀ , NO ₂ , SO ₂ , O ₃
							Ta	ble continues next page
a APC = aerosol p mobility particle hydrocarbons; P. soluble organic c	article counter; AS = sizer; EAS = electrid MF = positive matrix :arbon.	aerosol spectromete c aerosol spectromet ć factorization; SMP	r; BC = black carbc er; MAS = mobile S = scanning mobi	m; BS = black smoke aerosol spectromete: lity particle sizer; SC	; CPC = condensat r; MC = mass conc 0C – secondary org	ion particle count entration; NC = pe ganic carbon; TDM	r; DMA = differential mobility tritcle number concentration; I PS = twin differential mobility	analyzer; DMPS = differential AHs = polycyclic aromatic / particle sizer; WSOC = water
^b Study abbreviati Relationship bet	ons: AIRGENE = Ath ween Ultrafine and f	ens, Augsburg, Barce ïne Particulate matt	elona, Helsinki, Ro er in Indoor and O	me, Stockholm; NAS utdoor air and respi	= Normative Agin ratory Health; ULT	g Study; PEACE = 'RA = Exposure an	Pollution Effects in Asthmatic d Risk Assessment for Fine an	Children in Europe; RUPIOH = d Ultrafine Particles in Air.
^c Diameter size ra. diameter or UFP	nges are in micromet NC).	ers (µm). Several art	icles considered m	uultiple size fraction	s. The bolded metr	ic is the one close	st to the definition used in this	document (i.e., < 0.100 µm in
Appendix Tab Geographic Lo	le B.1 (<i>Continue</i> cation ^a	ed). Primary Re	search Article	s Presenting Res	ults of Total a	md UFP NC Ep	idemiologic Studies by	Outcome and
--	--	--	--	---	--	---	---	---
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Pulmonary Fu	nction (Continu	ed)						
Helsinki, Finland	Penttinen et al. 2001a	Panel (repeated measure)	1996–1997	NC0.01-0.1' NC _{0.1-1} ' NC _{0.01-10}	EAS	Central	PM ₁₀ , PM _{2.5} , PM ₁ , NC size fractions	No
	Penttinen et al. 2001b	Panel (repeated measure)	1996–1997	NC0.01-0.1' NC0.01-10	EAS	Central	CO, NO, NO ₂ , PM ₁₀ , PM _{2.5-10} , PM _{2.5} , PM ₁ , NC _{0.01-0.1} , NC _{0.1-1} , NC _{0.01-10}	For NC _{0.01–10} only: CO, NO, NO ₂
Erfurt, Germany	Peters et al. 1997	Panel (repeated measure)	1991–1992	NC _{0.01-2.5} , NC _{0.01-0.1} , NC _{0.1-0.5} , NC _{0.5-2.5} , (and related MC fractions)	Electrical mobility analyzer + CPC	Central	NC size fractions, MC fractions, PM ₁₀	PM ₁₀ , MC _{0,1-0.5}
Aberdeen, Scotland	Osunsanya et al. 2001	Panel (repeated measure)	NA	NC<0.1	SMPS	Central	PM_{10}	No
Amsterdam Athens Birmingham Helsinki (RUPIOH study)	de Hartog et al. 2010	Panel (repeated measure)	2002-2004	Total NC	CPC	Central, outdoors, indoors	PM ₁₀ , PM _{2.5} , PM _{2.5-10} Ta	No ble continues next page
a APC = aerosol pe mobility particle hydrocarbons; Pì soluble organic c	urticle counter; AS = sizer; EAS = electric <i>M</i> F = positive matrix arbon.	aerosol spectromete : aerosol spectromet factorization; SMP;	er; BC = black carbt ter; MAS = mobile S = scanning mobi	on; BS = black smoke aerosol spectrometer ility particle sizer; SC	; CPC = condensat r; MC = mass conc DC – secondary org	tion particle counte: centration; NC = par ganic carbon; TDMI	r; DMA = differential mobility ticle number concentration; F S = twin differential mobility	anal yzer; DMPS = differential AHs = polycyclic aromatic ^ particle sizer; WSOC = water
^b Study abbreviatic Relationship bet	ons: AIRGENE = Athe ween Ultrafine and fi	ens, Augsburg, Barce ine Particulate matte	əlona, Helsinki, Ro er in Indoor and C	me, Stockholm; NAS Jutdoor air and respir	= Normative Agin ratory Health; ULI	ıg Study; PEACE = F IRA = Exposure and	'ollution Effects in Asthmatic (l Risk Assessment for Fine an	Children in Europe; RUPIOH = d Ultrafine Particles in Air.
^c Diameter size rar diameter or UFP	iges are in micromet(NC).	ers (µm). Several art	ticles considered n	nultiple size fraction:	s. The bolded metı	ric is the one closes	t to the definition used in this	document (i.e., < 0.100 µm in
^d These are two-pc	illutant models unle	ss otherwise indicat	ted.					

97

Appendix Tab Geographic Lo	le B.1 (<i>Continu</i>є cation ^a	ə d). Primary Res	search Article:	s Presenting Res	ults of Total a	nd UFP NC Ep.	idemiologic Studies b	y Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Pulmonary Fu	nction (Continu	ed)						
Arnhem, the Netherlands	Zuurbier et al. 2011b	Panel (repeated measure)	2007–2008	Total NC	CPC	Personal	PM _{2.5} , PM ₁₀ , soot	No
Ottawa, Canada	Weichenthal et al. 2011	Panel (repeated measure)	2010	NC _{≤0.1}	CPC	Personal	CO, SO ₂ , NO ₂ , O ₃ , PM _{2.5} , BC	O ₃ , BC
HRV								
Erfurt, Germany	Schneider et al. 2010	Panel (repeated measure)	2000–2001	NC _{0.01-0.1}	MAS	Central	PM _{2.5} , EC, OC	No
Aberdeen, Scotland	Barclay et al. 2009	Panel (repeated measure)	2003–2005	NC _{0.01-0.1}	SMPS	Central, personal (estimated)	NO ₂ , NO, PM _{2.5} , PM ₁₀	No
Amsterdam, Erfurt, Helsinki (ULTRA Study)	Timonen et al. 2006	Panel (repeated measure)	1998–1999	NC 0.01-0.1, NC _{0.01-1}	AS + CPC	Central	NO ₂ , CO, NC _{0.01–1} , PM _{2.5} , PM _{2.5–10}	NO ₂ , CO, O ₃ , PM _{2.5}
Boston, MA	Park et al. 2005	Panel (cross- sectional)	2000-2003	Total NC	CPC	Central	No	No Iable continues next page
a APC = aerosol pi mobility particle hydrocarbons; Pi soluble organic c	urticle counter; AS = sizer; EAS = electric MF = positive matrix arbon.	aerosol spectrometer : aerosol spectromete : factorization; SMPS	;; BC = black carbc ar; MAS = mobile = scanning mobi	m; BS = black smoke; aerosol spectrometer lity particle sizer; SO	; CPC = condensat ; MC = mass conc)C - secondary org	ion particle counte entration; NC = par ganic carbon; TDMF	: DMA = differential mobili ticle number concentration; 35 = twin differential mobili	ty analyzer; DMPS = differential ; PAHs = polycyclic aromatic ity particle sizer; WSOC = water
^b Study abbreviati Relationship bet	ons: AIRGENE = Athe ween Ultrafine and fi	ens, Augsburg, Barcel ine Particulate matte	lona, Helsinki, Ro r in Indoor and O	me, Stockholm; NAS utdoor air and respir	= Normative Agin atory Health; ULT	g Study; PEACE = F 'RA = Exposure and	ollution Effects in Asthmatic l Risk Assessment for Fine a	c Children in Europe; RUPIOH = and Ultrafine Particles in Air.
^c Diameter size rai diameter or UFP	ages are in micrometo NC).	ers (μm). Several arti	cles considered π	ultiple size fractions	s. The bolded metr	ric is the one closes	t to the definition used in th	uis document (i.e., < 0.100 µm in

^d These are two-pollutant models unless otherwise indicated.

Appendix Tak Geographic Lo	ile B.1 (<i>Continu</i> e ocation ^a	ə d). Primary Re	search Article	s Presenting Res	ults of Total a	nd UFP NC Epi	idemiologic Studies b	y Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
HRV (Continu	ed)							
New Jersey	Laumbach et al. 2010	Panel (pilot study)	NA	NC _{0.01-1.0}	CPC	Personal	CO, NO ₂ , PM _{2.5}	No
Taipei, Taiwan	Chan et al. 2004	Panel (cross- sectional)	NA	NC _{0.02-1}	P-Trak Ultrafine particle counter	Personal	No	No
Rochester, NY	Rich et al. 2012	Panel (repeated measure)	2006–2009	NC0.01-0.1 NC _{0.1-0.5}	Wide range particle spectrometer	Central	PM _{2.5} , NC _{0.1-0.5}	PM _{2.5} , NC _{0.1-0.5}
Ottawa, Canada	Weichenthal et al. 2011	Panel (repeated measure)	2010	NC _{0.01-1}	CPC	Personal	CO, SO ₂ , NO ₂ , O ₃ , PM _{2.5} , BC	O ₃ , BC
Arrhythmia a	nd Related Endp	oints						
London, England	Anderson et al. 2010	Case- crossover	1995–2003	Total NC	CPC	Central	CO, SO ₂ , O ₃ , NO _x , NO, NO ₂ , SO ₄ , PM _{2.5} , PM ₁₀ , BS	No
Augsburg, Germany	Hampel et al. 2010	Panel (repeated measure)	2003-2004	Total NC	CPC	Central	CO, NO ₂ , PM _{2.5} , PM _{2.5-10} , PM ₁₀	No
							L	able continues next page
a APC = aerosol p mobility particl hydrocarbons; F soluble organic	article counter; AS = s sizer; EAS = electric MF = positive matrix carbon.	aerosol spectromete: c aerosol spectromet factorization; SMP5	r; BC = black carbo er; MAS = mobile 3 = scanning mobi	n: BS = black smoke aerosol spectrometer lity particle sizer; SC	; CPC = condensat r; MC = mass conc DC - secondary or;	ion particle counter :entration; NC = par ganic carbon; TDMF	r; DMA = differential mobilit ticle number concentration; SS = twin differential mobili	y analyzer; DMPS = differential PAHs = polycyclic aromatic ty particle sizer; WSOC = water
^b Study abbreviati Relationship bel	ons: AIRGENE = Athe ween Ultrafine and fi	ens, Augsburg, Barce ine Particulate matte	elona, Helsinki, Ro 3r in Indoor and C	me, Stockholm; NAS Jutdoor air and respir	= Normative Agin ratory Health; ULJ	g Study; PEACE = P TRA = Exposure and	'ollution Effects in Asthmatic l Risk Assessment for Fine a	: Children in Europe; RUPIOH = nd Ultrafine Particles in Air.
^c Diameter size ra	nges are in micromet	ers (μm). Several arti	icles considered n	nultiple size fractions	s. The bolded met	ric is the one closest	t to the definition used in thi	is document (i.e., < 0.100 µm in

biameter or UFP NC). ^d These are two-pollutant models unless otherwise indicated.

Appendix Tab Geographic Lc	le B.1 (<i>Continu</i> ¢)cation ^a	ə d). Primary Re	search Article	s Presenting Res	ults of Total a	nd UFP NC Epi	demiologic Studies by	Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Arrhythmia a	nd Related Endp	oints (Continue	(pé					
Erfurt, Germany	Berger et al. 2006	Panel (repeated measure)	2000–2001	NC _{0.01-0.1} , NC _{0.1-1}	MAS	Central	CO, NO, NO ₂ , SO ₂ , NC _{0.1-1} , PM _{2.5}	No
	Henneberger et al. 2005	Panel (repeated measure)	2000–2001	NC_{0.01-0.1} , NC _{0.01-1}	MAS	Central	NO, NO ₂ , CO, PM _{2.5} , NC _{0.01-1} , EC, OC	No
	Yue et al. 2007	Panel (repeated measure)	2000–2001	NC _{0.01-0.1} , NC source factors (PMF)	MAS	Central	No	No
Aberdeen, Scotland	Barclay et al. 2009	Panel (repeated measure)	2003–2005	NC _{0.01-0.1}	SMPS	Central, personal (estimated)	NO ₂ , NO, PM _{2.5} , PM ₁₀	No
Eastern MA	Dockery et al. 2005a,b	Case- crossover	1995–2002	Total NC	CPC	Central	No	No
Ischemia								
Helsinki, Finland (ULTRA	Lanki et al. 2008	Panel (repeated measure)	1999	NC<0.1	EAS	Central	Outdoor PM _{2.5} , personal PM _{2.5}	No
stuay	Pekkanen et al. 2002	Panel (repeated measure)	1998–1999	NC _{0.01-0.1} , NC _{0.1-1}	EAS	Central	CO, NO ₂ , PM _{2.5-10} , PM _{2.5} , PM ₁ , NC _{0,1-1}	PM _{2.5} , PM ₁ , NC _{0.1-1}
							Та	tble continues next page
a APC = aerosol p mobility particle hydrocarbons; P soluble organic e	article counter; AS = s sizer; EAS = electric MF = positive matrix carbon.	aerosol spectrometer c aerosol spectromet factorization; SMP5	;; BC = black carbc er; MAS = mobile } = scanning mobi	m; BS = black smoke; aerosol spectrometer lity particle sizer; SO	; CPC = condensat r; MC = mass conc)C – secondary orę	ion particle counter centration; NC = par ganic carbon; TDMF	; DMA = differential mobility ticle number concentration; F S = twin differential mobility	analyzer; DMPS = differential AHs = polycyclic aromatic y particle sizer; WSOC = water
^b Study abbreviati Relationship bet	ons: AIRGENE = Ath. ween Ultrafine and fi	ens, Augsburg, Barce ine Particulate matte	lona, Helsinki, Ro r in Indoor and O	me, Stockholm; NAS butdoor air and respir	= Normative Agin atory Health: ULT	ig Study; PEACE = P TRA = Exposure and	ollution Effects in Asthmatic (l Risk Assessment for Fine an	Children in Europe; RUPIOH = id Ultrafine Particles in Air.

^c Diameter size ranges are in micrometers (µm). Several articles considered multiple size fractions. The bolded metric is the one closest to the definition used in this document (i.e., <0.100 µm in diameter or UFP NC).

^d These are two-pollutant models unless otherwise indicated.

Understanding the Health Effects of Ambient Ultrafine Particles

Appendix Tab Geographic Lo	l e B.1 (<i>Continue</i> cation ^a	d). Primary Res	earch Article	s Presenting Res	ults of Total a	nd UFP NC Ep	idemiologic Studies ł	y Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Ischemia (Con	tinued)							
Los Angeles, CA	Delfino et al. 2011	Panel (repeated measure)	2005-2007	Total NC PM0.25	CPC, Sioutas Personal Cascade Impactor Sampler	Central & indoor	O ₃ , CO, NO _x , PM _{2.5-10} , PM _{2.5} , PM _{0.25-2.5} , PM _{0.25} , OC, BC	No
Vascular Reac	livity							
Boston, MA	O'Neill et al. 2005	Panel (cross- sectional)	1998–2002	Total NC	CPC	Central	SO ₄ ²⁻ , BC, PM _{2.5}	No
Ottawa, Canada	Dales et al. 2007	Panel (repeated measure)	ΥN	NC _{0.02-1}	P-Trak Ultrafine Particle Counter	Personal	$PM_{2.5}$	No
Blood Pressure								
Ottawa, Canada	Dales et al. 2007	Panel (repeated measure)	ΥN	NC _{0.02-1}	P-Trak Ultrafine Particle Counter	Personal	$PM_{2.5}$	No
Taipei, Taiwan	Chuang et al. 2005	Panel (repeated measure)	ΝΑ	NC _{0.02-1}	P-Trak Ultrafine Particle Counter	Personal (NC _{0.02–1}), Central	CO, NO ₂ , PM ₁₀	CO, PM ₁₀
								Table continues next page
a APC = aerosol pe mobility particle hydrocarbons; P ¹ soluble organic c	rticle counter; AS = s sizer; EAS = electric AF = positive matrix arbon.	lerosol spectrometer aerosol spectromete factorization; SMPS	; BC = black carbc x; MAS = mobile = scanning mobi	m; BS = black smoke; aerosol spectrometer lity particle sizer; SO	CPC = condensat ; MC = mass conc C - secondary org	ion particle counte entration; NC = pa ganic carbon; TDM	r; DMA = differential mobil rticle number concentratior PS = twin differential mobil	ity analyzer; DMPS = differential t; PAHs = polycyclic aromatic lity particle sizer; WSOC = water
^b Study abbreviatic Relationship bet	ons: AIRGENE = Athe veen Ultrafine and fi	ns, Augsburg, Barcel ne Particulate matte	ona, Helsinki, Ro r in Indoor and C	me, Stockholm; NAS : utdoor air and respir	= Normative Agin atory Health; ULT	g Study; PEACE = l 'RA = Exposure an	Pollution Effects in Asthmati d Risk Assessment for Fine	ic Children in Europe; RUPIOH = and Ultrafine Particles in Air.
^c Diameter size rar diameter or UFP	iges are in micromete NC).	rs (µm). Several arti	cles considered n	uultiple size fractions	. The bolded metr	ic is the one closes	st to the definition used in tl	his document (i.e., < 0.100 µm in

^d These are two-pollutant models unless otherwise indicated.

Appendix Tal Geographic L	ole B.1 (Continue ocation ^a	ə d). Primary Re	search Article	s Presenting Res	ults of Total a	nd UFP NC Epi	idemiologic Studies by	/ Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Blood Pressu	e (Continued)							
Los Angeles, CA	Delfino et al. 2010c	Panel (repeated measure)	2005–2007	Total NC	CPC	Central (outdoor home)	CO, NO _x , O ₃ , BC, OC	No
Rochester, NY	Rich et al. 2012	Panel (repeated measure)	2006–2009	NC 0.01-0.1 [,] NC _{0.1-0.5}	Wide range particle spectrometer	Central	PM2.5, NC0.1-0.5	PM _{2.5} , NC _{0.1-0.5}
Amsterdam, Erfurt, Helsinki (ULTRA Study)	Ibald-Mulli et al. 2004	Panel (repeated measure)	1998–1999	NC _{0.01-0.1} , NC _{0.1-1}	EAS	Central	SO ₂ , CO, NO ₂ , NC _{0.1-1} , PM _{2.5}	No
Soluble Mark	ers of Inflammat	tion, Coagulatio	n, and Oxidat	tive Stress ^e				
Augsburg, Germany (AIRGENE	Brüske et al. 2011*	Panel (repeated measure)	2003–2004	NC _{<0.1}	CPC	Central	O ₃ , SO ₂ , NO, CO. PM _{2.5} , PM ₁₀	No
stuayJ	Kraus et al. 2011*^	Panel (repeated measure)	2003–2004	Total NC	CPC	Central	PM _{2.5} , hopanes, PAHs	No
							Ľ	able continues next page
a APC = aerosol] mobility partic] hydrocarbons;] soluble organic	aarticle counter; AS = . e sizer; EAS = electric "MF = positive matrix carbon.	aerosol spectromete c aerosol spectromet : factorization; SMP	r; BC = black carb er; MAS = mobile S = scanning mobi	m; BS = black smoke aerosol spectrometer llity particle sizer; SC	; CPC = condensat r; MC = mass conc DC – secondary or;	ion particle counter entration; NC = par ganic carbon; TDMF	;; DMA = differential mobilit ticle number concentration; 55 = twin differential mobilit	y analyzer; DMPS = differential PAHs = polycyclic aromatic y particle sizer; WSOC = water
^b Study abbreviat Relationship be	ions: AIRGENE = Athe tween Ultrafine and fi	ens, Augsburg, Barce ine Particulate matt	elona, Helsinki, Ro er in Indoor and C	me, Stockholm; NAS Dutdoor air and respir	= Normative Agin ratory Health; ULI	g Study; PEACE = P 'RA = Exposure and	ollution Effects in Asthmatic l Risk Assessment for Fine a	Children in Europe; RUPIOH = nd Ultrafine Particles in Air.
^c Diameter size r diameter or UF	inges are in micromet ⁽ o NC).	ers (µm). Several art	icles considered n	aultiple size fractions	s. The bolded meti	ric is the one closes	t to the definition used in thi	s document (i.e., < 0.100 µm in

d These are two-pollutant models unless otherwise indicated.

e Markers examined: * systemic inflammation; ^ coagulation; # oxidative stress.

Geographic Lu	ocation ^a	6u). I IIIII AI Y		cavi Ammiraca i i ci	uits UL 1 Ulal a		remmonogic organist	
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Soluble Mark	ers of Inflamma	tion, Coagulatic	n, and Oxidat	tive Stress (Cont	inued) ^e			
Erfurt, Germany	Brüske et al. 2010*	Panel (repeated measure)	2001–2002	NC _{0.01-0.1} , NC _{0.1-1}	MAS	Central	CO, NO, NO ₂ , SO ₂ , OC, EC, PM ₁₀ , NC _{0.1-1}	CO, NO_2
	Hildebrandt et al. 2009*^	Panel (repeated measure)	2001–2002	NC _{0.01-0.1} , NC _{0.1-1}	DMPS, CPC + SMPS	Central	NC _{0.1-1}	No
	Rückerl et al. 2006*^	Panel (repeated measure)	2000–2001	NC _{0.01-0.1} , NC _{0.1-1}	MAS	Central	CO, NO ₂ , PM _{2.5} , PM ₁₀	No
	Rückerl et al. 2007b*	Panel (repeated measure)	2000–2001	NC _{0.01-0.1} , NC _{0.1-1}	MAS	Central	NO, PM _{2.5} , PM ₁₀	No
	Yue et al. 2007*	Panel (repeated measure)	2000–2001	NC _{0.01-0.1}	MAS, with factor analysis by PMF	Central	No	No
								Table continues next page
a APC = aerosol I mobility particl hydrocarbons; I soluble organic	article counter; AS = e sizer; EAS = electri 'MF = positive matriy carbon.	aerosol spectromet c aerosol spectrome c factorization; SMP	ar; BC = black carb ter; MAS = mobile S = scanning mobi	on; BS = black smoke s aerosol spectrometer ility particle sizer; SC	; CPC = condensat ;; MC = mass conc)C - secondary or	tion particle counter centration; NC = par ganic carbon; TDMF	; DMA = differential mobil ticle number concentration 'S = twin differential mobi	ity analyzer; DMPS = differential 1; PAHs = polycyclic aromatic lity particle sizer; WSOC = water
^b Study abbreviat Relationship be	ions: AIRGENE = Ath tween Ultrafine and 1	ens, Augsburg, Barc fine Particulate matt	elona, Helsinki, Ro er in Indoor and C	ome, Stockholm; NAS Dutdoor air and respir	= Normative Agin atory Health; ULJ	ıg Study; PEACE = P IRA = Exposure and	ollution Effects in Asthmat Risk Assessment for Fine	ic Children in Europe; RUPIOH = and Ultrafine Particles in Air.

^d These are two-pollutant models unless otherwise indicated.

 $^{\circ}$ Diameter size ranges are in micrometers (µm). Several articles considered multiple size fractions. The bolded metric is the one closest to the definition used in this document (i.e., < 0.100 µm in diameter or UFP NC).

e Markers examined: * systemic inflammation; ^ coagulation; # oxidative stress.

Appendix Tab Geographic Lo	le B.1 (<i>Continue</i> cation ^a	d). Primary Re.	search Article:	s Presenting Res	ults of Total a	nd UFP NC Ep	idemiologic Studies l	oy Outcome and
Outcome / Location ^b	Reference	Study Design	Study Period	Selected UFP Metric ^c	Sampling Equipment	UFP Exposure Assignment	Copollutant Correlations	Copollutants in Two-Pollutant Models? ^d
Soluble Marke	rs of Inflammat	ion, Coagulatio	n, and Oxidat	ive Stress (Cont	inued) ^e			
Ruhr area, Germany	Hertel et al. 2010*	Panel (cross- sectional)	2000-2003	Total NC	NA	Residence modeled from central site using Euro- pean Air Pollution and Dispersion Model (EURAD)	oZ	No
Aberdeen, Scotland	Barclay et al. 2009*^	Panel (repeated measure)	2003–2005	NC _{0.01-0.1}	SMPS	Central, personal (estimated)	NO ₂ , NO, PM _{2.5} , PM ₁₀	No
Arnhem, the Netherlands	Zuurbier et al. 2011a*^	Panel (repeated measure)	2007-2008	Total NC	CPC (TSI 3007)	Personal	$PM_{2.5}$, PM_{10} , soot	No
6-Cities (AIRGENE study)	Ljungman et al. 2009*	Panel (repeated measures)	2003-2004	Total NC	CPC	Central	CO, NO ₂	No
	Rückerl et al. 2007a*^	Panel (repeated measure)	2003–2004	Total NC	CPC	Central	Not reported	No
								Table continues next page
a APC = aerosol pe mobility particle hydrocarbons; Pl soluble organic c	urticle counter; AS = (sizer; EAS = electric AF = positive matrix arbon.	aerosol spectromete: : aerosol spectromet factorization; SMP5	r; BC = black carbc er; MAS = mobile 3 = scanning mobi	m; BS = black smoke aerosol spectrometer lity particle sizer; SO	; CPC = condensat c; MC = mass conc)C – secondary or{	ion particle counte :entration; NC = pa ganic carbon; TDM	r; DMA = differential mobil tticle number concentration PS = twin differential mobi	ity analyzer; DMPS = differential 1; PAHs = polycyclic aromatic lity particle sizer; WSOC = water
^b Study abbreviatic Relationship bet	ons: AIRGENE = Athe veen Ultrafine and fi	ans, Augsburg, Barce ine Particulate matte	lona, Helsinki, Ro 3r in Indoor and O	me, Stockholm; NAS utdoor air and respir	= Normative Agin atory Health; ULT	ig Study; PEACE = F TRA = Exposure and	² ollution Effects in Asthmat d Risk Assessment for Fine	ic Children in Europe; RUPIOH = and Ultrafine Particles in Air.
^c Diameter size rai diameter or UFP	nges are in micromete NC).	ərs (μm). Several art	icles considered m	ultiple size fractions	s. The bolded met	ric is the one closes	t to the definition used in t	his document (i.e., < 0.100 µm in

e Markers examined: * systemic inflammation; ^ coagulation; # oxidative stress.

^d These are two-pollutant models unless otherwise indicated.

$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	- .				Selected	-	ŗ	- - - - -	Copollutants in
	Uutcome / Location ^b	Reference	ətudy Design	Study Period	UFF Metric ^c	Sampling Equipment	Exposure Assignment	Correlations	1wo-Pollutant Models? ^d
Boston, MA Ren et al. Panel (cross 2006–2008 Total NC reported central NO ₂ , BC, EC, PM _{6,5} , No 2001–201 Panel (cross 2000–2004 NC _{0.007–3} , CPC Central NO ₂ , BC, PM _{6,5} , No 2006+A Rich et al. Panel (cross 2000–2004 NC _{0.007–3} , CPC Central SO ₄ ²⁻ , BC, PM _{6,5} , No 2006+A Rich et al. Panel (cross 2000–2004 NC _{0.001–41} , Wide range Central PM _{2,5} , NC _{01–0.5} , PM _{2,5} , NC _{01–0.5} , No 2005 Angeles, Delfino terasure) consure) Condant CPC Central Router NC _{01–0.5} , PM _{2,5} , NC _{01–0.5} , PM _{2,5} , NC _{01–0.5} , PM _{2,5} , NC _{01–0.5} , So 2003–2006 Total NC CPC CPC Central Router NC _{01–0.5} , PM _{2,5} , NC _{01–0.5} , So 2003 ⁺⁺ (repeated Router) Router Ro	Soluble Mark	ers of Inflamma	tion, Coagulatic	on, and Oxida	tive Stress (Co	ntinued) ^e			
Zeka et al. Panel (cross sectional) NC _{0.00-2004} NC _{0.00-304} NC _{0.00-304} NC _{0.00-305} NO <th< td=""><td>Boston, MA (NAS)</td><td>Ren et al. 2011[#]</td><td>Panel (cross sectional)</td><td>2006-2008</td><td>Total NC</td><td>Not reported</td><td>Central</td><td>CO, O₃, SO₄²⁻, NO₂, BC, EC, PM_{2.5}</td><td>No</td></th<>	Boston, MA (NAS)	Ren et al. 2011 [#]	Panel (cross sectional)	2006-2008	Total NC	Not reported	Central	CO, O ₃ , SO ₄ ²⁻ , NO ₂ , BC, EC, PM _{2.5}	No
Rochester, NY Rich et al. Panel 2006–2009 NC _{0.1-0.5} NC _{0.1-0.5} Wide range particle PM _{2.5} , NC _{0.1-0.5} PM _{2.5} , NC _{0.1-0.5} Los Angeles, CA Delfino Panel 2005–2006 Total NC CPC Central & No No Los Angeles, CA Delfino Panel 2005–2007 Total NC CPC Central & No No Los Angeles, CA Delfino Panel 2005–2007 Total NC CPC Central & No No Los Angeles, CA Delfino Panel 2005–2007 Total NC CPC Central & No No Los Angeles, CA Tenal 2005-2007 Total NC CPC Central & No No Delfino Panel 2005-2007 Total NC CPC Central & No No Ressure) measure) 2005-2007 Total NC CPC Central & No No Delfino Panel 2005-2007 Mo _{0.25} Sioutas No No Delfino Panel 2005-2007 <td< td=""><td></td><td>Zeka et al. 2006*^</td><td>Panel (cross sectional)</td><td>2000–2004</td><td>NC_{0.007-3}</td><td>CPC</td><td>Central</td><td>${\rm SO_4}^{2-}, {\rm BC}, {\rm PM}_{2.5}$</td><td>No</td></td<>		Zeka et al. 2006*^	Panel (cross sectional)	2000–2004	NC _{0.007-3}	CPC	Central	${\rm SO_4}^{2-}, {\rm BC}, {\rm PM}_{2.5}$	No
Los Angeles, Delfino Panel 2005–2006 Total NC CPC Central & No No CA et al. 2008*^ (repeated) 2005–2007 Total NC CPC Gentral & No No CA et al. 2008** (repeated) 2005–2007 Total NC CPC Gentral 03, CO, NO2, NOx, NO No Delfino Panel 2005–2007 Total NC CPC Gentral 03, CO, NO2, NOx, NO No Delfino Panel 2005–2007 PM0.25 Sioutas Central & Organic components PAHs, hopanes Delfino Panel 2005–2007 PM0.25 Sioutas Central & Organic components PAHs, hopanes Delfino Panel 2005–2007 PM0.25 Sioutas Central & Organic components PAHs, hopanes Delfino Panel 2005–2007 PM0.25 PM0.25 PM0.25 PM0.25 PMS.25 <td>Rochester, NY</td> <td>Rich et al. 2012</td> <td>Panel (repeated measure)</td> <td>2006–2009</td> <td>NC0.01-0.1, NC_{0.1-0.5}</td> <td>Wide range particle spectrometer</td> <td>Central</td> <td>PM_{2.5}, NC_{0.1-0.5}</td> <td>PM_{2.5}, NC_{0.1-0.5}</td>	Rochester, NY	Rich et al. 2012	Panel (repeated measure)	2006–2009	NC 0.01-0.1, NC _{0.1-0.5}	Wide range particle spectrometer	Central	PM _{2.5} , NC _{0.1-0.5}	PM _{2.5} , NC _{0.1-0.5}
Delfino et al. Panel 2005–2007 Total NC CPC Central 0.3, CO, NO, NO, NO No 2009 *# (repeated 0.1 PM0.25–2.3, PM.2.5–10 PM0.25–2.3, PM.2.5–10 PM0.25–2.3, PM.2.5–10 Delfino Panel 2005–2007 PM0.25 Sioutas Central & Organic components PAHS, hopanes Delfino Panel 2005–2007 PM0.25 Sioutas Central & Organic components PAHS, hopanes neasure) measure) 2005–2007 PM0.25 Sioutas Central & Organic components PAHS, hopanes at al. 2010b* (repeated 2005–2007 PM0.25 Sioutas Central & Organic components PAHS, hopanes at al. 2010b* (repeated 2005–2007 PM0.25 Sioutas Organic Of PM0.25 (PAHS, inclusion) PHS, inclusion) at al. 2010b* (repeated 2005–2007 PM0.25 Sioutas Organic Of PM0.25 (PAHS, inclusion) DM0.25 (PAHS, in	Los Angeles, CA	Delfino et al. 2008*^	Panel (repeated measure)	2005–2006	Total NC	CPC	Central & indoor	No	No
Delfino Panel 2005–2007 PM _{0.25} Sioutas Central & Organic components PAHs, hopanes et al. 2010b* (repeated 2005–2007 PM _{0.25} Personal indoor of PM _{0.25} (PAHs, hopanes measure) Cascade hopanes, WSOC, in-Alkanes, organic hopanes, WSOC, in-Alkanes, organic in-Alkanes, organic a APC = aerosol particle counter: AS = aerosol spectrometer; BC = black carbon; BS = black smoke; CPC = condensation particle counter; DMA = differential mobility analyzer; DMPS = differential a APC = aerosol particle sizer; BAS = electric aerosol spectrometer; MC = mass concentration; NC = particle number concentration; PAHs = polycyclic aromatic hydrocarbons; PMF = positive matrix factorization; SMPS = scanning mobility particle sizer; SOC - secondary organic carbon; TDMPS = twin differential mobility particle sizer; WSOC = water b Study abbreviations: AIRGENE = Athens, Augsburg, Barcelona, Helsinki, Rome, Stockholm; NAS = Normative Aging Study; PEACE = Pollution Effects in Asthmatic Children in Europe; RUPIOH = b Study abbreveen Ultrafine and fine Particulate matter in Indoor and Outdoor air and respiratory Health; ULTRA = Exposure and Risk Assessment for Fine and Ultrafine Particulate matter in Indoor and Outdoor air and respiratory Health; ULTRA = Exposure and Risk Assessment for Fine and Ultrafine Particles in Asthmatic Children in Europes in Air		Delfino et al. 2009*#	Panel (repeated measure)	2005–2007	Total NC	CPC	Central	O ₃ , CO, NO ₂ , NO _x , OC, BC, SOC, PM _{0.25} , PM _{0.25-2.5} , PM _{2.5-10}	No
a APC = aerosol particle counter; AS = aerosol spectrometer; BC = black carbon; BS = black smoke; CPC = condensation particle counter; DMA = differential mobility analyzer; DMPS = differential mobility particle sizer; EAS = electric aerosol spectrometer; MAS = mobile aerosol spectrometer; MC = mass concentration; NC = particle number concentration; PAHs = polycyclic aromatic hydrocarbons; PMF = positive matrix factorization; SMPS = scanning mobility particle sizer; SOC - secondary organic carbon; TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon; TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon; TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon. TDMPS = twin differential mobility particle sizer; WSOC = water soluble organic carbon.		Delfino et al. 2010b*	Panel (repeated measure)	2005–2007	PM _{0.25}	Sioutas Personal Cascade Impactors	Central & indoor	Organic components of PM _{0.25} (PAHs, hopanes, WSOC, n-Alkanes, organic acids)	PAHs, hopanes
^b Study abbreviations: AIRGENE = Athens, Augsburg, Barcelona, Helsinki, Rome, Stockholm; NAS = Normative Aging Study; PEACE = Pollution Effects in Asthmatic Children in Europe; RUPIOH Relationship between Ultrafine and fine Particulate matter in Indoor and Outdoor air and respiratory Health; ULTRA = Exposure and Risk Assessment for Fine and Ultrafine Particles in Air.	a APC = aerosol F mobility particl hydrocarbons; F soluble organic	article counter; AS = e sizer; EAS = electri. 'MF = positive matrix carbon.	aerosol spectromete c aerosol spectromet ¢ factorization; SMP	rr; BC = black carb ter; MAS = mobile S = scanning mob	on; BS = black smc 9 aerosol spectrom ility particle sizer;	ke; CPC = condensa eter; MC = mass cond SOC – secondary or	tion particle count centration; NC = p. ganic carbon; TDM	er; DMA = differential mobility article number concentration; I IPS = twin differential mobility	' analyzer; DMPS = differenti PAHs = polycyclic aromatic y particle sizer; WSOC = wate
	^b Study abbreviat Relationship be	ions: AIRGENE = Ath tween Ultrafine and 1	ens, Augsburg, Barce line Particulate matt	elona, Helsinki, Rc er in Indoor and C	ome, Stockholm; N. Outdoor air and res	AS = Normative Agir spiratory Health; UL	ıg Study; PEACE = IRA = Exposure ar	Pollution Effects in Asthmatic ad Risk Assessment for Fine an	Children in Europe; RUPIOH = id Ultrafine Particles in Air.

d These are two-pollutant models unless otherwise indicated.

e Markers examined: * systemic inflammation; ^ coagulation; # oxidative stress.

ABBREVIATIONS AND OTHER TERMS

ACES	Advanced Collaborative Emissions Study
AMS	aerosol mass spectrometer
APC	aerosol particle counter
AS	aerosol spectrometer
BC	black carbon
BP	blood pressure
BS	black smoke
CAP	concentrated ambient particles
CASAC	Clean Air Scientific Advisory Committee
CNG	compressed natural gas
CO	carbon monoxide
COPD	chronic obstructive pulmonary disease
CPC	condensation particle counter
CRP	C-reactive protein
CVD	cardiovascular disease
DE	diesel exhaust
DISI	direct injection spark ignition
DMA	differential mobility analyzers
DMPS	differential mobility particle sizer
DPF	diesel particle filters
EAS	electric aerosol spectrometer
EC	elemental carbon
ECG	electrocardiogram
EEPS	engine exhaust particle sizer
ELPI	electrical low pressure impactor
FEV_1	forced expiratory volume in 1 sec
FMPS	fast mobility particle sizer
FVC	forced vital capacity
HR	heart rate
HRV	heart-rate variability
ICD	implanted cardioverter defibrillator
ICRP	International Commission on Radiological Protection
Ig	immunoglobulin
IL	interleukin
MAS	mobile aerosol spectrometer
MC	mass concentration
MOUDI	micro-orifice uniform deposit impactor
NAMS	nanoaerosol mass spectrometer
NC	number concentration
NF-κB	DNA transcription factor NF-κB

NO _x	nitrogen oxides
NO	nitric oxide
NO_2	nitrogen dioxide
NPACT	National Particle Component Toxicity
O_3	ozone
OC	organic carbon
OVA	ovalbumin
PAHs	polycyclic aromatic hydrocarbons
PEF	peak expiratory flow
PM	particulate matter
$PM_{2.5}$	$PM \leq 2.5~\mu m$ in aerodynamic diameter
PM_{10}	$PM \leq 10~\mu m$ in aerodynamic diameter
PM ISA	Integrated Science Assessment for Particulate Matter
PMF	positive matrix factorization
SCR	selective catalytic reduction
SMPS	scanning mobility particle sizer
TDMPS	twin differential mobility particle sizer
TNFα	tumor necrosis factor alpha
UF1	UFPs > 3 nm
UF2	UFPs > 15 nm
UFP	ultrafine particle
UFP NC	UFP number concentration
ULTRA	Exposure and risk assessment for fine & ultrafine particles in ambient air, a European Union-funded study
U.S. EPA	U.S. Environmental Protection Agency
VACES	versatile aerosol concentrator enrichment system

WSOC water soluble organic carbon

HEI BOARD, COMMITTEES, and STAFF

Board of Directors

Richard F. Celeste, Chair President Emeritus, Colorado College

Sherwood Boehlert Of Counsel, Accord Group; Former Chair, U.S. House of Representatives Science Committee

Enriqueta Bond President Emerita, Burroughs Wellcome Fund

Purnell W. Choppin President Emeritus, Howard Hughes Medical Institute

Michael T. Clegg Professor of Biological Sciences, University of California-Irvine

Jared L. Cohon President, Carnegie Mellon University

Stephen Corman President, Corman Enterprises

Gowher Rizvi Vice Provost of International Programs, University of Virginia

Linda Rosenstock Dean Emerita and Professor of Health Policy and Management, Environmental Health Sciences and Medicine, University of California–Los Angeles

Henry Schacht Managing Director, Warburg Pincus; Former Chairman and Chief Executive Officer, Lucent Technologies Warren M. Washington Senior Scientist, National Center for Atmospheric Research; Former Chair, National Science Board

Archibald Cox, Founding Chair 1980-2001

Donald Kennedy, Vice Chair Emeritus Editor-in-Chief Emeritus, Science; President Emeritus and Bing Professor of Biological Sciences, Stanford University

Health Research Committee

David L. Eaton, Chair Associate Vice Provost for Research and Director, Center for Ecogenetics and Environmental Health, School of Public Health, University of Washington–Seattle

David Christiani Elkan Blout Professor of Environmental Genetics, Harvard School of Public Health

David E. Foster Phil and Jean Myers Professor Emeritus, Department of Mechanical Engineering, Engine Research Center, University of Wisconsin–Madison

Uwe Heinrich Professor, Medical School Hannover; Executive Director, Fraunhofer Institute for Toxicology and Experimental Medicine, Hanover, Germany

Grace LeMasters Professor of Epidemiology and Environmental Health, University of Cincinnati College of Medicine

Sylvia Richardson Professor and Director, MRC Biostatistics Unit, Institute of Public Health, Cambridge, United Kingdom

Allen L. Robinson Professor, Departments of Atmospheric Science and Mechanical Engineering, Colorado State University

Richard L. Smith Director, Statistical and Applied Mathematical Sciences Institute, University of North Carolina–Chapel Hill

James A. Swenberg Kenan Distinguished Professor of Environmental Sciences, Department of Environmental Sciences and Engineering, University of North Carolina–Chapel Hill

HEI BOARD, COMMITTEES, and STAFF

Health Review Committee

Homer A. Boushey, Chair Professor of Medicine, Department of Medicine, University of California–San Francisco

Ben Armstrong Reader in Epidemiological Statistics, Public and Environmental Health Research Unit, Department of Public Health and Policy, London School of Hygiene and Tropical Medicine, United Kingdom

Michael Brauer Professor, School of Environmental Health, University of British Columbia, Canada

Bert Brunekreef Professor of Environmental Epidemiology, Institute of Risk Assessment Sciences, University of Utrecht, the Netherlands

Mark W. Frampton Professor of Medicine and Environmental Medicine, University of Rochester Medical Center

Stephanie London Senior Investigator, Epidemiology Branch, National Institute of Environmental Health Sciences

Armistead Russell Howard T. Tellepsen Chair of Civil and Environmental Engineering, School of Civil and Environmental Engineering, Georgia Institute of Technology

Lianne Sheppard Professor of Biostatistics, School of Public Health, University of Washington-Seattle

Officers and Staff

Daniel S. Greenbaum President Robert M. O'Keefe Vice President Rashid Shaikh Director of Science Barbara Gale Director of Publications Jacqueline C. Rutledge Director of Finance and Administration Helen I. Dooley Corporate Secretary

Kate Adams Senior Scientist Johanna Boogaard Staff Scientist Aaron J. Cohen Principal Scientist Maria G. Costantini Principal Scientist Philip J. DeMarco Compliance Manager Suzanne Gabriel Editorial Assistant Hope Green Editorial Assistant (part time) L. Virgi Hepner Senior Science Editor Anny Luu Administrative Assistant Francine Marmenout Senior Executive Assistant Nicholas Moustakas Policy Associate Hilary Selby Polk Senior Science Editor Jacqueline Presedo Research Assistant Sarah Rakow Science Administrative Assistant Evan Rosenberg Staff Accountant Robert A. Shavers Operations Manager Geoffrey H. Sunshine Senior Scientist Annemoon M.M. van Erp Managing Scientist Katherine Walker Senior Scientist



H E A L T H EF F E C T S IN STITUTE

101 Federal Street, Suite 500 Boston, MA 02110-1817, USA Phone +1-617-488-2300 Fax +1-617-488-2335 *www.healtheffects.org*

HEI PERSPECTIVES 3 January 2013